Gas Retail Governance

Final Proposals

March 2004 50/04

Executive Summary

In June 2003 Ofgem consulted on the development and implementation of a Supply Point Administration Agreement ("SPAA"), intended to provide new governance for gas industry standard procedures which are currently ungoverned, or for which the existing governance is considered to be inappropriate. In particular, the SPAA will govern the new competitive metering processes developed under the Review of Gas Metering Arrangements project¹.

The SPAA has already benefited from three separate Ofgem consultation documents and discussion at numerous industry meetings. This document outlines the views we have received from respondents to our June 2003 consultation document and sets out Ofgem's final proposals, in light of those responses. Ofgem is at this stage consulting on the detail of what the agreement should cover, both now and in future and who should be a party. At the absolute minimum, it is vital that the processes contained within the RGMA Baseline document are appropriately owned and maintained. Further, Ofgem see no reason why the processes currently residing in the voluntary Domestic Codes of Practice cannot appropriately reside in SPAA.

Whilst the SPAA was developed primarily as an agreement between suppliers, it was envisaged that a second phase of development would allow GTs to accede to the agreement, thus facilitating a migration of SPA provisions from the existing Network Codes into SPAA. Respondents identified the need for GTs to be SPAA parties either at the outset, or at the soonest opportunity thereafter.

Following consideration of responses to the June document, Ofgem has amended the text of the SPAA, primarily in order to allow for GT accession and to address various other issues identified by respondents. A tracked version of the revised document, highlighting these changes is provided as Appendix 5.

Views on these final proposals are invited by 2 April 2004. Ofgem will then work towards implementing the SPAA and relevant licence condition(s) by the week commencing 24 May 2004, in readiness for the RGMA Go-live week commencing 12 July 2004. Further details are provided in Chapter 7.

¹ Details available on the Ofgem website at: www.ofgem.gov.uk/ofgem/work/index.jsp?section=/areasofwork/meteringrgma

Table of contents

| 1. Introduction | 1 |
|--|----|
| 2. Provisions of the SPAA Document | 4 |
| 3. The SPAA licence condition(s) | 15 |
| 4. Codes of Practice | 22 |
| 5. GT involvement in SPAA | 24 |
| 6. Governance of metering | 34 |
| 7. Way Forward | 37 |
| Appendix 1 Regulatory Impact Assessment | 45 |
| Appendix 2 Supply Licence Condition (revised draft) | 53 |
| Appendix 3 Gas Transporters Licence Condition | 56 |
| Appendix 4 List of Respondents | 57 |
| Appendix 5 The Supply Point Administration Agreement | 58 |

1. Introduction

- 1.1. In July 2001 the consultation paper 'Gas Supply Market: Change of supplier process and governance arrangements' ("the July document") initially discussed the establishment of a suppliers' agreement. This consultation paper was presented in two parts. Part I discussed the Change of Supplier (CoS) process and possible amendments that would be required as a result of increased competition in metering services. Part II discussed supply market processes and their governance arrangements, and discussed options for change.
- 1.2. Respondents to the July document were overwhelmingly in favour of the third of three options, the establishment of new governance arrangements that take a holistic approach to retail market processes. Respondents noted that this is consistent with market developments, in particular the supplier-hub principle, and would provide greater flexibility going forward. There was also a preference for much of the project management and drafting of the agreement to be undertaken by industry, with Ofgem facilitating rather than leading progress.
- 1.3. The Gas Forum established the Gas Industry Governance Group (GIGG) to progress this work and establish the agreement. The first meeting of the GIGG was held on 14 November 2001, when it was decided to set up a working group to concentrate on drafting the Supply Point Administration Agreement ("SPAA"). The GIGG, which is an open forum, has met regularly to discuss the proposals from the workgroup.
- 1.4. In December 2001 Ofgem published its *Summary of responses and way forward* document², which recognised the need for appropriate governance arrangements, with scope strictly limited through a Standard Licence Condition that would mandate accession.
- 1.5. In June 2003 Ofgem published a *Further Consultation* document³, which included a draft licence condition potentially to be applied to domestic and I&C suppliers, as well as a draft of the SPAA itself. It was further proposed that gas transporter ("GT") involvement and the potential migration of supply point

² 'Gas Supply Market: Governance arrangements, summary or responses and way forward' December 2001.

³ 'Gas Retail Governance, further consultation' June 2003.

administration ("SPA") provisions from Network Codes should form a second

phase of development, to follow the RGMA go-live, which at that time was

anticipated to be November 2003.

Structure of this document

1.6. The structure of this document generally follows that of the June 2003

document. Each chapter summarises responses to the issues raised in the

relevant chapter of the June 2003 document, details Ofgem's views and provides

comments on the way forward. Finally, Chapter 7 sets out a revised timetable

for the completion of further work necessary for full implementation of the

SPAA.

1.7. A summary regulatory impact assessment on the involvement of GTs is attached

at Appendix 1, rationale for the SPAA itself having been covered in previous

consultation documents. An amended draft Suppliers licence condition and a

draft Gas Transporters licence condition are attached as Appendix 2 and 3

respectively. Appendix 4 provides a list of respondents to the June 2003

document. The amended draft SPAA is attached as Appendix 5.

Views invited

1.8. If you wish to comment, it would be helpful if responses could be submitted to

Ofgem by 2 April 2004. Whilst it is open to respondents to mark all or part of

their responses as confidential, Ofgem would prefer as far as possible that

responses are provided in a form that can be placed in Ofgem's library and

published on our web-site. If only part of a response is to be treated as

confidential, please provide this as clearly marked annex to the document

response. Please write to, or email:

Nick Simpson

Director, Modifications

Office of Gas & Electricity Markets

9 Millbank

London

SW1P 3GE

E-mail: industrycodes@ofgem.gov.uk

- 1.9. If you have any questions about the issues raised in this document please contact either Bryony Sheldon (020 7901 7174, industrycodes@ofgem.gov.uk), Jonathan Dixon (020 7901 7354, industrycodes@ofgem.gov.uk) or Nigel Nash (020 7901 7065), nigel.nash@ofgem.gov.uk who will be happy to discuss them with you.
- 1.10. In addition to considering written responses, Ofgem will be hosting an open meeting on 22 March 2004 to seek views in respect of this document and, in particular, the way forward up to and beyond the RGMA Go-live. In view of limited places, if you would like to attend this meeting, please contact Bryony Sheldon (020 7901 7174 or industrycodes@ofgem.gov.uk) by 17 March 2004.

2. Provisions of the SPAA Document

2.1. Chapter 5 of the June 2003 document discussed the key aspects of the SPAA document. This chapter summarises the responses to issues raised and Ofgem's current view.

Principles of governance

2.2. The document outlined six principles of good governance that the SPAA should conform to, namely; effectiveness, efficiency, transparency, participation, accountability and consistency. The drafting of the agreement was then described within the context of those principles.

Respondents' views

2.3. Respondents generally endorsed the principles of good governance as proposed by Ofgem, and were broadly content that the SPAA conforms to them. One respondent compared the principles with those for good regulation (as developed by the Regulatory Impact Unit), and suggested that proportionality and targeting should also be on the list. Another respondent suggested a further principle based on 'value for money'.

- 2.4. Ofgem welcomes respondents' support for the principles of good governance outlined in the document. Ofgem also supports the principles of good regulation, and considers these when progressing regulatory initiatives.
- 2.5. Whilst Ofgem notes support for the inclusion of proportionality and targeting as principles of good governance, it considers that the main precepts are already encompassed under the principles of efficiency and participation. For instance, Ofgem stated in the June 2003 document that mandatory provisions should be kept to the minimum necessary, with this status only being conferred when it can be shown that voluntary or elective status is not sufficient to ensure the objective of that provision. Additionally, Ofgem considers that 'value for money' is also addressed by the principle of efficiency.

10 day consultation period

2.6. The document highlighted that the current drafting of the SPAA provides a minimum of 10 working days consultation prior to voting on change proposals. This is in line with the 10 days provided under MRA, but contrasts with National Grid Transco's ("NGT") Network Code (ordinarily 15 days). Views were invited on whether the proposed consultation period of 10 days is appropriate.

Respondents' views

- 2.7. Nine respondents expressed a preference for the consultation period remaining at 15 days or more. Comments included that this would put the SPAA processes in line with those of NGT's Network Code. One respondent stated that the only benefit of a 10 day consultation period was alignment with electricity.
- 2.8. Four respondents commented that a 10 day period was appropriate or at least sufficient, though one pointed out that this is merely a minimum period and that in reality it was likely to be much longer especially if joint impact assessments with the MRA are needed. However, it was also noted that the MRA is currently considering moving to 20 days consultation and that it may be prudent for SPAA to do the same.

- 2.9. As indicated in the document, Ofgem believes that decision making must balance the need for timely resolution and thorough consideration of issues. A longer consultation period may also allow a fuller participation from those parties with limited resources. Given that the SPAA has provisions allowing for an urgent change, and in order to reflect responses, the SPAA drafting has been amended to reflect a 15 day consultation period. However, this would not preclude the consultation period itself being subject to a change proposal, should experience later show it to be inappropriate.
- 2.10. Furthermore, in instances where a proposal relates to a mandatory provision, it will be referred to Ofgem for final approval. Where Ofgem considers that the proposal would have a significant impact upon persons or activities (whether they are parties to SPAA or not) it may consider it appropriate to conduct a Regulatory Impact Assessment (RIA) to determine the extent of the impact and

whether or not there are net benefits to the proposal. Such an RIA would be consulted upon, with the minimum period in which to respond being 28 days.

Criteria for urgent status

2.11. It was suggested that the title of clause 9.19 of 'Emergencies', for circumstances where the SPAA EC decides that a change proposal is of an urgent nature, be changed. The document also suggested that it may be appropriate for suppliers to consider developing urgency criteria to provide guidance for the SPAA EC when deciding whether to shorten the published timescales for consultation on change proposals.

Respondents' views

2.12. The only respondent to comment specifically on the title of clause 9.19 agreed with the proposed change. Of those who commented on urgency criteria, ten supported the development of clearly defined criteria. Two of these suggested that this should be one of the initial tasks of the SPAA Executive Committee, and one suggested that they should be similar to those used for NGT's Network Code.

Ofgem's view

2.13. Ofgem considers that urgency criteria should be developed by the SPAA EC. Whilst Ofgem recognises that urgent changes to the SPAA will be driven more by commercial considerations, it believes the current criteria⁴ for granting urgency to a change under existing codes would provide a useful starting point.

Introduction of schedules

2.14. The document identified two potential methods for the introduction of schedules into the SPAA. Firstly, that when introduced all new schedules would have voluntary status, which could then be amended to elective or mandatory status at a later stage, via a further change proposal. Alternatively, that a change proposal to introduce a new schedule would also set out its intended status, together with appropriate rationale.

⁴ See: www.ofgem.gov.uk/temp/ofgem/cache/cmsattach/2752 Urgency%20Criteria.pdf

Respondents' views

2.15. Respondent's views on this issue were evenly balanced – around half of those who expressed a view thought that a two-stage process should be adopted, as this would allow schedules to be tested before becoming mandatory. The remainder, who supported the status of a prospective schedule being considered at the time of introduction, believed that to do otherwise would be inefficient.

- 2.16. It is recognised that either approach to the introduction of schedules has relative merits and disadvantages.
- 2.17. Ofgem has previously indicated that schedules should only be classified as mandatory if it is demonstrated that they are essential to allow the efficient and effective inter-operation of parties to the agreement. The introduction of schedules on a voluntary basis, to be later amended if necessary, could allow for the development of a clear weight of evidence that greater enforcement is required. However, such a two-staged approach could prolong problems where voluntary arrangements are patently insufficient. Also, where provisions are already in existence, such as those currently in the Domestic Code of Practice ("DCoP"), relevant experience may indicate that mandatory status is warranted.
- 2.18. Under the existing Network Code arrangements a modification proposal is either wholly accepted or wholly rejected. However the varying status levels and associated governance of SPAA provisions seems to offer a greater degree of flexibility and discretion than the single-approach Network Code. For instance, the rejection by Ofgem of a proposal to introduce a mandatory schedule would not necessarily be the end of a process, and would not preclude the change from being introduced on a voluntary or elective basis. In fact, given that the proposal would have achieved 65% or more approval before even reaching Ofgem, adoption of the change on a voluntary basis is a more likely outcome than total rejection.
- 2.19. Ofgem would therefore favour the preferred status of a provision to form part of the change proposal, to be commented on by SPAA parties along with the details of the change itself.

Consumer representation

2.20. The document suggested that energywatch should have access to relevant documentation and be able to raise change proposals, but should not have the ability to vote or raise appeals.

Respondents' views

- 2.21. Views on this point were mixed, with twelve respondents expressing either full or partial support for Ofgem's proposals, and three others opposing any involvement of a consumer representative body. A large number of respondents agreed that the customer representative should not have the right to vote or raise appeals on change proposals.
- 2.22. One respondent stated that it had initially been against consumer representation in SPAA, but was now in favour. There was general support for energywatch being the appropriate body. energywatch stated that although it would have preferred full voting and modification rights, it would be content with the ability to raise change proposals and attend meetings, including those of the Executive Committee.
- 2.23. One respondent, whilst agreeing that current levels of representation should not be reduced, suggested that involvement in the SPAA could be decided on a schedule by schedule basis. Two other respondents suggested that the role should be equivalent to energywatch's role in NGT's Network Code.
- 2.24. Four respondents indicated that while they support energywatch having access to documents and the ability to make representations, they do not agree that it should be able to raise change proposals. One respondent went on to say that views expressed by energywatch should not be accorded equal weight to those of signatories.

Ofgem's view

2.25. Ofgem remains of the view that its initial proposal represents an appropriate level of consumer representation within the SPAA, and that when carrying out this role the views of the consumer representative should be given full consideration. The drafting of the SPAA has been amended to reflect the role of energywatch.

Appeals

2.26. The document questioned whether Ofgem should determine an appeal on whether a particular change proposal unfairly prejudices the interests of a supplier at the same time as deciding upon the change itself, rather than following separate procedures.

Respondents' views

- 2.27. Of the eleven respondents who commented, ten stated that there should be a separate process. Of these, one suggested that unfair prejudice should be the only ground for appeal, whilst another suggested that there should be a fifth appeal criterion of 'unjustified cost'.
- 2.28. One respondent, whilst supporting a separate process, indicated that they would also expect issues of unfair prejudice to form a part of Ofgem's decision on change proposals. A further respondent indicated that this should form a part of the decision making process, but went on to state that there would be little benefit in limiting a party's right to appeal.

- 2.29. In raising this issue, Ofgem's intended to highlight the potential for it to be required to make two separate (and potentially conflicting) decisions, rather than to question the right of appeal, which Ofgem fully supports. In advance of SPAA inception Ofgem intends to develop internal procedures on how SPAA appeals will be determined, either independently of, or concurrent with, an Ofgem decision on a change proposal. (It is expected that as far as possible the SPAA and MRA appeals processes will be the same). As such, Ofgem considers that the existing drafting should remain unchanged.
- 2.30. Whilst the suggested fifth criterion of 'unjustified cost' as grounds for an appeal seems reasonable, such considerations already form a fundamental aspect of the decision making process and will be further strengthened by transparent cost-reflective charging.
- 2.31. Ofgem also understands from responses to the consultation, and views received anecdotally, that some I&C suppliers are concerned at the influence of predominantly domestic suppliers who also declare an interest and vote as an

I&C supplier – i.e. pushing through a mandatory change that primarily benefits domestic suppliers.

2.32. Ofgem considers that any such risk is mitigated by the two-tier voting structure, which requires both 65% of the weighted vote and 65% of parties to approve a mandatory change, in addition to the approval of Ofgem. However, in order to provide further safeguards against such an eventuality, Ofgem has extended the existing provisions for appeals, to include class actions. This would allow an individual or group of I&C suppliers to appeal a decision which they consider unfairly prejudices a class of party to which they belong, even if they are not individually discriminated against: for instance, a mandatory change to common systems which only benefits domestic consumers. This provision may also provide an additional level of protection to other categories of SPAA party, such as GTs.

Protected provisions

- 2.33. In addition to those provisions listed in Clause 9, the document indicated that the following SPAA provisions required 'protected' status, so that they may not be amended without Ofgem's prior written consent:
 - Clause 4.2 Additional Parties
 - Clause 5.1 to 5.5 inclusive Mandatory, Elective and Voluntary Schedules
 - Clause 9 in its entirety Change Control
 - Clause 14 in its entirety Derogations

Respondents' views

- 2.34. Five respondents offered full support for this proposal, one of which stated that the provisions are crucial for governance purposes. One respondent, whilst offering support, queried whether it is appropriate for elective schedules (Clause 5.5) to be afforded protected status, and a further respondent suggested that only clause 14 warrants protection, as the others are covered elsewhere.
- 2.35. Two respondents proposed that in keeping with moves to light touch regulation, Ofgem should have to make the case for every additional provision that it wishes

to include in this category. One respondent indicated that whilst it understands Ofgem's interest, it believes that the provisions listed probably do not require protected status, and one respondent stated that no additional provisions should have protected status.

- 2.36. Clause 4.2 relates to the time in which the SPAA EC shall notify a new party to the SPAA and the Authority that either the party will be admitted, or what is required to facilitate this. This provision will be of particular relevance where accession to the SPAA is the subject of a licence obligation, in ensuring the ability of market entrants to accede and participate in a timely manner. In view of this, Ofgem maintains the view that this clause should be afforded 'protected' status.
- 2.37. Clause 5.1 relates to the binding nature of the SPAA. Ofgem notes that a licence condition could require parties to comply with the provisions of the SPAA. However, to the extent that a provision specified that the agreement (or parts thereof) was not binding on some or all parties, then the existence of a licence condition would not be material. Therefore, in order to preserve the intended effect of the licence condition, Ofgem considers that this provision should have 'protected' status.
- 2.38. Clauses 5.2 to 5.5 inclusive relate to mandatory schedules (amended Clause 5.5 inserted to reflect GT participation). Whilst Ofgem approval is required to introduce mandatory status under Clause 9, these provisions detail the effect of that status. Given its involvement in establishing mandatory status, which will determine the behaviour of participants in the market, Ofgem considers that these provisions should have 'protected' status.
- 2.39. Following consideration of responses, Ofgem agrees that Clause 5.6 (previously Clause 5.5) relating to elective schedules does not require 'protected' status.
- 2.40. Clause 9 relates to change control. Ofgem considers the ability of a party to effect change to be at the heart of effective market participation. In this respect, the methods by which a change may be made to the SPAA, including the ability to appeal, are key to the agreement. As such, while Ofgem would not wish to dictate the specific drafting of the change control provisions, it would be

- concerned to ensure that an adequate process exists. Ofgem therefore considers that this clause should be afforded 'protected' status.
- 2.41. Clause 14 sets out procedures for the grant of derogations, which are a potential means of lifting mandatory obligations. As previously indicated, Ofgem will only agree to obligations being classified as mandatory if it is demonstrated that they are essential to allow the efficient and effective interoperation of parties to the agreement. Given their potentially detrimental impact, Ofgem considers it appropriate that clause 14 be afforded protected status, so that the grant of derogations may not be made unduly permissive.

Voting thresholds

2.42. The document outlined the current voting arrangements, whereby the support of 65% of the votes cast (based on MPRNs) would be required to pass a change proposal, i.e. parties not exercising their vote would be taken out of the count before the 65% threshold is assessed. It was questioned whether voting should be by reference to the percentage of cast, as above, or by reference to the percentage of votes capable of being cast.

Respondents' views

- 2.43. Views on this issue were diverse. Seven respondents offered support for voting to be by reference to the percentage of votes cast, one offered qualified support, and four endorsed voting by reference to the percentage of votes capable of being cast.
- 2.44. Of those who supported voting by reference to the percentage of votes cast, three respondents suggested that sufficient safeguards exist to mitigate problems of voter apathy, including the ability to appeal and Ofgem's role in approving mandatory changes. The balance to be struck between limiting undue influence and not stifling change was highlighted, and one respondent stressed the significant resource requirements of participation.
- 2.45. Whilst indicating contentment with the current drafting, one respondent suggested that the voting thresholds should be revised in line with those for

Collective Licence Modifications⁵. A further respondent suggested that if parties believe that the change control is fair and equitable and that sufficient time is allowed, then voting should be on the basis of percentage cast. However, if the normal change control can be bypassed or shortened, then voting should be by reference to percentage capable of voting.

- 2.46. Those respondents who supported voting by reference to the percentage of votes capable of being cast expressed concern that changes could otherwise go through with little confirmed support, particularly where abstentions are discounted. A further respondent indicated preference for a rule that silence is acquiescence.
- 2.47. Other comments included that, for I&C constituents, voting should be on the basis of size, rather than according to numbers of MPRNs. Concern was also raised that suppliers with both domestic and I&C portfolios may be able to influence decisions in the I&C section, based on the requirements of domestic businesses.

Ofgem's view

- 2.48. The June document highlighted the need for voting arrangements to strike a balance between preventing undue levels of influence and not stifling change. It is of note that proposals to modify the NGT Network Code rarely receive more than a dozen representations. Proposals to modify iGT Network Codes generally receive only 2 or 3. If the threshold referred to the percentage of votes capable of being cast, rather than actually cast, there is therefore a stronger possibility that change would be stifled by either apathy or ambivalence.
- 2.49. As previously discussed, Ofgem are aware of the concerns of some I&C suppliers that the effectiveness of their constituency may, in effect, be diluted by suppliers active in both markets declaring and interest in the I&C vote. However, given that there are more I&C suppliers than Domestic suppliers and that a mandatory change requires acceptance by both 65% of MPRN's and 65% of parties to

_

⁵ The CLM rules provide that a proposed licence modification may be made where no more than 20% of relevant licensees, firstly by number of licences and secondly by number of MPRNs, register their objection. Further details are provided in the Ofgem document: Guidance on modifying the standard licence conditions of gas and electricity licences, September 2003

proceed, the best means of ensuring robust and equitable decision making would be for all parties to be actively engaged in the process. The safeguards of an appeals process, as mentioned in paragraph 2.32, would therefore be rightly relegated to being a last resort.

Derogations

2.50. The document sought comments on the extent of Ofgem's role, if any, in the granting of derogations from obligations imposed through the SPAA.

Respondents' views

2.51. There was a broad range of views on the appropriate level of Ofgem involvement in this process. Five respondents supported an increased level above that presently proposed. Of these, two suggested that Ofgem should be responsible for granting derogations and two proposed that Ofgem should review decisions to grant them. A further three respondents supported Ofgem's role as currently drafted, and two stated that Ofgem should not have any involvement beyond its role in determining appeals.

Ofgem's view

2.52. With Clause 14 being afforded 'protected status', (as discussed at paragraph 2.41), the existence of a robust procedure for the granting of derogations is assured. Ofgem considers that the current drafting provides the appropriate level of its involvement in the granting of derogations, and will make representations and determine any appeal as necessary.

3. The SPAA licence condition(s)

Licensing of suppliers

- 3.1. During the development of SPAA there was a general consensus that the gas suppliers' licence should contain an obligation to sign and comply with the SPAA. Suppliers considered that this was necessary to ensure that all suppliers would be bound by the requirements of the SPAA, which in turn would enable the effective operation of the gas supply market, in particular those aspects which require a degree of interaction between suppliers or their agents. It would also increase the viability of, and confidence of investing in, automated systems for dealing with agreed industry processes.
- 3.2. In line with the views expressed by I&C supplier representatives both at GIGG and the SPAA seminar the July document proposed that the licence condition would apply to both domestic and I&C suppliers, within section B (General Obligations) of the licence. However, it was recognised that there are currently more incentives on domestic suppliers to accept a licence condition, given that they are subject to the metering licence obligations (in particular Standard Licence Condition 34) that have formed the initial focus of the agreement, through provision of governance arrangements to support RGMA.
- 3.3. Mandating the use of certain metering processes through a SPAA licence condition could have the effect of extending regulation into an area of I&C supply which is currently unregulated. This concern could be addressed if, for example, metering related provisions that would ordinarily be mandated by SPAA for domestic suppliers are voluntary for I&C suppliers.
- 3.4. The June 2003 document queried whether a licence condition containing an obligation to sign and comply with the SPAA should be placed on both domestic and I&C suppliers.

Respondents' views

3.5. A large number of respondents commented on this issue. Of those, six thought that a licence condition should placed upon both domestic and I&C suppliers from the outset. A further respondent indicated preference for a licence

- obligation to be on both, but suggested that it may be pragmatic for it only to apply to domestic suppliers initially.
- 3.6. Whilst there was general acceptance that there should be a domestic supply licence obligation to sign and comply with the SPAA, ten respondents stated they did not support the introduction of a licence condition on I&C suppliers from the outset. Of these, many thought that a licence condition would be an increase in regulation, and that compliance with the SPAA could be achieved through voluntary accession, though some queried what the competition law implications of this might be.
- 3.7. The key concern of I&C suppliers, expressed both in response to the June 2003 paper and subsequent discussions with Ofgem, is that reforms to the customer transfer process will be driven by domestic suppliers which are primarily of benefit to their market and would be disproportionately costly to the I&C market, where despite volumes being high, customer numbers are low.

Ofgem's view

- 3.8. Ofgem recognises that as the SPAA will initially focus on metering, which is not currently an I&C supply licence obligation, I&C suppliers may perceive a licence obligation to comply with the SPAA (and therefore the RGMA Baseline) as an extension of regulation. However, as the RGMA Baseline will in effect set the standard upon which metering operators such as Meter Asset Managers are likely to model their services, it will be crucial that it remains relevant and appropriate to the I&C market, and that I&C suppliers continue to have a role in its future maintenance and development.
- 3.9. In addition, other than the metering arrangements mentioned above, much of the SPAA may not be directly applicable to I&C suppliers unless, or until, SPA provisions are migrated from Network Code(s). Ofgem therefore does not intend to ask I&C suppliers to accept a licence condition mandating accession to the SPAA.

Accession to SPAA

3.10. Ofgem is off the view that it would be preferable for I&C suppliers to accede to SPAA. This is in part because the RGMA Baseline is as important to I&C

suppliers as to domestic and both sectors have been instrumental in its development. More generally, voluntary accession will enfranchise I&C suppliers in the development of industry processes which will ultimately have a direct or indirect impact upon them. For example, although I&C suppliers are under no licence obligation to provide metering services, they invariably do so as part of the supply contract. Unless the supplier wishes to obtain bespoke metering services, with the inevitable cost implications, they will be procuring services that have been shaped in compliance with the RGMA Baseline.

3.11. However, as discussed in the June 2003 consultation document, much of the scope of SPA will remain voluntary, or only applicable to the domestic market. Therefore the voluntary accession of I&C suppliers to SPAA will not detrimentally impact its due functioning, though it may have some implications. Firstly, 100% participation will be required in order to maintain the current levels of interoperability for some procedures, particularly those relating to the transfer of consumers. Without mandating accession through a licence condition this degree of participation cannot be guaranteed, though it is hoped there are sufficient incentives for all I&C suppliers to accede on a voluntary basis. Secondly, licence enforcement will not be an option for ensuring direct compliance with the relevant provisions of the SPAA.

Funding

- 3.12. The treatment of the recovery of costs within the initial draft of the SPAA made no distinction between Domestic and I&C suppliers, with both categories making a contribution to funding based on the number of MPRNs held. However, unlike voting privileges, this has not been capped at 20%. With the SPAA budget for the first year being capped at £200,000 (around 0.01p per MPRN) this would mean the expected contribution from an I&C supplier with 10,000 MPRN's would be only £100 per year. The budget for all subsequent years will be subject to the approval of parties (clause 8.2) using the same resolution and voting mechanism as for other change proposals.
- 3.13. Another major concern of I&C suppliers was that domestic suppliers would push for major reforms to the SPA arrangements, particularly suited to domestic market. Whilst the establishment of the SPAA should facilitate all suppliers

being in a better position to influence the nature of data services offered to them, it in no way prescribes the outcome.

3.14. Furthermore, Ofgem has already stated in its papers on the DN sales proposals that it does not envisage revisiting the price controls during the current period. The level and nature of SPA charges can therefore be expected to remain unaltered until 2007.

Drafting of the proposed licence condition

3.15. The June 2003 document sought comments on a draft licence condition appended to it, intended to mandate accession to, and compliance with, the SPAA.

Respondents' views

- 3.16. Nine respondents made reference to the proposed drafting of the licence condition, though apart from noting general concerns, few made specific comments or suggestions on potential improvements.
- 3.17. Three respondents did express a preference that the licence condition drafting be similar to that for the MRA in the Electricity Supply Licence (Standard Condition 20), which simply states: 'the licensee shall become a party to and thereafter comply with the provisions of the Master Registration Agreement'.

- 3.18. Ofgem shares respondents' view that the licence condition should be simple and straightforward. Indeed, the draft condition is based largely on existing licence obligations with which all potential SPAA parties should be familiar.
- 3.19. Respondents expressed a desire to replicate the simplicity of Standard Condition 20 of the Electricity Supply Licence and its equivalent amended Standard Condition 14(1) of the Electricity Distribution Licence requiring accession to the MRA. However, these conditions are themselves complemented by Standard Condition 37 of the Distribution Licence, which sets out the required form and content of the MRA in more detail. Without the equivalent of Standard Condition 37 in gas, it is necessary to include more details in the proposed condition, and it has always been intended that the scope of the SPAA will to

some extent be limited by the drafting of a licence condition. Given that the SPAA will deal largely with inter-supplier issues and in line with the supplier-hub principle, Ofgem also considers it appropriate that the primary link between the SPAA and the regulatory regime is the suppliers' licence.

- 3.20. Taking into account the views of respondents, Ofgem has sought to amend the draft licence condition where appropriate. A revised draft of this condition is appended to this document as Appendix 2. In summary the changes made are as follows:
 - a relevant objective of promoting efficiency in the implementation and administration of the SPAA has been added.
 - the proposed requirement to consult with energywatch (and/or other bodies designated by the Authority) has been removed, as the rights of the consumer representative are now captured within SPAA itself;
 - the areas previously listed in paragraph 6 of the draft condition have been removed, as being unnecessarily prescriptive;
 - a new requirement to notify the Authority of any modification which is made has been added.

Is the SPAA likely to have anti-competitive effects?

3.21. It was questioned whether the SPAA has, or is likely to have, any anticompetitive effects, especially in relation to small suppliers or new entrants.

Respondents' views

- 3.22. Four respondents commented that the agreement is likely to be pro-competitive. In particular, the exclusion of entry testing was welcomed. One respondent envisaged that the role of Ofgem would ensure that the SPAA does not have anticompetitive effects. A further respondent commented that anti-competitive effects are unlikely, as long as mandatory requirements are kept to a minimum.
- 3.23. Two respondents mentioned that the SPAA has the potential to create barriers to entry. One thought this would result from the application of a liabilities regime,

while the other didn't elucidate further. Three respondents commented that SPAA may favour large suppliers, to the detriment of small suppliers.

Ofgem's view

- 3.24. Ofgem stated in the previous document that it did not consider the SPAA to have anti-competitive effects. One factor leading to this position was the principle that mandatory requirements in the SPAA should be kept to a minimum, and be based on firm objective criteria. In addition, the capping and split constitution systems should negate much of the power of the larger suppliers.
- 3.25. Although Ofgem maintains the above view, it is recognised that SPAA parties will have the ability to influence change, and that the result of changes over time could create anti-competitive effects. Ofgem will therefore monitor the content and application of the agreement, as facilitated by the requirement to notify Ofgem of any modification to the agreement.
- 3.26. The Chapter I prohibition of the Competition Act 1998 does not apply to an agreement to the extent to which it is made in order to comply with a legal requirement. This would include accession to the agreement and compliance with any mandatory provisions that may otherwise attract licence enforcement. However, the scope of the exclusion would not preclude application of the Competition Act 1998 to agreements or behaviour outside of that specifically required by the proposed licence conditions, including voluntary or elective schedules and provisions.
- 3.27. Ofgem would stress that it is for parties to the agreement to satisfy themselves that any actions are compliant with their legal obligations, including those under the Competition Act 1998 and the Gas Act 1986, both at the inception of the SPAA and on an ongoing basis. In this respect I&C suppliers will be in no different position by virtue of their voluntary accession than Domestic suppliers or GT's.

3.28. Ofgem is therefore seeking views on:-

 Its proposal not to ask I&C suppliers to accept a licence condition mandating accession to the SPAA;

| • | how full participation of I&C suppliers could be achieved if accession is not mandatory. |
|---|--|
| | not manuatory. |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |
| | |

4. Codes of Practice

- 4.1. Domestic and I&C suppliers operate voluntary codes of practice that set out procedures for dealing with a number of operational processes that take place between suppliers.
- 4.2. Domestic suppliers have signalled that they will seek to introduce schedules into the SPAA which would replace the key processes currently set out in the Domestic Code of Practice (DCoP). This would make those processes subject to defined change control and voting arrangements and where appropriate allow the establishment of specific procedures as mandatory requirements, backed by licence obligations. This reflects developments in electricity, where under the MRA suppliers have chosen to migrate the governance of certain procedures, such as the resolution of erroneous transfers from voluntary arrangements (MRA Working Practices) to mandatory arrangements (MRA Agreed Practices).

The Domestic Code of Practice

4.3. The June 2003 document queried whether the inclusion within SPAA of schedules such as those outlined would entirely replace the existing DCoP.

Respondents' views

4.4. Respondents were overwhelmingly of the opinion that once the key provisions are migrated, the DCoP will no longer exist. However, as pointed out by one respondent, this is for suppliers themselves to decide. Additional comments included that the inclusion of the DCoP in the SPAA is of secondary importance to governance of the change of supplier process and gas metering arrangements.

Ofgem's view

4.5. Ofgem considers that it is likely that the DCoP will be replaced in its entirety. However, it is for suppliers to determine what should be done once the SPAA is operational.

The Industrial and Commercial Code of Practice

4.6. Given the intention that the DCoP be replaced by schedules within the SPAA, the document questioned whether the same rationale should be applied to the equivalent Industrial and Commercial Code of Practice (ICoP).

Respondents' views

- 4.7. The majority of respondents expressed a strong preference for the ICoP remaining outside of the SPAA, many of whom believed that it should remain under the auspices of the Gas Forum. Whilst indicating preference for the Gas Forum, one respondent commented that all parties must be allowed to participate (given that not all suppliers are currently members of the Gas Forum).
- 4.8. Three respondents did think that aspects of the ICoP would benefit from inclusion within the SPAA, although one noted that this is not urgent and should only be given consideration once the key schedules are operational. Three respondents indicated that a review of the ICoP is necessary, similar to that undertaken for the DCoP.

Ofgem's view

4.9. Whilst the document briefly discussed the ICoP, it was recognised that this would be an issue for suppliers to progress if considered appropriate. Ofgem believes this approach remains correct. Given the strength of feeling expressed by I&C suppliers and the need to target resources, development of a SPAA schedule to replace the ICoP has been removed from the Schedule 5 list of topics to be pursued as a priority by the SPAA EC. However, this does not preclude future inclusion of such a schedule into the SPAA, should I&C suppliers propose it.

5. GT involvement in SPAA

- 5.1. The initial focus of the SPAA has been to provide a robust means of governance for the inter-supplier communications currently managed under the Domestic Codes of Practice (DCoP) and the Review of Gas Metering Arrangements (RGMA) documentation. However, it has from the outset also been envisaged that SPAA may also address some of the more fundamental issues with the provision of SPA services.
- 5.2. Whilst the SPA service currently provided by GTs has been a cornerstone of the competitive supply market to date, there are aspects of its operation which may hinder the further development of that market if they are not addressed. These could be broadly categorised into:
 - the need for consistency across the various networks (lack of consistent between iGTs, and between iGTs and NGT, creates increasing problems in supply markets, a problem that could be intensified by the sale of an NGT Distribution Network ("DN")⁶), and;
 - the need for appropriate incentives to manage the service in accordance with the needs of its customers (in this context, gas suppliers).
- 5.3. The processes which support the customer change of supplier procedure are currently governed via the host GT's Network Code. Although each GT operates in broadly the same way as NGT, the slightest divergence may require the Shipper/Supplier to resort to bespoke, low-tech procedures. This has an obvious cost implication for suppliers, with around 500,000 consumers now connected to independent networks. In many cases, these costs are already being passed through to the end consumer in network-differentiated tariffs.
- 5.4. In providing both transportation services and retail market processes, the GT is subject to conflicting incentives. The perception is that it is suppliers, being non-contracted parties, who suffer from not being able to get the SPA service they want. However, this is not simply a GT behavioural issue, but inextricably linked to the way the service is currently provided, being governed by the

⁶ National Grid Transco – Potential sale of network distribution businesses; July 2003

- Network Code (to which suppliers are not party) and funded by the bundled transportation charge to Shippers.
- 5.5. Since a fixed allowance is made within the price control for the incremental development of SPA functions, NGT in its transportation role has little incentive to invest in such improvements or actively maintain accurate data which is of no interest to it. Furthermore, as from 1 January 2004 iGT's have been subject to a Relative Price Control for activities including the provision of SPA services. The current arrangements therefore have the effect of incentivising Code parties to resist changes which do not directly benefit themselves.
- 5.6. Whilst to date suppliers to consumers on iGT networks may have been able to work around any differences between SPA procedures, using bespoke low-tech solutions, this would not be a practicable option if a DN, having been sold by NGT, were to develop a different SPA system. It has therefore been proposed that as part of a potential DN sale, SPA services would continue to be provided centrally by an agency, to be set up by NGT in facilitation of such a sale. The creation of such an agency, with possibly new relationships and incentives, may create opportunities for it to provide new services. For instance, it has been suggested that the agency could provide services to iGT's as well as NGT and former-NGT DN's.
- 5.7. One of the key drivers for inclusion of GTs as signatories to the SPAA and eventual migration of SPA provisions is that it would enable supplier involvement in the retail processes that affect them directly. This is not currently available under Network Code arrangements and does not form part of the proposals to create a post-DN sale Uniform Network Code.

Should GTs be part of SPAA?

5.8. In the June 2003 document Ofgem welcomed the principle of GT accession, but indicated that at that time it was not yet clear how this would work in practice. Major issues were highlighted such as the interaction of the SPAA with Network Codes (especially prior to, and during, any migration of SPA provisions into the SPAA) and the funding of change. The document invited comments on GT involvement, in particular whether GTs should become party to the SPAA.

Respondents' views

5.9. There was overwhelming support from suppliers for GT involvement in the SPAA, although there was some concern at the amount of work that this may entail. The two GTs who responded indicated that their accession would be dependent on the benefits available to them, with one stating that these are not yet clear.

- 5.10. Whilst the SPAA was developed primarily as an agreement between suppliers, it was envisaged that a second phase of development would allow GTs to accede to the agreement, thus facilitating a migration of Supply Point Administration (SPA) provisions from the existing Network Codes into SPAA. Responses to the June consultation and subsequent developments have led Ofgem to consider that it would be appropriate for GT's to be involved in SPAA from its inception.
- 5.11. GT's have no involvement with the processes set out in the current DCoP. Their involvement must therefore be warranted on the basis of RGMA and other procedures that may eventually be covered by the SPAA, such as SPA services.
- 5.12. With regard to the RGMA, it is apparent that GT's will retain an interest in the provision of metering services, given their function to provide read data for billing and settlement purposes. It is envisaged that the GT's Network Code will continue to prescribe the metering requirements for their network, for instance setting the threshold for consumption above which sites should be daily read, policy on the use of converters etc. It would be beneficial for the industry as a whole for there to be seamless arrangements in this regard, negating the risk of mismatched obligations or 'double jeopardy'. GT involvement in the deliberations of the SPAA and therefore of changes to the RGMA Baseline would facilitate this.
- 5.13. Furthermore, whilst the RGMA Baseline was written largely with the NGT network in mind, it is likely that suppliers will eventually seek to apply the standard procedures to all networks, thereby being able to offer their customers a uniform service. It would be preferable for iGTs to be fully enfranchised in the decisions on changes which may be applied to them, rather than reacting to the Network Code modification proposals.

Timing of GT accession

5.14. Following the discussion on GT involvement, views were sought on the appropriate timing of GT accession to the SPAA. The document highlighted the two options of immediate or phased GT incorporation, with Ofgem at that time advocating a phased approach, but encouraging GTs' active participation in the meantime.

Respondents' views

- 5.15. Opinions were broadly divided on timing, with six respondents supporting immediate accession and five supporting a phased approach. One stated that GTs should ideally be in from the commencement of SPAA, but if this were not possible then a firm date for accession is vital. Two respondents felt that NGT should be involved from the start, with iGTs to follow at a later date, though another stated that all GTs should accede at the same time.
- 5.16. A common concern of those who did not support immediate accession was the significant amount of work that may be necessary to enable GT accession. One respondent suggested that unless GT accession was a second phase, there was a risk to the SPAA ever becoming a reality.

- 5.17. The deferment of RGMA implementation from November 2003 to July 2004 has provided an opportunity to include GT's as parties to the SPAA from the outset, albeit none of the initial schedules may be directly applicable to them. To the extent that changes to the RGMA Baseline may have implications for, or be prompted by, the GT it could be expected that Network Code modifications will be raised subsequent to, and in line with, any changes made through SPAA. GT participation would ensure that the two governance regimes remain aligned, preventing either gaps in governance, or the creation of 'double jeopardy'.
- 5.18. GT participation in SPAA will also establish it as an appropriate forum for Suppliers and GT's to discuss and ultimately decide upon ongoing and future developments to the provision of SPA and other retail services. Ofgem considers that these issues apply equally to iGT networks as to those of Transco, and

considers it appropriate that all iGT's accede to SPAA at the same time, with the same opportunity to shape future developments.

GT licence condition

5.19. The document questioned whether becoming party to and compliance with SPAA should be a condition of the GT licence.

Respondents' views

- 5.20. Of the eleven respondents who provided specific comment on this point, nine (including an iGT) were in favour of a licence condition being placed upon GTs. Some stated that this was necessary to ensure full accession and compliance. One respondent drew comparison to the equivalent obligations on electricity suppliers and distributors to be party to and comply with the MRA. Another went so far as to say they could not support a licence condition being placed upon domestic suppliers unless it also applied to GTs.
- 5.21. One of the respondents who opposed the proposal stated that until the benefits to GTs of accession were clear they would not support the introduction of a licence condition. A second respondent stated that SPA obligations already stem from the GT licence and therefore should not be duplicated by a further licence condition.

Ofgem's view

5.22. Ofgem considers that the rationale for placing a licence condition upon suppliers, i.e. in order to mandate accession to SPAA and ensure compliance with the relevant provisions, also applies to GTs. However, Ofgem is keen to ensure there is no duplication of governance and remains committed to the principal of reducing regulation where appropriate. We would therefore envisage that any obligations placed upon GTs through the proposed licence condition and/or SPAA itself are commensurate with obligations being lifted elsewhere. There may be exceptions to this where GTs accept a new obligation of their own volition. Crucially, the SPAA cannot place upon any party an obligation, which has not been subject to a successful vote (of the relevant constituency) and in the case of a mandatory change, approved by the Authority.

- 5.23. It is not the function of the SPAA to facilitate improvements to the operation of a GT's pipeline system, this is appropriately done via the Network Code governance. SPAA will however facilitate improvements to the retail services that GT's are currently obligated to undertake, which in the short term should lessen the burden to GTs of providing such services, and in the longer term may facilitate the removal of such obligations altogether. This will be the primary benefit of SPAA accession to GT's.
- 5.24. Given the above, together with the relevant objectives of the Network Codes and those of the SPAA, Ofgem consider it will be sufficient for there to be a straightforward condition placed upon GTs, similar to that upon Electricity Suppliers and Distributors to be a party to the MRA. A draft GT licence condition along these lines has been attached to this document as Appendix 3.
- 5.25. Ofgem and NGT have been involved in extensive discussions on the possible governance arrangements that may apply to the UK gas market in the event of a sale of one or more of its DNs. NGT considers that the SPAA offers the potential for a more self governing model for the development of supply competition in the UK gas market for the benefit of customers, and has agreed in principle to accede to the SPAA, subject to a clearer appreciation of the potential risks and costs involved. It is however recognised that the transfer to such a model requires careful consideration of the transitional requirements of the Network Code, licences and the SPAA. Ofgem and NGT are committed to working these through with all the relevant parties.

Funding

- 5.26. The document questioned how the funding of change should be apportioned, and made reference to the final report of the SPAA GT Forum which recommended the following:
 - SPAA services (secretariat) to be funded by suppliers
 - SPAA meeting attendance costs to be met by attendees
 - Funding of SPAA changes to be on a 'gain share' basis.

Respondents' views

- 5.27. Respondents expressed a diverse range of opinions, but only one respondent offered full support for the GT Forum's recommendations. Four opposed 'gain sharing', believing that it would be unworkable and inconsistent with MRA and Network Code arrangements. Two respondents indicated that GTs should meet a proportion of costs, with one suggesting this should be in the same way as under the MRA (i.e. with suppliers funding two-thirds and GTs funding one-third).
- 5.28. Two respondents stated that each party should fund their own costs, as this would encourage efficiency. One suggested that suppliers should pay for everything. Three respondents indicated it is crucial that funds for SPA are unbundled from NGT's price control.

- 5.29. Like the GT Forum, Ofgem considers it appropriate to differentiate methods of funding, broadly between the costs of operating the SPAA and the costs of implementing change. Whilst it would seem appropriate in principle for all SPAA parties to contribute to its operation (i.e. secretariat costs), as drafted its benefits accrue mainly to suppliers, with the benefits to GT's coming later, with the insertion of appropriate schedules.
- 5.30. The SPAA is based to a large extent on the electricity MRA, which is funded jointly by suppliers and distributors, with the former contributing two thirds. The remaining third is divided evenly between the 14 ex-PES's. However, this may not be appropriate in gas, given the huge differences in size between NGT and any (or all) of the independent Gas Transporters.
- 5.31. Ofgem would therefore suggest that if and when GT contribute to the funding of SPAA, it should be on the same per MPRN basis as suppliers' contributions. In any event, given the anticipated total costs of operating SPAA and the £200,000 cap, it is likely that any cost to NGT and other GT's will be de minimis.
- 5.32. As the SPAA evolves it may be possible for the funding arrangements to themselves become more sophisticated. For instance, with experience it may be

- possible to apportion costs on a schedule by schedule basis, recovered only from those parties to whom the schedule applies.
- 5.33. With respect to the funding of change, Ofgem considers that it will be appropriate for each party to bear its own implementation costs. In the case of SPA services, an allowance is already made for incremental changes within the NGT price control, and by association the iGT Relative Price Controls. While SPA remains a price controlled activity of the GT, the treatment of costs and criteria for a change being approved will remain the same.
- 5.34. However, in order to address the concerns over lack of appropriate incentives NGT and Ofgem have agreed to review the funding of the agent SPA services in preparation for the next price control review. Such a review may lead to the establishment of unbundled transparent and cost-reflective charges, perhaps on a transactional basis. The implications of this are that the costs of change would also become visible, and potentially passed through in the form of higher transaction costs or one-off implementation charges. This would also introduce an additional level of financial responsibility upon parties proposing change, and allow for more robust analysis of the relative merits of the proposal.
- 5.35. Some I&C suppliers have been advocating the SPA function being split into I&C and domestic functionality, allowing it to develop to meet the requirements of each market, without placing costs upon parties which do not benefit. A review of SPA funding could facilitate such a move, away from the current one-size-fits-all service, perhaps leading to differentiated offerings.

Changes to allow GT accession

- 5.36. As the SPAA was initially drafted as a supplier-only agreement a significant number of changes to the text were required in order to facilitate GT accession. However, many of these changes are of a straightforward nature such as references to 'supplier' now being used only in connection with that licensed activity, rather than as a proxy for 'party', as previously. More substantially, in facilitation of GT accession a separate GT constituency has been created with associated voting procedures equivalent to those of suppliers.
- 5.37. Membership of the SPAA Executive Committee has also been amended, with two additional seats being created for use by the GTs. In line with existing

provisions for suppliers, one of these seats will be open to GTs with less than one million connected supply points, and the other to a GT with more than one million. In effect this will guarantee a position for NGT and one iGT, though absent further amendments the position of the former would not be guaranteed in the event of a DN sale, which could be expected to create another GT with more than one million connected supply points.

5.38. A revised (marked up) copy of the SPAA is attached to this document as Appendix 5.

Interaction with Network Code(s)

- 5.39. It will be crucial for the effective governance of the gas market for the SPAA and Network Code(s) to work in tandem, being complementary to each other, rather than duplicating provisions or leaving gaps. This will be especially important for the end to end governance of the gas metering market, and to facilitate an orderly migration of SPA provisions.
- 5.40. Absent a formal link between the two documents being created, Ofgem's role in the decision making process will ensure the two documents remain synchronised, i.e. not approving a change to one until it is in a position to make a commensurate decision on the other. However, it would also seem sensible for there to be formal links between the two governance regimes during the early stages of the change process.
- 5.41. First, it will be necessary to ensure that the SPAA remains aligned to the Network Code(s) and vice versa. As well as being good practice, this will avoid the potential situation of a party be placed in a double bind, of being in breach of either the SPAA or Code. With GT's being party to SPAA this could be achieved relatively simply by placing a requirement upon them to keep their own Network Code aligned to relevant developments in SPAA, or to raise a SPAA change in line with a modification to their Code, as appropriate. Alternatively, the Modification Rules could be amended to allow suppliers or a SPAACo representative to raise Network Code Modification proposals for the same purpose. In addition to keeping Network Code(s) aligned with SPAA, this may have the additional benefit of keeping Codes aligned with each other.

- 5.42. Where complementary changes have been raised to both the SPAA and Network Code(s), it would seem sensible to avoid duplication wherever possible. For instance, where a SPAA change proposal requires further development, the SPAA allows that a workgroup may be set up for this specific purpose. Most Network Codes have similar provisions. Therefore a single workgroup could achieve both objectives. Similarly, when the proposal goes out to consultation, it would reduce the burden upon parties if a single representation could, if appropriately marked, serve both SPAA and Code.
- 5.43. Any change to the Network Codes, whether it is removing or inserting provisions, will need to demonstrably facilitate the relevant objectives of the Code, as currently.

Conclusion

5.44. Consideration of responses to its June 2003 consultation, together with recent industry developments, has led Ofgem to conclude that it is appropriate for the SPAA to extend to GT's at its inception, albeit that there may be no provisions immediately applicable to them.

5.45. GT accession will:

- ensure co-ordinated end-to-end governance of metering requirements;
- enfranchise GTs in the decision making process
- facilitate the orderly migration of SPA provisions from Network Code(s) to an industry-wide agreement featuring the key stakeholders (suppliers);
- mitigate some of the risks of a potential DN sale, and;
- facilitate progress towards the lifting of non-core GT obligations to provide retail services.

5.46. Ofgem is therefore seeking views on:-

- Its proposal to introduce a licence condition upon all GTs, mandating accession to the SPAA;
- the proposed GT licence condition (see Appendix 3).

6. Governance of metering

6.1. The need for formal governance to underpin the competitive metering market has been one of the key drivers for the development of the SPAA. This need relates predominantly to the RGMA Baseline and the NGT Metering Contract. However, changes to the license regime will equally impact upon independent GT's. Ofgem would also anticipate that, as competition develops, suppliers will want common metering arrangements for their customers, whether they are connected to a NGT or an independently owned network.

Collective governance of meter asset transfers

- 6.2. The need to govern metering competition, and in particular the provision of a metering asset by a party other than the GT, has been one of the major drivers for the SPAA. Although it is envisaged that the RGMA Baseline will be a schedule to the SPAA, there is currently little or nothing within the Baseline on how the transfer of meters on change of supplier is to be conducted (e.g. the timing and methods of negotiation).
- 6.3. Whilst the SPAA must not become embroiled with the pricing of meters, there may be a case for 'negotiation' procedures to be defined. In the case of electricity, the onus is upon the outgoing supplier to set their terms and conditions. This is not the case in gas. The document asked whether the transfer of the meter asset between suppliers or their agents should be subject to collective governance under the SPAA.

Respondents' views

- 6.4. Respondents generally agreed that this is an area of concern and that some form of guidance and/or governance is needed. Of those who commented, two respondents believed that the SPAA could provide an appropriate mechanism, while four commented that there may be merit in aligning the gas and electricity licence conditions.
- 6.5. A further two respondents commented that such collective governance might make sense in the domestic market, but would not in the I&C market, while another suggested that there should be separate schedules covering domestic

and I&C metering. Respondents were generally agreed that a transfer mechanism, of whatever description, should not include the price of the meter, which should be left to the market to determine.

Ofgem's view

- 6.6. Ofgem has recently issued a consultation paper⁷ which considers the need, or otherwise, to refine the existing licence obligations in line with market developments. In particular, the paper proposes to recognise within the GT licence the direct relationship that has emerged between the GT and supplier for the provision of domestic metering services. This will have inevitable implications for the complementary provisions of the Standard Conditions of the Gas Shippers', Gas Transporters' and Gas Suppliers' Licences.
- 6.7. Once the responses to the above consultation have been considered, Ofgem intends to produce guidance on the various gas metering licence conditions, as they will apply to the emerging competitive market, in order to provide increased regulatory clarity. It is intended to publish draft guidance before the RGMA project is implemented in July 2004.

The relationship of the SPAA with the NGT metering contract

6.8. The June 2003 document detailed the inevitable parallels between the governance provided by SPAA and the NGT metering contract, as the RGMA Baseline will form a schedule of SPAA. The document asked whether the SPAA should have any role in, or influence over, the NGT metering contract, and if so, to what extent.

Respondents' views

6.9. Again, views on this issue were mixed. Several respondents commented that as the RGMA Baseline will be governed by the SPAA, and the NGT contract must comply with the Baseline, there is inevitably a link between the NGT contract and the SPAA. However, several respondents stated specifically that there should be no formal link between the SPAA and the contract, with some

⁷ Competition in Gas Metering Services: Proposed Licence Amendments. December 2003.

preferring the regulatory role of Ofgem and the proposed 'reasonableness' clause (as recently consulted upon) as a means of influencing NGT.

Ofgem's view

6.10. Although metering agents will not be a party to the SPAA, it is anticipated that suppliers will fully discuss proposed changes both internally and with their relevant service providers before voting upon any change to the RGMA Baseline. In this respect NGT will be in no different position to any other service provider, other than discussions taking place in the formal and multilateral setting of the Metering Contract Group ("MCG"). In recognition of this above, Ofgem does not consider it necessary for the relationship between the RGMA Baseline and the NGT contract to be formally codified within the SPAA itself, though it is apparent that the two must be able to operate effectively in tandem. This is discussed further in Chapter 7.

Other Comments

6.11. A number of respondents expressed concern at the idea of liabilities being imposed through the SPAA, with some believing that this was an Ofgem proposal.

Ofgem's view

6.12. Ofgem is aware that the issue of liabilities has previously been raised and discussed by industry participants at GIGG meetings. Whilst Ofgem notes that other industry agreements (for example NGT's Network Code and metering contract) contain liabilities provisions, Ofgem believes that this issue should be entirely for parties to the SPAA to consider.

7. Way Forward

Timetable

7.1. The June consultation document included and sought views on an indicative timetable that aimed to have an SPAA in effect in time for the RGMA go-live, which at that time was envisaged to take place in November 2003. Shortly afterwards, the IMSIF⁸ decided because of system build issues to defer the go-live date until 12 July 2004.

Respondents' views

7.2. Although no respondent stated that the indicative timetable was unachievable, only one commented that it was appropriate. Of those who commented more generally, five suggested that Ofgem should allow more time for development of the SPAA, given the slippage to the RGMA timetable. Two respondents indicated that SPAA could be redrafted to involve other parties ahead of RGMA go-live.

Ofgem's view

- 7.3. As the key driver for SPAA is metering competition, the deferment of the RGMA go-live made the indicative timetable at best unnecessarily ambitious and at worst untenable. It did however give an opportunity for many of the outstanding issues identified in responses to the June paper to be given further consideration, prior to the implementation of the SPAA. This paper seeks to address those issues.
- 7.4. Ofgem is aware that a significant amount of work remains to be done prior to the RGMA go-live, not least of all the successful completion of all associated systems build and testing. Work is also ongoing under the auspices of the Gas Forum on the appointment of the successful tender applicant to provide services on behalf of the SPAA Company. Other areas of further work are set out below, together with an indicative timetable for their completion.

Establishing the Supply Point Administration Agreement

- 7.5. In general, very few comments from the June 2003 consultation related on the drafting of the SPAA itself, focusing instead on the scope and extent of participation in the agreement. The SPAA document has already been amended to reflect those comments, and to incorporate GT's as a party.
- 7.6. It remains Ofgem's intention to designate the final version of the SPAA document at the same time as formally consulting on the introduction of the new licence conditions. This will hopefully give licensee's confidence in what they are being asked to sign up to. Once the document has been designated neither Ofgem nor the GIGG⁹ will be able to make further changes, from that point on any change must follow a proposal be raised by a party, and progressed through the change control governance set out in SPAA itself.

The SPAA licence condition(s)

7.7. Following consideration of responses to this consultation and completion of any drafting amendments to ensure that both the SPAA and the proposed licence conditions are fit for purpose, Ofgem will issue formal consultation under Section 23 of the Gas Act 1986 on introduction of the new licence condition.

RGMA Baseline document

- 7.8. The RGMA sought to facilitate competition by developing standard industry processes and interoperable data transaction formats, thus enabling market participants to communicate effectively in the changed metering market. These processes and data flows are contained in documentation titled "RGMA Processes and Data" ("The RGMA Baseline").
- 7.9. In order to maintain this level of interoperability, market participants will be required to comply with the standard processes. Failure to do so could have a detrimental effect on the quality of metering data that would not only impact upon the execution of meter works, but the customer transfer process, which relies heavily upon such data. To the extent that current governance such as licence conditions or commercial contracts do not ensure compliance, an

⁸ The Industry Metering Separation Implementation Forum

alternative means of governance is required. It is on that basis that the SPAA has been developed, complementing rather than replacing the existing tools of governance.

- 7.10. In addition to issues of compliance, ongoing ownership of the RGMA Baseline must be addressed in order to ensure that it is appropriately maintained and developed, thus ensuring ongoing consistency and interoperability. Ofgem also believes it is important not to replicate the non-price barriers to entry, such as asymmetric access to information on metering requirements, which in part prompted the development of the RGMA project in the first place. An accurate and comprehensible RGMA Baseline must be maintained and made freely available for the benefit of potential new entrants to the gas industry, as well as its incumbents.
- 7.11. Absent further development, it is proposed that the RGMA Baseline documentation will be inserted as a schedule, in its entirety, into the SPAA. Responsibility for ownership and ongoing change control of the RGMA Baseline will therefore reside with SPAA parties, as facilitated by SPAACo. This will ensure the RGMA Baseline remains relevant and provides the appropriate level of guidance on the requirements of the metering market to current and potential participants.
- 7.12. In order for the RGMA Baseline to form part of SPAA, it will be necessary for a SPAA party to raise a change proposal, with the intent of the Baseline becoming a SPAA schedule. Given the development that has gone into the Baseline, both during and since its initial publication, Ofgem consider this to be a robust document with a high level of support by industry parties. We therefore do not foresee any problems with a proposal for it to be inserted as a schedule for ownership and change control purposes achieving the required level of support.
- 7.13. The level of support for the provisions of the Baseline being enforced is less clear. As much of the RGMA Baseline has been designed with flexibility and choice in mind, it would be inappropriate and impractical for it to have mandatory status in its entirety. Ofgem also considers that mandatory requirements within the Baseline (or any other part of the SPAA) should be kept

⁹ Gas Industry Governance Group.

to a minimum, and only conferred where the interoperability between suppliers, particularly in respect of the processes associated with change of supplier, would be significantly impacted if they are not complied with. Furthermore, they should not duplicate any regulations which are already in place; for instance many of the data flows set out in the Baseline must already be complied with as a requirement of Licence, or the Gas Meters (Information on Connection and Disconnection) Regulations 1996.

7.14. To the extent that any of the procedures set out in the RGMA Baseline may need a level of enforcement to ensure compliance; this will need to be identified on a step by step basis. Ofgem understands that much work has already been done in this regard by the GIGG and associated sub-groups. It would therefore be helpful if this work was reviewed, in order to ensure it remains aligned to the RGMA Baseline and incorporates any recent developments. Ofgem would envisage such a review being conducted under the auspices of GIGG, possibly during April, in readiness for the implementation of the SPAA in May.

NGT metering contract

- 7.15. It is envisaged that upon RGMA Go-live the NGT metering services will cease to be provided under the terms of the Network Code and will instead form a separate contract between NGT and Suppliers. Both current and future development of this contact is being undertaken by the MCG. Given NGT's position in the metering market, the relationship between the SPAA and the MCG is likely to crucial, particularly in the period immediately following RGMA implementation when the associated processes are bedding in and lessons being learned.
- 7.16. It is likely that a number of early changes to the RGMA Baseline will be proposed by SPAA parties. As previously discussed, it is anticipated that suppliers will fully discuss proposed changes both internally and with their relevant service providers before voting upon any change to the RGMA Baseline. In this respect, discussions between suppliers and NGT will be facilitated by the MCG.
- 7.17. Particularly during this early period, as a better understanding is acquired, it may also be helpful if change proposals were discussed under the auspices of the SPAA Forum, or an appropriately convened sub-group, prior to being put to a

- vote as a formal change request. This could assist the Change Control Administrator ("CCA") in identifying whether the proposal was sufficiently well developed, properly understood or indeed necessary. In effect, this would replicate the role of the NGT Network Code Panel, but could be on an informal basis and may drop away as and when the CCA is more established.
- 7.18. Such a development period would also allow for more detailed discussion with service providers than may be the case if the matter was put straight to a vote. Once the matter had been sufficiently discussed and suppliers had obtained the views of the service providers, the matter would then be referred back to the SPAA Forum (or appropriate sub-group) for a decision on whether the proposal could proceed to a vote. Ofgem would expect such detailed discussion to be time-bound in order to avoid unnecessarily hindering the decision making process of the SPAA.
- 7.19. Over recent months there has been a great deal of discussion in the MCG on the means by which proposed changes to the contract will be progressed between suppliers and NGT, in particular how agreement will be reached. The MCG has developed proposals in relation to the governance of the contract, though some suppliers remain concerned that given its de facto monopoly and regulated pricing structure NGT does not have the same incentives or flexibility to respond to its customers' demands as a normal commercial entity would.
- 7.20. Further to the proposals developed by the MCG and in response to concerns raised over the future governance of NGT's metering contract, NGT has suggested the introduction of a "reasonableness" provision within the special conditions of its licence. This could, for instance, allow suppliers to refer issues of dispute arising from the metering contracts to Ofgem on the basis that they believed NGT to be acting unreasonably by, for example, failing to adopt changes required by the majority of suppliers. Similarly, NGT could refer issues of dispute to Ofgem in the context that the position taken by suppliers was unreasonable and could inhibit their ability to act reasonably within the context of their licence obligations.
- 7.21. The reasonableness clause is amongst the proposals outlined in Ofgem's recent consultation document 'Competition in Gas Metering Services: Proposed Licence Amendments, December 2003'. Ofgem considers that such a provision

- may provide a mechanism by which the early views of Ofgem can be sought, particularly in order to resolve an impasse. Ofgem expect to publish final proposals on this and other licence amendments by the end of March 2004.
- 7.22. Notwithstanding the above, to the extent that it relates to a mandatory provision of the RGMA Baseline the proposed change would in any case be submitted to Ofgem for its approval. In reaching its decision, Ofgem will have regard to the relevant objectives of the SPAA and its own statutory duties. The latter includes the need to secure that licence holders are able to finance the activities which are the subject of obligations imposed upon it.
- 7.23. In this context, Ofgem considers that regardless of the outcome of these discussions, NGT will not have a power of veto over changes proposed to the RGMA Baseline, by virtue of refusing to make the necessary changes to its metering contract. The SPAA and the schedules contained within it are for the use of the whole gas industry and one party's reluctance or inability to adapt to change cannot be allowed to prohibit others from innovating.
- 7.24. If considered appropriate, Ofgem is willing to work together with the GIGG and MCG to develop detailed procedures for the interaction between the SPAA and NGT contract, ahead of the RGMA Go-live.

Domestic Code of Practice

7.25. Ofgem is aware that a significant amount of work has already gone into readying the procedures currently captured under the DCoP for eventual migration into the SPAA. Details were provided in the June 2003 document. It may therefore be a relatively straightforward task for a Domestic Supplier (being a party to the DCOP and the SPAA) to raise a change request to the effect of inserting the DCoP as it currently stands into the SPAA and a new schedule.

Supply Point Administration

7.26. As part of its programme of work to develop proposals relating to the regulatory, commercial and operational arrangements that would support a sale of one or more NGT DNs, Ofgem has set up industry workgroups to look at the key areas. One of these will focus on the proposal to develop an Agency, which would assume certain functions currently provided by NGT.

- 7.27. The objective of this group is to develop a proposal on the functions, funding, ownership and governance of the Agency. The workgroup has particular regard to the manner in which supply point administration is carried out by the Agent, though it will also consider other potential roles.
- 7.28. It is possible that the governance of agency services may ultimately evolve and be configured to reflect the different requirements of the I&C and Domestic sectors of the gas retail market, and of the wholesale market. The proposed Uniform Network Code may therefore continue to govern the balancing and settlement activities of the Agency, whilst SPA services could be governed by the SPAA. Further details are provided in the document: 'National Grid Transco Potential sale of network distribution businesses, Next steps. December 2003.
- 7.29. Consideration of the potential migration of SPA provisions from the Network Code(s) to the SPAA will remain a post-RGMA second phase of development, and it will be helpful if SPAA parties are mindful of the DN sales proposals when developing the SPAA, and vice versa.

Conclusion

- 7.30. As detailed above, there are several pieces of work to be completed in order for the SPAA to be fully effective. Some of these pieces of work, such as designation of the final SPAA and the introduction of associated licence conditions are being undertaken by Ofgem, whereas other such as the introduction of schedules must be undertaken by parties themselves. Ofgem therefore intends to work closely with industry parties, in particular through the GIGG, in order to develop and deliver the programme of work.
- 7.31. In addition to considering written responses, Ofgem will be hosting an open meeting on 22 March 2004 to seek views in respect of this document and, in particular, the way forward up to and beyond the RGMA Go-live. To this end, following a presentation and discussion on Ofgem's proposals the meeting will handover to the GIGG Chairman and thereafter constitute a meeting of the GIGG. In view of limited places, if you would like to attend this meeting, please contact Bryony Sheldon on 020 7901 7174 or industrycodes@ofgem.gov.uk by 17 March 2004.

7.32. Revised indicative timetable:

- Ofgem/GIGG host meeting to agree detailed program of further work 22 March 2004;
- Responses to final proposals received 2 April 2004;
- Issue Section 23 notice and publish SPAA w/c 12 April 2004;
- Responses to Section 23 notice w/c 10 May 2004;
- Licence condition(s) and SPAA come into effect w/c 24 May 2004;
- SPAA change proposals raised, with the effect of introducing the DCoP and RGMA Baseline as schedules to the SPAA w/c 31 May 2004;
- RGMA Go-live w/c 12 July 2004.

Appendix 1 Regulatory Impact Assessment

Introduction

1.1 This appendix is Ofgem's regulatory impact assessment of the proposal to introduce new standard licence conditions into the gas suppliers and transporters licences. It follows the rationale of previous consultation documents¹⁰, discussion and consideration of responses to those documents.

Objective

1.2 The purpose of the consultation is to present final proposals for the introduction of standard licence conditions, in facilitation of the implementation of a Supply Point Administration Agreement. The aim of this agreement is to ensure appropriate levels of governance around existing and emergent industry processes, particularly those associated with the transfer of consumers between suppliers.

Overview of key issues

RGMA

- 1.3 Ofgem believe that the introduction of competition into gas metering services will deliver real benefits, initially to suppliers and then, in turn, to consumers, in terms of reduced costs and improved levels of service. However, competition will mean that suppliers will no longer be able to assume that the meter in place is owned by the GT when they take over a site. The supplier will therefore need to make arrangements with the outgoing supplier or their agent to ensure that the meter remains in place.
- 1.4 Whilst the RGMA Baseline documentation provides standard industry-wide processes that will facilitate the above arrangements, existing governance does not ensure these processes are maintained or complied with. Such compliance

¹⁰ 'Gas Supply market: Change of supplier process and governance arrangements' July 2001, 'Gas Supply Market: Governance arrangements, summary of responses and way forward' December 2001, and 'Gas Retail Governance: Further consultation' June 2003.

will need to be ensured, in order to maintain the levels of interoperability necessary for a smooth and efficient change of supplier procedure.

DCoP

1.5 Procedures set out in DCoP are key for suppliers to be able to manage customer transfers effectively. Governance of these arrangements is based on voluntary participation and goodwill for developing and accepting changes. Ofgem agrees with domestic suppliers' views that more robust governance arrangements would give greater confidence to invest in development, which would lead to more efficient management of the change of supplier process and better service to customers.

SPA

- 1.6 A consistent approach to SPA is critical to the effective functioning of the competitive supply market. Consequently, there is a growing need for harmonisation of the services and procedures provided by the various GT's in respect of Supply Point Administration.
- 1.7 Already there are approaching 0.5 million consumers attached to non-Transco networks. Having to deal with non-standard systems contributes to the administration costs of providing a service to consumers on independent GT networks being greater than those on Transco's network. This has already led some suppliers to introduce network-differentiated tariffs, with consumers on non-Transco networks paying a premium of up to £40 per annum.
- 1.8 NGT's proposal to sell one or more of its gas distribution networks may have implications for the continued delivery of SPA and other services. Whilst recognising that the development of an agency to provide these services is a pragmatic approach, Ofgem has highlighted¹¹ the need to ensure that the proposed agency is appropriately governed.
- 1.9 In both the case of iGTs and the agency, the inclusion of GT signatories within SPAA, facilitating the eventual migration of SPA provisions from the various

¹¹ 'National Grid Transco – Potential sale of network distribution businesses', July 2003, and 'National Grid Transco – Potential sale of network distribution businesses', December 2003

Network Codes into a single agreement, would enable involvement of users of the services. This should, to a degree, facilitate harmonisation of processes and systems, which will improve the overall efficiency and reduce costs of the transfer process.

Options

- 1.10 Previous consultations on the SPAA have already determined the need for new arrangements to be put in place to govern the procedures contained within the RGMA Baseline. Domestic suppliers have also indicated their desire that the new arrangements incorporate those procedures that are currently contained within the voluntary codes of practice. Furthermore, the potential to extend such arrangements to incorporate governance of the proposed agency role has been recognised.
- 1.11 Ofgem has considered a number of options for addressing the issue of appropriate governance of the gas retail market. These were initially set out in the 'Gas Supply market: Change of supplier process and governance arrangements' July 2001 consultation document.
- 1.12 The first option would be to move away from direct regulation through the requirement on GTs to provide a SPA service, instead relying on non-sector specific consumer protection legislation and industry self-regulation. Whilst Ofgem will, as a principle, withdraw from prescriptive regulation wherever appropriate, this option is only viable if the effectiveness of competition is such that customers are protected by market forces and the licence condition was no longer required. As things stand, there is no competition for the provision of SPA services and they will continue to be funded via transportation charges for the immediate future. It is therefore appropriate that regulation provides a proxy for competition in placing the appropriate incentives upon the provider of those services.
- 1.13 The second option would be to take no action to change the existing arrangements, thereby maintaining the status quo. This has already been discounted as a viable option by the results of previous consultations, as it would do nothing to address existing problems with gas retail governance

- already identified, or those that will be created with the advent of metering competition.
- 1.14 The third option, which formed the basis of Ofgem's June 2003 document and enjoyed a high level of support, albeit in some cases qualified, would be to impose obligations solely upon suppliers to create new governance arrangements. Generally however, even those in support of this option saw it only as an interim solution, prior to GT's acceding to SPAA as a second phase of development. Without GT's as party, it was recognised that there may be some disjoint between the provisions of the SPAA and those of Network Code(s), especially in the context of metering. It was also recognised that a supplier-only agreement would be limited in scope and would not allow, for example, the orderly migration of SPA provisions from Network Code(s).
- 1.15 The fourth option would be to extend the participation in SPAA to both suppliers and GT's, thus ensuring a co-ordinated approach to gas governance between the SPAA and Network Codes, and facilitating the migration SPA provisions between those documents. This is direction suggested by responses to the June 2003 consultation paper and forms the essence of Ofgem's final proposals contained within this document.

Risks and unintended consequences

- 1.16 There are risks associated with each of the options identified. These can be categorised as follows:
 - risk of consumer detriment, e.g. inconvenience and distress caused by a problematic transfer;
 - risk of unnecessary costs being borne by suppliers; and
 - the risk that competition will be adversely affected by the impact on either GTs, suppliers or consumers.

Competition

1.17 The SPAA provides a new tool of governance for the retail gas industry. It is important that regulation strikes the right balance between positively influencing the behaviour suppliers and their agents and not being unduly restrictive.

Disproportionate regulation would not only create inefficiencies through the burden placed upon parties, but could have a detrimental impact upon competition, raising barriers to entry. Ofgem believes the SPAA will achieve this right balance, being in large part a self-regulated agreement, whilst retaining a clear role for Ofgem.

1.18 Whilst metering competition introduces new opportunities for suppliers and consumers, there is inevitably an additional level of complexity when it comes to transferring a consumer with non-standard metering arrangements. If consumer confidence is undermined, for example through a poor experience of a transfer, having to have their meter prematurely changed, or simply as a result of negative media coverage, then it is possible that consumers will be deterred from switching supplier. Equally if regulation in this area is too onerous or prescriptive, then suppliers may bear unnecessary and inefficient costs, or be constrained from developing innovative procedures and service offerings.

Costs and benefits

- 1.19 Ofgem is aware that it is in the interests of all stakeholders to ensure that excessive costs do not result from the introduction and operation of the SPAA.
- 1.20 In considering the costs and benefits of the proposal to amend and extend the terms of the gas supply and transportation standard licence conditions Ofgem has identified different types of costs and benefits which could be attributable to consumers, suppliers, GTs and other organisations including Ofgem and energywatch.

1.21 Consumers

Costs:

- the potential for additional costs of participation and compliance incurred by suppliers and GTs may eventually be reflected in increased price; and
- the introduction of uncertainty or barriers to entry would reduce the positive impacts of competition.

Benefits:

- the avoidance of the inconvenience (and sometimes distress) caused by unnecessary meter exchanges;
- SPAA can be expected to facilitate a reduction in costs associated with transfers, particular for consumers on non-NGT networks; and
- greater degree of influence over the service that is offered to them, through the involvement of the consumer representative.

1.22 Suppliers

Costs:

- it is not expected that the proposed changes to the licence condition would significantly increase the costs of participating in the market. For example, the SPAA budget for the first year has been capped at £200,000 or roughly 0.01p per MPRN supplied;
- there will be additional costs associated with participating in the SPAA, responding to change proposals, etc. Parties will, to an extent, be able to control their costs by use of proxy voting etc, rather than having representatives attend a meeting in person;
- there may be additional costs associated with the complying with provisions
 of the SPAA, though many of the provisions will be voluntary. The parties
 themselves will determine the degree of compliance required;

Benefits:

- adherence to the agreed procedures will mitigate the risk of problems occurring with a customer transfer and provide greater confidence for the investment in systems and processes;
- suppliers will benefit from improved consumer confidence and their increased willingness to engage in the market, which may also offer longer term opportunities for the cross selling of other domestic or utility services.

1.23 GT's

Costs:

- the costs to GT's of participating in SPAA may eventually be commensurate with those of suppliers, though it is being proposed in this document that the operational costs of SPAA are initially funded by suppliers;
- GT's should bear no additional development or systems costs as a result of
 participating in the SPAA. To the extent any new obligations refer to
 regulated activities, an allowance is already made with the GT's price
 control and it is envisaged that the approval criteria for changes to such
 activities will remain the same, whether prompted by changes to SPAA or
 their Network Code.

Benefits:

- the SPAA has been developed primarily as an agreement to ensure interoperability between suppliers and there agents, particularly during the transfer of consumers. The benefits to GT's will therefore be derived mainly from reducing the burden upon them in providing retail services, effectively acting as an agent for the supplier.
- it is envisaged that the effective operation of the SPAA will allow obligations to be migrated from elsewhere, whether Network Code or even the GT licence, with the eventual aim of removing such obligations entirely.

1.24 Other organisations:

- being largely self-regulating the SPAA will, once implemented, have limited involvement from Ofgem. As provisions are migrated from Network Codes to SPAA, Ofgem will be called upon to make fewer modification related decisions, freeing up resources;
- improved governance and understanding of industry processes can be expected to lead to a reduction in complaints, which would provide a direct benefit to energywatch and other consumer organisations; and

 energywatch will be able to make a more direct contribution to the development and governance of retail gas processes.

1.25 Environment

These proposals will not have an impact on the environment.

Conclusion

1.26 Ofgem was originally of the view that the SPAA should initially be implemented as a supplier only agreement, covering the existing DCoP provisions and RGMA Baseline. However, consideration of responses to its June 2003 consultation together with recent industry developments has led Ofgem to conclude that it is appropriate for the agreement to extend to GT's at its inception, albeit that there may be no provisions immediately applicable to them. In making this proposal Ofgem has balanced the potential costs and benefits from these proposals and concluded that they will have a positive impact upon competition, benefiting suppliers, GT's and consumers. Ofgem invites views on this and other proposals contained in this consultation document.

Appendix 2 Supply Licence Condition (revised draft)

- 1. The licensee shall, in conjunction and co-operation with all other suppliers, prepare, maintain and be a party to a form of agreement to be known as the Supply Point Administration Agreement, as may be designated by the Authority for the purposes of this condition generally, being a document
 - (a) designed to facilitate achievement of the objectives set out in paragraph4; and
 - (b) including the modification procedures required by paragraph 5 and the matters required by paragraphs 6 and 7.
- 2. The licensee shall comply with the relevant provisions of the Supply Point Administration Agreement.
- 3. The Supply Point Administration Agreement shall be an agreement made between:
 - (a) on the one part, the licensee and all other licensed gas suppliers; and
 - (b) on the other part:
 - (i) all licensed Gas Transporters in their capacity as providers of supply point administration services; and
 - (ii) such other persons as are necessary parties, as determined by parties to the agreement.
- 4. The objectives referred to in paragraph 1(a) are:
 - a) the efficient discharge of the licensee's obligations under this licence;
 - b) the development, maintenance and operation of an efficient, coordinated and economical change of supplier process;
 - c) the furtherance of effective competition between gas suppliers and between relevant agents; and

- d) the promotion of efficiency in the implementation and administration of the supply point administration arrangements.
- 5. The Supply Point Administration Agreement shall contain:
 - a) provisions for admitting as an additional party to the Supply Point Administration Agreement any person who accepts the terms and fulfils the conditions (each as specified in the Supply Point Administration Agreement) on which accession to the Supply Point Administration Agreement is offered;
 - b) provisions for the licensee to refer to Ofgem for determination, whether of its own motion or as provided in the Supply Point Administration Agreement, any dispute which shall arise as to whether a person seeking to be admitted as a party to the SPAA has fulfilled any accession conditions; and if Ofgem determines that the person seeking admission has fulfilled all relevant accession conditions, for admitting such person as a party to the Supply Point Administration Agreement;
 - c) arrangements enabling modification of the Supply Point Administration Agreement:
 - (i) so as to better facilitate the achievement of the relevant objectives as set out in paragraph 5; and
 - (ii) following consultation with the parties, or representatives of the parties, to that agreement and other interested parties.
 - d) provisions (which shall be approved in advance by Ofgem) by virtue of which specified parts of the Supply Point Administration Agreement shall not be capable of modification without the prior approval of the Authority; and
 - e) provisions enabling parties to the Supply Point Administration Agreement to appeal against any proposed modification of the Supply Point Administration Agreement to the Authority for determination.
- 6. The licensee shall:

- a) take all reasonable measures to secure and implement (consistently with the procedures applicable under or in relation to the core industry documents to which it is party (or in relation to which it holds rights in respect of amendment)), and shall not take any steps to prevent or unduly delay, such changes which are appropriate in order to give full and timely effect to any modification which has been made to the Supply Point Administration Agreement; and
- b) furnish the Authority with a copy of any modification which is made.
- 7. The licensee shall provide for a copy of the Supply Point Administration Agreement to be provided to any person requesting the same upon payment of an amount not exceeding the reasonable costs of making and providing such copy.

Appendix 3 Gas Transporters Licence

Condition

1. The licensee shall become a party to and thereafter comply with those provisions of the Supply Point Administration Agreement relevant to it.

Appendix 4 List of Respondents



Appendix 5 The Supply Point Administration Agreement



Dated 2004 Deleted: 3

The Parties as named herein

- and -

SPAA Limited

Supply Point Administration Agreement

Version 1.0

Release date February 2004

Deleted: 3rd

Deleted: March

Deleted: 3

This copy of the SPAA is provided to enable the user to make copies of this document but only for the purpose of a copy being given to such personnel within the user's company as is necessary for the user's company to operate and supply or transport Gas within the United Kingdom. It is a condition of this licence that copies of the SPAA made in accordance with this licence may not be distributed for any other purpose, and each copy will be destroyed when the person possessing such a copy no longer needs such copy to fulfil the purpose set out above.

©2002 **SPAA Limited.** All rights reserved. No part of this publication may be reproduced or transmitted in any material form (including photocopying or storing it in any medium, by electronic means and whether or not transiently or incidentally to some other use or publication) without the permission of SPAA Limited.

INDEX

PART I: PRELIMINARY

| 1. | DEFINITIONS AND INTERPRETATION | 2 | | |
|----------|--|----|-----|--------------------------------------|
| 2. | CONDITIONS PRECEDENT | 8 | | |
| 3. | DURATION | 8 | | |
| 4. | ADDITIONAL PARTIES | 9 | | |
| 5. | MANDATORY, ELECTIVE AND VOLUNTARY SCHEDULES | 10 | | |
| | PART II: GOVERNANCE | | | Deleted: AND CHANGE CONTROL |
| 6. | CONSTITUTION OF SPAA EXECUTIVE COMMITTEE | 14 | | |
| 7. | CONSTITUTION OF THE SPAA FORUM | 23 | | |
| 8. | COSTS | 28 | | |
| v | | | | Deleted: 9. CHANGE CONTROL 31 |
| | PART III: CHANGE CONTROL | | | |
| 9. | CHANGE CONTROL. | 31 | | |
| | DADT IV. EVENTS OF DEFAULT, CONSEQUENCES OF DEFAULT AND | | | Deleted: III |
| | PART <u>IV</u> : EVENTS OF DEFAULT, CONSEQUENCES OF DEFAULT <u>AND LIMITATION OF LIABILITY</u> | | /- | Deleted: AND |
| 10. | EVENTS OF DEFAULT_CONSEQUENCES OF DEFAULT AND LIMITATION OF LIABILITY | 36 | | Deleted: AND |
| l I | | | , · | Deleted: I |
| | PART V: CONFIDENTIALITY | | 2 | Deleted: X |
| 11. | CONFIDENTIALITY | 39 | | |
| | PART VI: FORCE MAJEURE | | | |
| 12. | FORCE MAJEURE | 42 | | |
| 1 | | | , e | Deleted: X |
| | PART <mark>VII.; DISPUTES</mark> | | | |
| 13 | DISPUTES | 43 | | |
| | DADE WAS ANGED A AND ONE | | , | Deleted: X |
| I | PART VIII: MISCELLANEOUS | | 1 | |

| 14. | DEROGATIONS 44 | |
|-----|--|---------|
| 15. | CONTRACT MANAGEMENT | |
| 16. | ENTIRE AGREEMENT | |
| 17. | SEVERABILITY45 | |
| 18. | WAIVERS | |
| 19. | ASSIGNMENT AND SUB-CONTRACTING | |
| 20. | NOTICES45 | |
| 21. | AUDIT AND RECORDS | |
| 22. | COUNTERPARTS | |
| 23. | CONTRACTS (RIGHTS OF THIRD PARTIES) ACT 1999 | // / |
| 24. | GOVERNING LAW | |
| COL | DADTIEG | 40 |
| SCH | EDULE 1 PARTIES | 49 |
| SCH | EDULE 2 | 50 |
| SCH | EDULE 3 | 52 |
| SCH | EDULE 4SPAA SERVICE COMPANY LIMITED | 53 |
| ANN | NEX 1 TO SCHEDULE 4 | 61 |
| ANN | NEX 2 TO SCHEDULE 4 | 63 |
| ANN | NEX 3 TO SCHEDULE 4 | 64 |
| ANN | NEX 4 TO SCHEUDLE 4 | 65 |
| SCH | EDULE 5 | 73 |
| SCH | EDULE 6 SPAA PRODUCTS | 74 |

Deleted: R
Deleted: O

SUPPLY POINT ADMINISTRATION AGREEMENT

VERSION CONTROL – DRAFT DOCUMENT ONLY

Incorporating the changes set out in the following table:

| Version | Date of Release | Change Proposal Number | Change Proposal Title | Affected Clauses/Schedules | Effective Date |
|---------|-------------------------------------|------------------------------|--|--|-------------------|
| D0.1 | Feb | | Taken from the draft document, sections removed and some added, pending policy decisions | | N/A |
| D0.2 | March | | Some sections changed by GIGG CG | | N/A |
| D0.3 | 9 April 02 | | Angela and Afroze's comments added | Most | N/A |
| D0.4 | 24 April 02 | | Kirsten clears section 4, Afroze changes to section 5, David changes to section 8, Itret changes, Paul and Matt changes to sections 9, Henry changes to Schedule X on Spaa Ltd | " | N/A |
| D 0.5 | 1 st May 02 | | Changes from Paul to section 7, Angela's comments on Ofgem, Nigels on mandatory, comments from the minutes and DT on costs, added paper from Nigel on costs | " | N/A |
| D 0.6 | 22 nd May 02 | | Changes from Ruth's', Afroze', Steve's, and David's lawyers; and re-numbered throughout by BGT | " | N/A |
| D 0.7 | 27 th May 02 | | Renumbered, all clauses referenced, Ofgem's changes added | All | N/A |
| D 0.8 | 23 rd August 02 | | Herbert Smith redraft following consultation with SPAA Team and GIGG Members | All substantively except Clause 9 and incidental definitions | N/A |
| D 0.9 | 13 th September 02 | | Herbert Smith redraft following consultation with SPAA Team and GIGG Members | Incidental and Clause 9 | N/A |
| D 1.0 | 17 th January | | Herbert Smith redraft of clause 9 and incidental | Primarily Clause 9 | N/A |

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA_(suppliers

Deleted: SPAA (suppliers amended draft) 18 March.DOC

| Version | Date of Release | Change Proposal Number | Change Proposal Title | Affected Clauses/Schedules | Effective Date |
|---------|-------------------------------|------------------------------|--|---|-------------------|
| | 03 | | clauses following consultation with SPAA Team. | | |
| D.1.1 | 3 rd March 2003 | | Herbert Smith redraft in accordance with GIGG and Ofgem comments | Clause 5.2, 5.19, 6.9, 9.15, 9.16 and 9.20. | |
| D.1.2 | February 2004 | | Ofgem redraft following consideration of responses to consultation | <u>Various</u> Clause <u>s</u> | |

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

day of

Deleted: 3 2004

BETWEEN:

THE PERSONS whose names, registered numbers and registered or principal offices are set (1) out in Parts 1 to 4 of Schedule 1; and

SPAA LIMITED a company incorporated in England and Wales (registered number (2) 04365599) whose registered office is at 63 Duke Street, London W1K 5NS ("SPAA Ltd").

WHEREAS:

(A) Condition [to be inserted] of the Gas Suppliers, Licence and condition [to be inserted] of the $\textbf{Deleted:}\ X$ Gas Transporters Licence provides respectively that each Supplier and each Transporter is Deleted: y required to be a Party to and comply with the provisions of this Agreement;

(B) Each Supplier and each Transporter has accordingly agreed to enter into this Agreement on the basis of the terms and conditions set out below.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

PART I: PRELIMINARY

1. DEFINITIONS AND INTERPRETATION

1.1 Definitions in this Agreement:

"Act" means the Gas Act 1986 as amended by the Gas Act

1995 and the Utilities Act 2000;

"Accession Agreement" means an agreement in the form set out in

Schedule 2;

"Affected Party" has the meaning given to that term in Clause 12.1;

"Affiliate" in relation to any Party, means any holding company

of that Party, any subsidiary of that Party or any subsidiary of a holding company of that Party, in each case within the meaning of Sections 736, 736A

and 736B of the Companies Act 1985;

"Agreement" means this Supply Point Administration Agreement

the Schedules, Annexes and Appendices thereto;

"**Application for Derogation**" has the meaning given in Clause 14.2;

"Authority" means the Gas and Electricity Markets Authority

established by Section 1(1) of the Utilities Act 2000;

"Change Control Administrator" means the person appointed by SPAA EC pursuant

to Clause 9.3;

"Change Proposal" means a notice in writing from any Party or Parties

in accordance with Clause 9, proposing an amendment to this Agreement or to any of the SPAA

Products;

"Change Report" has the meaning given to it in Clause 9.11;

"Change Voting Date" means those dates set or amended by SPAA EC in

each Financial Year when relevant interested Parties may vote upon Change Proposals which are designated by SPAA EC to be voted on upon those

dates pursuant to Clause 9;

"Competent Authority" means the Secretary of State, the Authority and any

local or national agency, authority, department, inspectorate, minister, ministry, official, or public or statutory person (whether autonomous or not) of, or of the government of, the United Kingdom or the

European Community;

"Confidential Information" means, in relation to a Party, all data or other

information supplied to that Party by another Party under or pursuant to the provisions of this

Agreement;

"Contract Manager" means a person appointed by each Party pursuant to

Clause 15.1;

Deleted:

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers

amended draft) 18

March.DOC

| "Consumer" | means any person supplied or requiring to be | Deleted: Customer |
|-------------------------|---|---|
| | supplied with Gas at any Premises by a Supplier; | |
| "Data Protection Act" | means the Data Protection Act 1998; | |
| "Defaulting Party" | has the meaning given in Clause 10.1; | |
| "Designated Agreements" | means: | |
| | (i) this Agreement; | |
| | (ii) the Network Codes and any agreements entered into by a Party pursuant to the Network Codes; | |
| | (iii) any other agreement specified as a Designated Agreement from time to time by SPAA EC; and | |
| | (iv) any agreement which the Authority from time to time approves as a Designated Agreement; | |
| "Directive" | includes any present or future directive, requirement, instruction, direction or rule of any Competent Authority (but only, if not having the force of law, if compliance with the Directive is in accordance with the general practice of persons to whom the Directive is addressed) and includes any modification, extension or replacement thereof then in force; | |
| "Disclose" | means disclose, reveal, report, publish or transfer and "Disclosure" shall be construed accordingly; | |
| "Dispute" | has the meaning given to that term in Clause 13.1; | |
| "Disputing Party" | has the meaning given to that term in Clause 13.2; | |
| "Domestic Supplier" | means a Party who is entitled to make Domestic Supplies and who is described in Part 2 of Schedule 1; | Deleted: Customer Deleted: (as such term is used in the Gas Suppliers Standard |
| "Domestic Supply" | means a supply of Gas to a <u>premises which is taken</u> / wholly or mainly for domestic purposes. | Licence Conditions issued pursuant to the Utilities Act 2000) Deleted: 5.8 |
| "Effective Date" | means the date of this Agreement; | Deleted: Domestic Supplier, an |
| | | Industrial and Commercial Supplier or a Supplier |
| "Elective Register" | has the meaning given to that term in Clause 5.10; | Deleted: |
| "Elective" | means a Clause(s) of and/or a Schedule(s) to (or any | Deleted: 5.6 |
| | part(s) thereof) this Agreement which a <u>Party</u> may 's elect to comply with pursuant to and in accordance | Deleted: 5.14 |
| | with Clauses 5.7 to 5.9 (inclusive); | Deleted: 5.5 |
| "Floativo Schodulo" | has the magning given to that term in Clause 5.6. | Deleted: SPAA_180204.DOC |
| "Elective Schedule" | has the meaning given to that term in Clause 5.6; | Inserted: SPAA_180204.DOC |
| | | Deleted: SPAA (suppliers amended draft) 18 March.DOC |

| "energywatch" | means the Gas and Electricity Consumer Council established by section 2 of the Utilities Act 2000; | |
|---|--|--|
| "Event of Default" | has the meaning given to that term in Clause 10.1; | |
| "Extra Votes" | has the meaning given to that term in Clauses 6.9 and 6.11 ; | |
| "Financial Year" | means the period from 1st April to 31st March each year provided that the First Financial Year shall begin on the date of this Agreement and end on the next following 31st March and the final Financial Year shall, in the event of termination of this Agreement otherwise than on the anniversary of the last day of the Financial Year, be such shorter period as shall end on the date of termination; | |
| "Force Majeure" | means any event or circumstance which is beyond the reasonable control of any Party acting as a Reasonable and Prudent Operator and which results in or causes the failure of that Party to perform any of its obligations under this Agreement, provided that lack of funds shall not be interpreted as a cause beyond that Party's reasonable control; | |
| "Forum Chairman" | has the meaning given to that term in Clause 7.16; | |
| "Gas" | has the meaning given in Section 48(1) of the Act: | |
| "Gas Suppl <mark>iers L</mark> icence" | means a licence to supply Gas granted by the Authority pursuant to section 7A(1) or 7A(4) of the Act or treated as so granted by virtue of a scheme made under Schedule 5 to the Gas Act 1995; | Deleted: y Deleted: |
| "Gas Transporter" | means a Party who is entitled to transport gas and who is described in Part 4 of Schedule 1 and "Transporter(s)" shall be construed accordingly: | |
| "Gas Transportation Database" | means the database of information (including on MPRNs) held by each holder of a Gas Transportation Licence; | |
| "Gas Transport <mark>ers</mark> Licence" | means a licence to transport Gas granted by the Authority pursuant to section 7 or 7(4) of the Act; | Deleted: ation |
| "Group of Domestic Suppliers" | means a group of Domestic Suppliers and their Affiliates who are also Parties and Domestic Suppliers; | |
| "Group of Gas Transporters" | means a group of Gas Transporters and their Affiliates who are also Parties and Gas Transporters; | |
| "Group of Industrial and Commercial Suppliers" | means a group of Industrial and Commercial Suppliers and their Affiliates who are also Parties and Industrial and Commercial Suppliers; | Deleted: SPAA_180204.DOC Inserted: SPAA_180204.DOC Deleted: SPAA (suppliers amended draft) 18 March.DOC |
| | | P. |

| "Group of Parties" | means a group of Suppliers or Transporters and their | | Deleted: |
|---|--|--------------|---------------------------------------|
| Group of Farties | Affiliates who are also Parties, provided that any | | Deleted: |
| | Affiliates who are not Suppliers or Transporters shall | | |
| | be excluded and for the avoidance of doubt where | | |
| | the phrase Group of Parties in a category of Parties is | | |
| | used only those Affiliates who are in the same | | Deleted: y |
| | category of Parties shall be included in the group; | | Deleted: y |
| "I&C <u>Supplier Member"</u> | has the meaning given to that term in Clause 6.1. | | Deleted: |
| Supplier Wellber | this the meaning given to that term in Oldese of the | - | Deleted: 6.1 |
| "I&C Supply" | means a supply of Gas to a Consumer taken wholly | | Inserted: 6.1 |
| | or mainly for non-domestic purposes (as such term is | | Deleted: 6.1 |
| | used in the Gas Suppliers Standard Licence | ``\ | Deleted: Customer |
| | Conditions issued pursuant to the Utilities Act 2000); | | |
| "Industrial and Commercial | means a Party who is entitled to make I&C Supplies | | |
| Supplier" | and who is described in Part 3 of Schedule 1; | | |
| | , | | |
| "Intellectual Property" | means patents, registered design rights, unregistered | | |
| | design rights, copyrights, rights in trade marks | | |
| | whether registered or not, goodwill and rights in | | |
| | confidential information and know-how and any | | |
| | associated or similar rights (including, in all cases, applications and rights to apply therefor); | | |
| | applications and rights to apply therefor), | | Deleted: |
| "Large Domestic <u>Supplier</u> Member" | has the meaning given to that term in Clause 6.3.2; | | |
| "Large Domestic Supplier" | means a Domestic Supplier who has one (1) million | | |
| a grant and Fr | or more MPRNs for Domestic Supplies Registered | | |
| | on all Gas Transportation Databases; | | |
| "Large Transporter" | mana a Cas Transporter who has more than one (1) | | |
| Large Transporter | means a Gas Transporter who has more than one (1) million or more MPRNs Registered on its Gas | | |
| | Transportation Database; | | |
| | * | | |
| "Large Transporter Member" | has the meaning given to that term in Clause 6.3.4 | | |
| "Mandatory" | means a Clause(s) of and/or a Schedule(s) to (or any | | |
| ivialitator y | part(s) thereof) this Agreement which a <u>Party</u> is | | Deleted: Domestic Supplier, an |
| | obliged to comply with pursuant to Clauses 5.3 and | | Industrial and Commercial |
| | <u>5.4</u> : | | Supplier, or a Supplier |
| "Mandatory Schedule" | has the meaning given to that term in Clause 5.2; | | Deleted: 5.4 |
| Mandatory Schedule | has the meaning given to that term in Clause 3.2, | | Deleted: 5.4 |
| "MPRN" | means a unique reference number that identifies each | | Inserted: 5.4 |
| | point on a pipeline system, the conveyance of Gas | | |
| | through which is authorised by a Gas Transportation | | |
| | Licence, at which Gas may, in accordance with a Network Code, be offtaken from such pipeline | | |
| | system for the purpose of supply directly to a | | |
| | Premises; | | |
| 100 J. G. J. II | , | | |
| "Network Code" | means the document prepared by the holder of a Gas | , | Deleted: SPAA_180204.DOC |
| | Transportation Licence, pursuant to Condition 9(2) of a Gas Transportation Licence, that forms the basis | /. | Inserted: SPAA_180204.DOC |
| | of arrangements between the holder of a Gas | $-i'_i$ | Deleted: SPAA (suppliers |
| | | 1/1 | amended draft) 18 |
| | | 11/ | March.DOC |

Transportation Licence and the shippers whose Gas it transports, such as the Transco Network Code;

"New Party"

has the meaning given to that term in Clause 4.1;

"Operational Issue"

means an issue or problem perceived by one or more Parties arising out of the operation of the arrangements designed to facilitate competition in the Gas industry, which, for the avoidance of doubt, shall not be limited to issues or problems arising out

of or impacting upon this Agreement;

"Party"

means a party to this Agreement and "Parties" shall be construed accordingly provided that (a) nothing shall preclude a Party from being a Domestic Supplier and an Industrial and Commercial Supplier and (b) unless expressly stated otherwise SPAA Ltd shall only be a Party for the purposes of Schedule 4 (and in such case as expressly specified);

"Party Change Administrator"

means the person appointed by each Party under Clause 9.4;

"Party Liable"

has the meaning given to that term in Clause 10.15;

"Premises"

means any land, building or structure;

"Ouarter"

means the period of three calendar months ending on

a Quarter Day;

"Quarter Day"

means each or as the context may require any or a particular one of 31st March, 30th June, 30th September and 31st December or where the Quarter Day is not a Working Day, the next Working Day

thereafter;

"Reasonable and Prudent Operator" means a person exercising that degree of skill,

diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled and experienced operator engaged in the same type of undertaking under the same or similar circumstances:

"Registered" means the recording on the Gas Transportation

Database of a Supplier as being responsible for a MPRN from a particular date and "Registration"

shall be construed accordingly;

"Related Undertakings"

means in relation to any Party an undertaking in which that Party has a participating interest as defined by section 260 of the Companies Act 1985;

"Report"

has the meaning given in Clause 21.2;

"Secretariat"

has the meaning given to that term in Clause 6.54;

"Secretary"

means the person appointed as secretary of the

SPAA EC pursuant to Clause 6.26;

Deleted: supply

Deleted: 10.15

Deleted: 10.14

Inserted: 10.15

Deleted: SPAA_180204.DOC
Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

| "Secretary of State" | has | the | meaning | given | to | that | term | in | the |
|----------------------|--------------------------|-----|---------|-------|----|------|------|----|-----|
| | Intermediation Act 1079. | | | | | | | | |

Interpretation Act 1978;

"Shadow SPAA EC" means the Shadow SPAA EC Members, acting as

SPAA EC in a shadow capacity, before the execution

of this Agreement;

"Shadow SPAA EC Member" means those persons whose names are set out in

Clause 6.17 acting in a shadow capacity as SPAA EC Members before the execution of this

Agreement;

"Small Domestic Supplier Member" has the meaning given to that term in Clause 6.3.3;

"Small Domestic Supplier" means a Domestic Supplier who has less than one (1)

million MPRNs Registered on all Gas Transportation

Databases;

"Small Transporter" means a Gas Transporter who has less than one (1)

million MPRNs Registered on its Gas Transportation

Database;

"Small Transporter Member" has the meaning given to that term in Clause 6.3.5;

"SPAA EC" means the executive committee of SPAA Ltd

constituted pursuant to Clause 6.3 consisting of the

SPAA EC Members;

"SPAA EC Chairman" means the person appointed as chairman of the

SPAA EC pursuant to Clause 6.22;

"SPAA EC Member" has the meaning given to that term in Clause 6.3;

"SPAA Forum" means the body constituted pursuant to the terms of

Clause 7;

"SPAA Products" means those items listed in Schedule 6:

"Supplier" means an Industrial and Commercial Supplier and/or

a Domestic Supplier and "Suppliers" shall be construed accordingly provided that nothing shall preclude a person from being a Party as a Domestic Supplier and an Industrial and Commercial Supplier;

"Total Weighted Vote" has the meaning given to that term in Clause 6.9;

"VAT" has the meaning given to that term in the Value

Added Tax Act 1994 and any tax of a similar nature which may be substituted for or levied in addition to

ıt;

"Voluntary" means a Clause(s) of and/or a Schedule(s) to (or

part(s) thereof) this Agreement with which a Party is

not obliged to comply;

"Voluntary Schedule" has the meaning given to that term in Clause 5.18;

"Weighted Votes" has the meaning given to that term in Clause 6.8; and

Deleted: Domestic Supplier, an Industrial and Commercial Supplier, or a Supplier

Deleted: 5.15

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers

amended draft) 18

March.DOC

SPAA 040304.DOC

Deleted:

"Working Day"

means any day other than a Saturday, a Sunday, Christmas Day, Good Friday or a day which is a bank holiday within the meaning of the Banking and Financial Dealings Act 1971.

- 1.2 In this Agreement, unless the context requires otherwise, any reference to:
 - 1.2.1 a "person" includes a reference to an individual, body corporate, association or partnership;
 - 1.2.2 the singular shall include the plural and vice versa;
 - 1.2.3 this "Agreement" shall mean this agreement, the Schedules, Annexes and Appendices thereto;
 - 1.2.4 a Clause, Schedule or Part is a reference to a clause of or schedule to or part of this Agreement;
 - 1.2.5 writing includes all methods of reproducing words in a legible and non-transitory form;
 - 1.2.6 any statute or any other subordinate legislation, any other agreement or instrument shall be construed as a reference to that statute, subordinate legislation, other agreement or instrument as amended, or re-enacted or consolidated from time to time;
 - 1.2.7 the masculine gender includes the feminine gender.
- 1.3 The headings in this Agreement are for the ease of reference only and shall not affect its interpretation.
- 1.4 In this Agreement, references to "include" or "including" are to be construed without limitation to the generality of the preceding words.

2. CONDITIONS PRECEDENT

- 2.1 The rights and obligations of a <u>Party</u> in a category of <u>Parties</u> (which does not hold a <u>Gas Suppliers</u> Licence or <u>Gas Transporters Licence</u> (as appropriate)) pursuant to a <u>Mandatory Schedule</u>, and the right of such <u>Party to make an election in respect of an Elective Schedule</u>, shall be subject to such <u>Party holding a Gas Suppliers</u> Licence or <u>Gas Transporters Licence</u> (as appropriate) in that capacity and SPAA EC notifying the Parties that it has received evidence of such licence pursuant to Clause 4.
- 2.2 A <u>Party</u> shall not be obliged to give a <u>Party</u> which does not hold a <u>Gas Suppliers Licence or Gas Transporters Licence (as appropriate)</u> the benefit of any provision in a <u>Mandatory Schedule</u> which relates to such category of <u>Parties</u> and shall not be under any obligation in relation thereto unless and until such <u>Party holds a Gas Suppliers Licence or Gas Transporters Licence (as appropriate)</u> in that capacity, and <u>SPAA EC</u> has notified the <u>Parties pursuant to Clause 4.</u>
- 2.3 From the date of SPAA EC's notice in Clause 4.8, a <u>Party</u> in a category of <u>Parties</u>, shall be entitled to receive the benefit of, exercise rights and be subject to obligations under, a Mandatory Schedule and shall be entitled to make an election in respect of Elective Schedule, in each case relating to such category of <u>Parties</u>.

3. **DURATION**

3.1 This Agreement shall take effect on the Effective Date save for any rights or obligations of a Party, which are expressed in Clause 2 to be conditional.

Deleted: Supplier

Deleted: y

Deleted: y

Deleted: Supplier

Deleted: Supplier

Deleted: y

Deleted: Supplier

Deleted: y

Deleted: y

Deleted: y

Deleted: Supplier

Deleted: y

Deleted: y

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

- 3.2 Subject to Clauses 10.13 and 10.14 this Agreement shall remain in effect in respect of a Party until such Party ceases to be a party to this Agreement in accordance with Clause 10 and Clause 20.
 - 3.3 Subject to Clauses 10.13 and 10.14 this Agreement shall remain in effect until:
 - 3.3.1 all the Parties cease to be Parties in accordance with Clause 10; or;
 - 3.3.2 there remains just one (1) Supplier as a Party.
 - 3.4 This Clause is without prejudice to Clause 5 and Clause 14,

4. ADDITIONAL PARTIES

- 4.1 Subject to the following provisions of this Clause 4, the Parties shall admit as an additional party to this Agreement any person (a "New Party") who is not at that time already a Party, who applies to be admitted in the capacity requested by the New Party subject to the New Party holding a Gas Suppliers Licence in that capacity, or a Gas Transporter Licence in that capacity, or being in the process of application for a Gas Suppliers Licence in that capacity, or a Gas Transporters Licence in that capacity.
- 4.2 Subject to Clause 4.3, a New Party wishing to be admitted as an additional Party shall apply to SPAA EC for admission on a form of application issued by SPAA EC from time to time and shall deliver such form to SPAA EC together with any other documents referred to in the form. A New Party shall self certify in the form set out in Schedule 3 that its system design is such that on becoming Party it will be able to fully comply with all the Mandatory Schedules and Elective Schedules that it has elected or intends to elect to comply with in accordance with this Agreement. Within 30 Working Days of receipt of the application, SPAA EC shall notify the New Party and the Authority that either the New Party shall be admitted as a Party or that it requires from the New Party the information and/or documents referred to in the application form.
- 4.3 Where SPAA EC notifies the New Party that it requires the information specified in Clause 4.2, the New Party shall within 20 Working Days of receiving SPAA EC's notice provide such information, failing which the New Party's application shall lapse and be of no effect and the New Party shall not be, and shall not be entitled to be, admitted as a New Party consequent upon such application without prejudice to any new application for admission it may make thereafter.
- 4.4 SPAA EC may determine not to admit a New Party if such New Party does not provide the evidence requested in its application within the time period referred to in Clause 4.3. Where SPAA EC determines not to admit a New Party it shall provide such New Party with the reasons for its decision. Where SPAA EC determines not to admit a New Party as a Party or fails to notify the New Party within 30 Working Days of receipt of the New Party's application, the New Party may refer the matter to the Authority for its determination. The determination of the Authority shall be final and binding for all purposes.

4.5 Where:

- 4.5.1 SPAA EC notifies the New Party and the Authority in accordance with Clause 4.2 that the New Party is to be admitted as a Party; or
- 4.5.2 following a request for information pursuant to Clause 4.2 the New Party provides sufficient information satisfactory to SPAA EC within the time period specified in Clause 4.3; or

Deleted: 10.13

Deleted: 10.12

Deleted: 10.14

Deleted: 10.13

Inserted: 10.13

Inserted: 10.13

Inserted: 10.13

Deleted: 10.13

Deleted: 10.12

Deleted: 10.14

Deleted: 10.14

Deleted: 10.13

Deleted: 10.14

Deleted: 10.14

Deleted: 10.19

Deleted: 10.19

Inserted: SPAA_180204.DOC
Deleted: SPAA (suppliers
amended draft) 18

Deleted: SPAA_180204.DOC

March.DOC

4.5.3 the Authority determines that the New Party shall become a Party pursuant to Clause 4.4.

SPAA EC shall, within 5 Working Days, prepare an Accession Agreement, which shall be executed by a delegate authorised by SPAA EC on behalf of all Parties other than the New Party. Each Party hereby authorises and instructs any delegate authorised by SPAA EC to sign any such Accession Agreement on its behalf and undertakes not to withdraw, qualify or revoke any such authority or instruction at any time. Upon execution of the Accession Agreement, the New Party shall become a Party for all purposes of this Agreement from the date specified in such Accession Agreement.

- 4.6 SPAA EC shall promptly notify all Parties and the Authority of the execution and delivery of the Accession Agreement.
- Where a New Party accedes to this Agreement and does not hold a Gas Suppl<u>iers</u> <u>Licence or Gas Transporters Licence</u>, but is in the process of applying therefor:

to 9

4.7.1 it shall not be entitled to exercise any voting rights pursuant to Clauses 6 to 9 (inclusive) until Clause 4.8 applies;

4.7.2 it shall forthwith inform SPAA EC following:

(A) the grant of a Gas Suppl<u>iers, Licence or Gas Transporters Licence to it (and in</u> such a case shall send a copy of its Gas Suppl<u>iers, Licence or Gas Transporters Licence</u> (or other evidence thereof) to SPAA EC); or

Deleted: y

Deleted: y

Deleted: y

(B) the refusal to grant a Gas Suppliers Licence or Gas Transporters Licence to it.

4.8 If Clause 4.7.2(A) applies, SPAA EC shall within 10 Working Days of receipt of such information, notify the New Party, the Authority and the Parties that as from the date of SPAA EC's notice the voting restrictions on the New Party will cease to apply and Clause 2.3 shall apply.

5. MANDATORY, ELECTIVE AND VOLUNTARY SCHEDULES

5.1 Subject to Clause 3 and Clauses 5.3, 5.4, 5.5, 5.11, 5.12, 5.14, and 5.18, this Agreement shall be binding on all Parties.

Deleted: 5.9,

Deleted: 5.10,

Deleted: 5.12 Deleted: 5.16

Mandatory Schedules

- A Clause(s) of and/or a Schedule(s) to (or part(s) thereof) this Agreement may be designated as Mandatory for a Supplier, a Domestic Supplier or an Industrial and Commercial Supplier, or a Transporter as part of a Change Proposal (a "Mandatory Schedule").
- 5.3 A Domestic Supplier shall be obliged to comply with a Mandatory Schedule designated as Mandatory for Domestic Suppliers and/or Suppliers but shall not be obliged to comply with a Mandatory Schedule designated as Mandatory for Industrial and Commercial Suppliers only.
- 5.4 An Industrial and Commercial Supplier shall be obliged to comply with a Mandatory Schedule designated as Mandatory for Industrial and Commercial Suppliers and/or Suppliers but shall not be obliged to comply with a Mandatory Schedule designated as Mandatory for Domestic Suppliers only.
- 5.5 A Transporter shall be obliged to comply with a Mandatory Schedule designated as Mandatory for Gas Transporters

Elective Schedules

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

- A Clause(s) of and/or a Schedule(s) to (or part(s) thereof) this Agreement may be designated as Elective for a Supplier, a Domestic Supplier or an Industrial and Commercial Supplier, or a Transporter as part of a Change Proposal (an "Elective Schedule").
- 5.7 A Domestic Supplier may elect to comply with an Elective Schedule designated as Elective for Domestic Suppliers and/or Suppliers by notice in writing to SPAA EC.
- An Industrial and Commercial Supplier may elect to comply with an Elective Schedule designated as Elective for Industrial and Commercial Suppliers and/or Suppliers by notice in writing to SPAA EC.
- 5.9 A Transporter may elect to comply with an Elective Schedule designated as Elective for Transporters by notice in writing to SPAA EC.
- 5.10 The SPAA EC shall maintain and send to <u>all Parties</u> and the Authority each month a list of <u>Parties</u> ("Elective Register") who at such time have elected, pursuant to Clauses 5.7, 5.8 or 5.9, to comply with Elective Schedules (and the parts of any Elective Schedule with which they have elected to comply) and are continuing to so comply.
- 5.11 Once a Party has informed SPAA EC of its written election to adhere to an Elective Schedule and SPAA EC has entered such Party on the Elective Register in respect of such Schedule such Party shall be obliged to comply with the Elective Schedules for so long as it remains on the Elective Register in respect of such Elective Schedule.
- Where a <u>Party</u> has elected to comply with an Elective Schedule it may give notice in writing to SPAA EC notifying SPAA EC that from a date not earlier than one (1) month after the date of its notice it no longer wishes to comply with such Elective Schedule. SPAA EC shall remove such <u>Party</u> from the Elective Register in respect of such Elective Schedule at the end of the time period specified in the <u>Party's</u> notice and, from such date, the <u>Party</u> shall not be obliged to comply with such Elective Schedule.

5.13

- 5.13.1 Any Party who is registered on the Elective Register may report ("Reporting Party") any suspected non-compliance with an Elective Schedule by another Party ("Non Compliant Party") to SPAA EC describing the non-compliance and requesting that such Party be removed from the Elective Register.
- 5.13.2 On receipt of such report SPAA EC shall notify the Non Compliant Party and shall invite such Party to submit a report to SPAA EC within 10 Working Days of receipt of such notice setting out any reasons why it may dispute the non-compliance.
- 5.13.3 If the <u>Party does not submit a report within the period referred to in Clause 5.13.2</u>, SPAA EC shall remove such <u>Party from the Elective Register at the end of such period.</u>
- 5.13.4 If the Non Compliant <u>Party</u> submits a report within the period referred to in Clause 5.13.2:
 - (A) SPAA EC shall hear representations from the <u>Party</u> concerned;
 - (B) following consideration of the <u>Party's reports and representations it shall</u> resolve whether the Non Compliant <u>Party should be retained on or removed from the Elective Register;</u>
 - (C) SPAA EC shall notify the Reporting <u>Party</u> and the Non Compliant <u>Party</u> of its decision pursuant to Clause 6.41;

Deleted: the

Deleted: Suppliers

Deleted: 5.6

Deleted: or

Deleted: 5.7

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier's

Deleted: Supplier

Formatted: Indent: Left: 1.27 cm, Hanging: 1.27 cm

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted:

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: 5.11.2

Deleted: Supplier

Deleted: Supplier **Deleted:** 5.11.25

Deleted: Suppliers

Deleted: Suppliers'

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

(D) if the SPAA EC resolves that the Non Compliant Party has not complied with an Elective Schedule, it shall remove such Party from the Elective Register in respect of such Elective Schedule 10 Working Days (or such longer period as the SPAA EC may decide, in relation to that resolution) after notification of the minutes of the relevant SPAA EC decision pursuant to Clause 6.41 unless before the expiry of such period the Non Compliant Party appeals to the SPAA Forum for its determination (and provides evidence thereof to SPAA EC within such period); and

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

(E) if the SPAA EC resolves that the Non Compliant Party has complied with the Elective Schedule it shall not remove such Party from the Elective Register provided that the Reporting Party may appeal SPAA EC's determination to the SPAA Forum for its determination within 10 Working Days (or such longer period as the SPAA EC may decide in relation to that resolution) of receiving the minutes of the relevant SPAA EC decision pursuant to Clause 6.41.

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

5.13.5 A SPAA EC Member shall be disqualified from acting, and shall not act in his capacity as a SPAA EC Member, in relation to a resolution pursuant to this Clause and his alternate shall act in his place (unless the provisions of Clause 5.13.5.5.4(A) and (B) apply to him as well) in relation to a resolution where:

Deleted: 5.11.5

Deleted: (A)

- (A) his employer is the Non Compliant Party or an Affiliate of the Non Compliant Party; or
- (B) his employer is the Reporting Party or an Affiliate of the Reporting Party.

5.14 If the SPAA Forum dismisses any appeal made by the Non Compliant Party or allows any appeal made by the Reporting Party, SPAA EC shall remove the Non Compliant Party from the Elective Register within 5 Working Days of receipt by SPAA EC of the SPAA Forum decision. If the SPAA Forum allows any appeal of the Non Compliant Party or dismisses any appeal of the Reporting Party, SPAA EC shall not remove the Non Compliant Party from the Elective Register in respect of the non-compliance reported.

A <u>Party</u> may apply to SPAA EC to be reinstated on the Elective Register if it considers that it

is at that time capable of complying with the Elective Schedule. SPAA EC may reinstate the Party to the Elective Register in respect of such Elective Schedule, after having considered

any report or representations from such Party. Such Party may appeal any such decision of

SPAA EC to the SPAA Forum for its determination within 10 Working Days (or such longer

period as the SPAA EC may decide in relation to that resolution) of receiving the minutes of the relevant SPAA EC decision pursuant to Clause 6.41. If the SPAA Forum allows such <u>Party's</u> appeal, SPAA EC shall reinstate such <u>Party</u> onto the Elective Register within 5

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier **Deleted:** Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier's

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: 5.11

The right for a <u>Party</u> who is registered on the Elective Register to request removal of another <u>Party</u> from the Elective Register pursuant to Clause <u>5.13</u>, is the sole and exclusive remedy available to any Party for non-compliance with an Elective Schedule and Clause 10.1.1 shall not apply to breach of any part of an Elective Schedule.

Working Days of receipt of the SPAA Forum determination.

Voluntary Schedules

- 5.17 A Clause(s) of and/or Schedule(s) to (or parts thereof) this Agreement may be designated as Voluntary for a Supplier, a Domestic Supplier or an Industrial and Commercial Supplier, or Transporter as part of the Change Proposal (a "Voluntary Schedule").
- 5.18 A Voluntary Schedule shall contain statements of best practice and guidelines that a <u>Party</u> may chose to follow. A Voluntary Schedule is not intended to be legally binding nor have

Deleted: s

Deleted: Supplier

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

any legal effect whatsoever as between the Parties. For the avoidance of doubt Clause 10.1.1 shall not apply to breach of any part of a Voluntary Schedule.

Changes to status

_The status of a Mandatory Schedule, an Elective Schedule and/or a Voluntary Schedule may change as part of a Change Proposal.

Appeals

5.20 If a Non Compliant Party or Reporting Party appeals to the SPAA Forum pursuant to Clauses 5.13.4(D), 5.13.4(E) or 5.15, the SPAA Forum's determination shall be final and binding in the absence of fraud or manifest error.

Deleted: Supplier Deleted: (D) **Deleted:** 5.11.4(D) Deleted: (E) Inserted: (E)

Deleted: Supplier

Deleted: 5.13 Inserted: (D)

Deleted: 5.11.4(E)

Deleted: SPAA_180204.DOC Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

PART II: GOVERNANCE

6. CONSTITUTION OF SPAA EC

6.1 The Parties hereby delegate to SPAA EC all powers necessary to fulfil the objects contained in Clause 6.2.

Objects

- 6.2 SPAA EC shall, subject to and in accordance with the other provisions of this Agreement, have the powers to:
 - 6.2.1 consider and co-ordinate communications relating to, the voting on, and implementation of, Change Proposals, (including any appeal to the Authority), set and amend Change Voting Dates and, where the written consent of the Authority is required to any Change Proposals, recommend such Change Proposal to the Authority on behalf of the Parties;
 - 6.2.2 consider and approve and co-ordinate any applications from potential New Parties to become a Party;
 - 6.2.3 develop budgets in accordance with Clause 8;
 - 6.2.4 hire any professional advisers, including accountants to audit its costs;
 - 6.2.5 appoint and remove and make arrangements for the appointment and removal of a Secretariat including a Change Control Administrator;
 - 6.2.6 check and notify Parties that they are Defaulting Parties in accordance with the provisions of Clause 10;
 - 6.2.7 consider and grant derogations in accordance with Clause 14;
 - 6.2.8 consider Operational Issues and make recommendations in relation thereto to the Parties;
 - 6.2.9 consider, approve and authorise the licensing, sub-licensing or otherwise dealing with Intellectual Property belonging to SPAA Ltd, for any use which does not hinder, delay or frustrate, in any way whatsoever, supply competition in the Gas industry in Great Britain;
 - 6.2.10 maintain and distribute the Elective Register and consider and approve applications for removal from, retention on and/or reinstatement to the Elective Register;
 - 6.2.11 consider, approve and authorise the entering into by SPAA Ltd of any contract or arrangement whereby SPAA Ltd procures the performance by a third party of any activities which might otherwise be carried out by the Secretary and/or the Secretariat under this Agreement, and/or the transfer assignment, leasing, licensing or other dealing by SPAA Ltd of any property and/or rights and liabilities of SPAA Ltd relating to the performance of such activities before the commencement of such contract or arrangement of any third party, as SPAA EC considers necessary or desirable;
 - 6.2.12 consider and resolve disputes on the categorisation of items on the agenda for SPAA Forum meetings pursuant to Clause 7.13;
 - 6.2.13 consider the future development issues as set out for consideration in Schedule 5 and make proposals to the Parties concerning such issues; and

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

6.2.14 constitute sub-committees to deal with any of the above matters.

SPAA EC Membership

- 6.3 The SPAA EC shall consist of the following persons ("SPAA EC Members"):
 - two SPAA EC Members (the "I&C Supplier Members") appointed pursuant to Clause 6.7 by the Industrial and Commercial Suppliers;
 - 6.3.2 two SPAA EC Members (the "Large Domestic Supplier Members") appointed pursuant to Clause 6.7 by Large Domestic Suppliers
 - 6.3.3 one SPAA EC Member (the "Small Domestic Supplier Member") appointed pursuant to Clause 6.7 by Small Domestic Suppliers.
 - one SPAA EC Member (the "Large Transporter Member") appointed pursuant to 6.3.4 Clause 6.7 by Large Transporters.
 - one SPAA EC Member (the "Small Transporter Member") appointed pursuant to 635 Clause 6.7 by Small Transporters.
- 6.4 Subject to Clause 6.5, a SPAA EC Member proposed pursuant to Clause 6.7 shall be an employee of a Party or any Affiliate of a Party provided such Party is within the category of Parties that is entitled to propose the relevant SPAA EC Member. No person may simultaneously be proposed pursuant to Clause 6.7 for appointment to the post of more than one SPAA EC Member (or alternates thereof).

- 6.5 Notwithstanding Clause 6.3, no Group of Parties may propose more than one candidate for appointment as an SPAA EC Member pursuant to Clause 6.7.
- 6.6 All SPAA EC Members shall be elected in accordance with the election procedures set out in Clause 6.7.
- Subject to Clause 6.5, no later than 40 Working Days before 1st November in each year: 6.7
 - 671 each Industrial and Commercial Supplier may propose to the Secretary one (1) candidate for election as an I&C Supplier Member;
 - 6.7.2 each Domestic Supplier who has (on the date of such proposal) one (1) million or more MPRNs for Domestic Suppliers Registered on all Gas Transportation Databases may propose to the Secretary one (1) candidate for election as a Large Domestic Supplier Member;
 - 6.7.3 each Domestic Supplier who has (on the date of such proposal) less than one (1) million MPRNs for Domestic Suppliers Registered on all Gas Transportation Databases may propose to the Secretary one (1) candidate for election as a Small Domestic Supplier Member:
 - each Transporter who has (on the date of such proposal) one (1) million or more 6.7.4 MPRNs Registered on its Gas Transportation Database may propose to the Secretary one (1) candidate for election as a Large Transporter Member;
 - 6.7.5 each Transporter who has (on the data of such proposal) less than one (1) million Registered on its Gas Transportation Database may propose to the Secretary one (1) candidate for election as a Small Transporter Member.

and the Secretary shall no later than 30 Working Days before the 1st November in each year notify the list of candidates to each Party. Where the number of candidates proposed in Clause Deleted: y

Deleted:

Deleted: SPAA 180204.DOC Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

6.7 in a particular category exceed number of persons allowed to constitute SPAA EC Members in such category pursuant to Clause 6.3, the Secretary shall invite (and may do so by e-mail) the Parties, from the relevant category, to cast votes (and they may do this by e-mail) for their favoured candidate to the Secretary within 10 Working Days of receipt of such list of candidates. Each category of Parties, entitled to vote for a Large Domestic Supplier Member, a Small Domestic Supplier Member, an I&C Supplier Member, a Large Transporter Member, or a Small Transporter Member shall have the number of votes calculated in accordance with Clause 6.8. For the avoidance of doubt, only Large Domestic Suppliers shall be entitled to vote for a Large Domestic Member, only Small Domestic Suppliers shall be entitled to vote for a Small Domestic Member, only Large Transporters shall be entitled to vote for an I&C Member, only Large Transporters shall be entitled to vote for a Large Transporter Member, and only Small Transporters shall be entitled to vote for a Small Transporter Member.

Deleted: y

Deleted: or

Deleted: and

Subject to Clauses 6.9, 6.10, and 6.11 the percentage of the vote to which each <u>Party in a category</u> of Parties, shall be entitled in respect of any election pursuant to Clause 6.7 ("Weighted Votes") shall be calculated in accordance with the following formula:

Deleted: Supplier

Deleted: y

$$V = \frac{N}{SN + X} \times 100$$

Where:

6.8

"V" means the percentage of the vote to which a <u>Party in a category of Parties shall be</u> entitled, calculated to two decimal places;

Deleted: Supplier

Deleted: that

"N" means either:

the number of MPRNs for which a <u>Party in a category of Suppliers was Registered on</u> all Gas Transportation Databases in the month preceding the election which shall be determined from the Report submitted by that Supplier in respect of that month pursuant to Clause 20.2; or

Deleted: Supplier

Deleted: that

Deleted: Part

- (ii) the number of MPRNs which a Party in a category of Transporters had Registered on its Gas Transportation Database in the month preceding the election which shall be determined from the Report submitted by that Transporter in respect of that month pursuant to Clause 20.2; or
- (ii) one,

whichever is the greater.

"SN" means either:

(i) the total number of MPRNs for which <u>Parties</u> in that category of <u>Suppliers</u> were Registered on all Gas Transportation Databases in the month preceding the election which shall be determined from the Report submitted in respect of that month by each such <u>Party</u> pursuant to Clause 20.2; or

Deleted: all Suppliers

Deleted: Party

Deleted: Supplier

(ii) the total number of MPRNs which Parties in that category of Transporters had Registered on their Gas Transportation Databases in the month preceding the election which shall be determined from the Report submitted in respect of each month by each such Party pursuant to Clause 20.2.

"X" means either:

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

the number of <u>Parties</u> in that category of <u>Suppliers</u> for whom no MPRNs were Registered on any Gas Transportation Database in the month preceding the election which shall be determined from the Report submitted in respect of that month by such <u>Parties</u> pursuant to Clause 20.2; or

Deleted: Suppliers

Deleted: Party

Deleted: Suppliers

(ii) the number of Parties in that category of Transporters who had no MPRNs

Registered on their Gas Transportation Databases in the month preceding the election which shall be determined from the Report submitted in respect of that month by such Parties pursuant to Clause 20.2.

Deleted:

6.9 If, pursuant to the formula set out in Clause 6.8, any <u>Party</u> (not being in a Group of Parties) and Group of Parties in a category of Parties has in excess of 20% of the total percentage of the vote to which all the <u>Parties</u> in <u>that</u> category are entitled ("Total Weighted Vote"):

Deleted: Supplier

Deleted: v

the vote to which all the <u>Parties in that category are entitled</u> ("**Total Weighted Vote**"):

6.9.1 subject to Clause 6.9.2, the percentage of the vote to which such <u>Party</u> and such Group of Parties is entitled shall be reduced by such percentage of the vote ("Extra

Deleted: Suppliers

Deleted: its

6.9.1 subject to Clause 6.9.2, the percentage of the vote to which such <u>Party and such</u> Group of Parties is entitled shall be reduced by such percentage of the vote ("Extra Votes") as will give each such <u>Party and each such Group of Parties 20% of the Total</u> Weighted Vote, such Extra Votes to be reallocated to the remaining <u>Parties in its category taking part in that election in accordance with Clause 6.10 and added to each such <u>Parties'</u> Weighted Vote calculated in accordance with Clause 6.8. For a Group of Parties, the remaining Weighted Votes shall be allocated equally between <u>Parties</u> in the Group of Parties; or</u>

Deleted: Supplier

Deleted: Supplier

Deleted: Suppliers

- Deleted: Suppliers'

 Deleted: Suppliers
- 6.9.2 where there are five or less <u>Parties</u> (not being in a Group of Parties) and Groups of Parties in a category of Parties, each such <u>Party</u> and Group of Parties shall be deemed to be entitled to 20% of the Total Weighted Vote in respect of any election pursuant to Clause 6.7 and Clause 6.8 and the foregoing provisions of Clause 6.9.1 shall not apply. Where this Clause 6.9.2 applies, each such <u>Party</u> and <u>Group of Party shall</u> only be entitled to cast up to half of its Weighted Vote determined pursuant to this Clause 6.9.2 (i.e. 10%) for any one candidate proposed for election to the SPAA EC pursuant to Clause 6.7 and must therefore cast its remaining Weighted Votes for another candidate.

Where Clause 6.9.1 applies, any Extra Votes shall be reallocated to each of the other <u>Parties</u>

in the same category of Parties, (provided that they are not part of a Group of Parties which

Deleted: Suppliers

Deleted: y

Deleted: Supplier

Deleted: y

Deleted: 6.8

Inserted: 6.8

Deleted: 6.8

Deleted: Supplier

Deleted: Suppliers

Deleted: v

together holds in excess of 20% of the Total Weighted Vote) to which the Extra Votes relate who have less than 20% of the Total Weighted Vote in accordance with the following formula:

$$EXV = EV x \frac{N}{XN + X}$$

where:

6.10

"EXV" means the additional percentage of the vote which are to be added to the percentage of vote held by a <u>Party</u> in that category of <u>Parties</u>, holding less than 20% of the Total Weighted Vote;

"EV" means the aggregate percentage of Extra Votes available for reallocation in accordance with Clause 6.9.1;

"N" means either:

(i) the number of MPRNs for which the <u>Party in that category of Suppliers was</u>, Registered on all Gas Transportation Databases in the month preceding the election

Deleted: Supplier

Deleted: y

Deleted: Supplier

Deleted: Part

Deleted: y

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

- which shall be determined from the Report submitted in respect of such month by that Supplier pursuant to Clause 20.2; or
- (ii) the number of MPRNs which the Party in a category of Transporters has Registered on its Gas Transportation Database in the month preceding the election which shall be determined from the Report submitted by that Transporter in respect of that month pursuant to Clause 20.2; or
- (iii) one,

whichever is the greater.

"XN" means the total number of MPRNs on all Gas Transportation Databases in the month preceding the election which shall be determined by summing the total number of MPRNs for each <u>Party</u> in that category of <u>Parties</u> whose vote has not been reduced in accordance with Clause 6.9.1 and shall be determined from the Reports submitted in respect of that month by each such <u>Party</u> pursuant to Clause 20.2;

Deleted: Supplier

Deleted: y

Deleted: Supplier

"X" means either:

the number of <u>Parties</u> in that category of <u>Suppliers</u> for whom no MRPNs were Registered on any Gas Transportation Database in the month preceding the election which shall be determined from the Report submitted in respect of that month by each such <u>Party</u> pursuant to Clause 20.2; or

Deleted: Suppliers

Deleted: Party

Deleted: Supplier

(ii) the number of Parties in that category of Transporters who had no MPRNs

Registered on their Gas Transportation Databases in the month preceding the election

which shall be determined from the Report submitted in respect of that month by each
such Party pursuant to Clause 20.2.

Deleted:

Where, as a result of the reallocation of Extra Votes in accordance with Clause 6.10, any Party in a category of Parties or Group of Parties in a category of Parties has in excess of 20% of the Total Weighted Vote ("Extra Votes"), the Extra Votes shall be reallocated in accordance with Clauses 6.8 to 6.9.1 (inclusive) and Clause 6.10, mutatis mutandis. If more than one Party in a category of Parties or Group of Parties in a category of Parties, has more than 20% of the Total Weighed Vote, this Clause 6.11 shall be applied for each Party or Group of Parties in that category of Parties, with more than 20% of the Total Weighed Vote.

Deleted: Supplier

Deleted: y

Deleted: y

Deleted: Supplier

Deleted: y

Deleted: y

Deleted: Supplier

Deleted: y

- The two I&C <u>Supplier</u> Members that receive the most percentage of the vote, or, where any two candidates are proposed, those candidates, shall be appointed as the I&C <u>Supplier</u> Members from 1st November in that year. The two Large Domestic <u>Supplier</u> Members that receive the most percentage of the vote, or, where only two candidates are proposed, those candidates, shall be appointed as the Large Domestic <u>Supplier</u> Members from 1st November in that year. The Small Domestic <u>Supplier</u> Member that receives the most percentage of the vote, or, where only one candidate, is proposed, that candidate shall be appointed as the Small Domestic <u>Supplier</u> Member from 1st November in that year. The large Transporter Member that receives the most percentage of the vote, or, where only one candidate is proposed, that candidate shall be appointed as the Large Transporter Member from 1st November in that year. The Small Transporter Member that receives the most percentage of the vote, or, where only one candidate is proposed, that candidate shall be appointed as the Small Transporter Member from 1st November in that year.
- 6.13 All SPAA EC Members shall retire on 1 November next following their appointment as SPAA EC Members, but each retiree may be a candidate for reappointment in respect of the following year in accordance with Clause 6.7.

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

6.14 Each category of Parties, entitled to appoint a SPAA EC Member pursuant to Clause 6.7 to Clause 6.11 (inclusive) may where those Parties, in that category of Parties holding Weighted Votes in aggregate totalling more than 50% of the Total Weighted Vote of the category of Parties, that appointed such SPAA EC Member, agree at any time, remove the SPAA EC Member from office and elect or appoint another individual to be a SPAA EC Member in his place. A category of Parties, will only have the right to remove from office a SPAA EC Member which it or they have elected or appointed, and will have no right to remove from office any SPAA EC Member elected or appointed by another category of Parties, Any appointment to replace a SPAA EC Member removed from office pursuant to this Clause 6.14 shall be made in accordance with the procedure set out in Clause 6.7 to Clause 6.11 (inclusive), but on such timescale as the Secretary shall reasonably direct. Only Parties who are Parties at the point in time when the existing SPAA EC Member is removed pursuant to this Clause shall be entitled to nominate candidates and to vote.

Deleted: v

Deleted: y

Deleted: y

Deleted: y

Deleted: y

If, at any time, a vacancy arises in any category of SPAA EC Member otherwise than as a 6.15 result of retirement in accordance with Clause 6.13 or removal in accordance with Clause 6.14, those Parties in the category of Parties represented by the outgoing SPAA EC Member who are Parties at the point in time when the vacancy arises and entitled to appoint such SPAA EC Member may elect a replacement. Any election to replace a SPAA EC Member pursuant to this Clause shall be conducted in accordance with the procedure set out in Clause 6.7 to Clause 6.11 (inclusive), but on such timescale as the Secretary shall reasonably direct.

Deleted: v

6.16 If at any time any category of Parties, fails to provide a SPAA EC Member, the Secretary shall request the Authority to make the appointment and the Authority shall have the power, until the category of Parties, has decided upon an appointment and notified the Authority accordingly, to appoint a SPAA EC Member on behalf of that category of Parties or to remove any such person so appointed by the Authority.

Deleted: v

Deleted: v

Deleted: y

From the Effective Date until 1st November 2004, SPAA EC Members shall be [6.17 as the Large Domestic Members. as the I&C Members, [as the Small Domestic Member, as the Large Transporter Member and [las the Small Transporter Member

Deleted: 2003

Deleted: and

Deleted:

Alternates

6.18 Each SPAA EC Member shall have the power to appoint any individual who is either an employee of a Party or its Affiliates, from the category of Parties, that has appointed him, to be his alternate provided that no SPAA EC Member may appoint an alternate who is an employee of any Party or its Affiliate which already has one of its employees acting as a SPAA EC Member or alternate. Each SPAA EC Member may, at his discretion, remove an alternate so appointed and shall remove that alternate, in circumstance as described in Clause 6.45 as if the alternate were himself a SPAA EC Member. Any appointment or removal of an alternate shall be effected by notice in writing executed by the appointor and, delivered to the Secretary or tendered at a meeting of the SPAA EC. If his appointor so requests, an alternate shall be entitled to receive notice of all meetings of the SPAA EC which take place while his appointor is a SPAA EC Member. An alternate shall also be entitled to attend and vote as the SPAA EC Member at any such meeting at which the SPAA EC Member appointing him is not personally present and at the meeting to exercise and discharge all the functions, powers and duties of his appointor as a SPAA EC Member and for the purpose of the proceedings at the meeting the provisions of this Clause 6 shall apply as if he were a SPAA EC Member.

Deleted: y

6.19 Every person acting as an alternate shall exercise the voting rights of his appointor. Execution by an alternate of any resolution in writing of SPAA EC shall, unless the notice of his appointment provides to the contrary, be as effective as execution by his appointor.

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

6.20 When a SPAA EC Member that has appointed an alternate pursuant to Clause 6.18, ceases to be a SPAA EC Member for any reason, provided that the alternate's employer is still a <u>Party</u> and has not had its Gas Suppliers <u>Licence or Gas Transporters</u>, <u>Licence (as appropriate)</u> revoked, the alternate shall discharge all the functions, powers and duties of his appointor until a replacement SPAA EC Member is appointed pursuant to Clause 6.7.

Deleted: Supplier

Deleted: y

6.21 References in this Clause 6 to a SPAA EC Member shall, unless the context otherwise requires, include his duly appointed alternate.

The SPAA EC Chairman

- 6.22 The SPAA EC Chairman shall be a SPAA EC Member and shall be appointed by a simple majority of the SPAA EC Members.
- 6.23 The SPAA EC Chairman may at any time be removed from office by a simple majority of SPAA EC Members.
- 6.24 The SPAA EC Chairman shall preside at every meeting of SPAA EC at which he is present. If the SPAA EC Chairman is unable to be present at a meeting, he may nominate another SPAA EC Member (or any alternate appointed pursuant to Clause 6.18) to act as SPAA EC Chairman. If neither the SPAA EC Chairman nor his alternate is present within half an hour after the time appointed for holding the meeting, the SPAA EC Members present may appoint by simple majority any of their number to be the SPAA EC Chairman of that meeting.
- 6.25 The SPAA EC Chairman, or the person appointed to act as the SPAA EC Chairman in accordance with Clause 6.24, shall be entitled to vote in his capacity as a SPAA EC Member. The SPAA EC Chairman shall in no circumstances be entitled to an extra or casting vote.

The Secretary

- 6.26 The Secretary shall be appointed to or removed from office by a resolution of SPAA EC Members. In no event shall a Secretary be an employee of any Party or any of its Affiliates. The Secretary shall be entitled to speak but not to vote on any issue at a SPAA EC meeting or SPAA Forum meeting.
- 6.27 The Secretary's duties shall be to facilitate the SPAA EC and any industry body approved by SPAA EC and in particular to:
 - 6.27.1 attend to the requisition of meetings and to serve requisite notices;
 - 6.27.2 maintain a register of names and addresses of SPAA EC Members and alternates as appointed from time to time;
 - 6.27.3 keep minutes of all meetings;
 - 6.27.4 circulate all relevant notices, papers and minutes; and
 - 6.27.5 manage the process for the appointment of SPAA EC Members pursuant to Clauses 6.3 to 6.16.

Meetings

- 6.28 The SPAA EC shall hold meetings at such times as it may decide but in any event shall meet at least once every three months. The venue for meetings shall be determined by the SPAA EC Members from time to time.
- 6.29 Any SPAA EC Member may, by giving notice in writing to the Secretary, request the Secretary to requisition further meetings. The notice given to the Secretary shall contain a list

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

of matters to be included in the agenda of the meeting to be convened pursuant to this paragraph. The Secretary shall proceed to convene meetings of SPAA EC within 10 Working Days of such a notice and shall circulate a copy of the agenda which shall contain such items as are contained in the notice of meeting.

6.30 A quorum will be one (1) I&C Member, one (1) Large Domestic Member, one (1) Small Transporter Member and one (1) Small Transporter Member.

Deleted: and

6.31 Nothing in the above shall preclude a meeting being held by telephone or other technological means, subject to all the criteria regarding notice and minutes being kept.

Notice of Meetings

- 6.32 All meetings of the SPAA EC shall be convened by the Secretary on at least 10 Working Days' notice. Notice may be given by e-mail.
- 6.33 The notice of each SPAA EC meeting shall contain the time, venue and confirmation of date of the meeting and an agenda and any available supporting papers which shall be given to each SPAA EC Member, all Parties, the Authority and energywatch.

By notice to the Secretary, any SPAA EC Member may request matters to be considered at an SPAA EC meeting and provided that such notice is given at least 5 Working Days before the date of the meeting, those matters will be included in the agenda for the meeting. Any such request may be given by e-mail. If necessary, the Secretary shall circulate a revised agenda to each SPAA EC Member and all Suppliers as soon as practicable.

6.35 The accidental omission to give notice of a meeting to, or the non-receipt of notice of a SPAA EC meeting by, a person entitled to receive notice shall not invalidate the proceedings of that meeting.

Proceedings of Meetings

6.36 SPAA EC may meet for the transaction of business, and adjourn and otherwise regulate its meetings as it thinks fit, but shall at all times act reasonably and in compliance with the other provisions of this Clause 6.

Representation and Voting

- 6.37 Each SPAA EC Member shall be entitled to attend, speak and vote, at every meeting of SPAA EC. The Authority and energywatch shall be entitled to send a representative to any meeting who shall be entitled to speak but not vote at the meeting. Any Party shall be entitled to send a representative to attend a SPAA EC meeting provided it gives the Secretary 2 Working Days' written notice in advance of such meeting. A representative of Party attending a SPAA EC meeting shall only be admitted as an observer and shall not be entitled to speak or vote at the meeting. The Chairman may, in his sole discretion, exclude a representative of a Party from a meeting (or any part thereof) where matters being discussed are confidential.
- 6.38 All decisions of the SPAA EC shall be by resolution. A resolution put to the vote of any meeting of SPAA EC to be passed, it shall require the unanimous support of all SPAA EC Members present at the meeting entitled to vote in relation to that resolution.
- 6.39 A resolution in writing signed by or on behalf of all the SPAA EC Members entitled to vote in relation to that resolution shall be as valid and effective as if the same had been passed at a meeting of SPAA EC duly convened and held, and may consist of several instruments in like form executed by or on behalf of one or more SPAA EC Members.

Deleted: and to

Deleted: SPAA_180204.DOC
Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

6.40 Any resolution passed by SPAA EC shall have no effect until the expiry of any period in which a <u>Party</u> is entitled to appeal that decision pursuant to Clause 6.44, or such later date as the terms of such resolution may provide, as the case may be.

Deleted: Supplier

Minutes

6.41 The Secretary shall circulate copies of the minutes of each meeting of the SPAA EC or any sub-committees of SPAA EC to each SPAA EC Member, all Parties, the Authority and energywatch as soon as practicable (and in any event within 5 Working Days) after the relevant meeting has been held. If any SPAA EC Member disagrees with any item of the minutes, he shall, within 5 Working Days of receipt of the minutes, notify the Secretary of those items with which he disagrees, and the Secretary shall incorporate those items upon which there is disagreement into the agenda for the next following meeting of SPAA EC, as the first item for resolution.

Deleted: and

Deleted: 5 (E)

Deleted: Supplier

Deleted: Supplier

Inserted: 5.(E)

Deleted: y

6.42 The Secretary shall maintain a record of all resolutions voted on by SPAA EC, indicating how each SPAA EC Member voted on each resolution and shall make such record available on request to any Party.

Appeals

- 6.43 Where any resolution put to the vote at any meeting of SPAA EC is not passed, SPAA EC shall, if requested by any SPAA EC Member who voted in favour of such resolution, within 10 Working Days after receipt of the minutes of the SPAA EC meeting setting out such resolution pursuant to Clause 6.41 appeal the SPAA EC decision to the SPAA Forum for its determination.
- Where pursuant to Clause <u>5.4(E)</u> or where a <u>Party</u> reasonably believes that a resolution passed by SPAA EC, or SPAA EC's failure to pass any resolution put to the vote at any meeting of SPAA EC, will or is likely to prejudice unfairly the interests of that <u>Party</u> or will cause that Party to be in breach of this Agreement, of its Gas Suppliers <u>Licence or Gas Transporters Licence (as appropriate)</u>, the Act or any other legal requirement, it may within 10 Working Days, or such longer period as SPAA EC may decide in relation to that resolution, of receiving the minutes of the relevant SPAA EC meeting pursuant to Clause 6.41 appeal the SPAA EC decision to the SPAA Forum for its determination. Pending the outcome of any such appeal, the relevant decision shall have no effect.

Vacation of Office

- 6.45 The office of a SPAA EC Member shall be vacated forthwith if:
 - 6.45.1 he resigns his office by notice in writing delivered to the Secretary;
 - 6.45.2 he fails, in person or by alternate, to attend 3 consecutive meetings of SPAA EC that have been duly convened but have not been held as a result of a lack of quorum due to his (or his alternate's) non-attendance;
 - 6.45.3 a Party ceases to be a Party and the SPAA EC Member is employed by either:
 - (i) that Party; or
 - (ii) an Affiliate of that Party; and
 - (iii) such Party is in the category of Parties, that has appointed him under Clause 6.3.1 to 6.3.5;

Deleted: y

Deleted: 3
Inserted: 3

Deleted: 6.31

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

6.45.4 he ceases to be in the employment of one of the Parties, or an Affiliate of one of the Parties, and such Party is in the category of Parties, that has appointed him under Clause 6.3.1 to 6.3.5; or

6.45.5 a Party has its Gas Suppliers Licence or Gas Transporters Licence (as appropriate) (or

Deleted: y

Deleted: 3

Inserted: 3

Deleted: 6.3

Deleted: y

relevant section thereof) revoked and the SPAA EC Member is employed by either:

that Party; or

(i)

- (ii) an Affiliate of that Party; and
- (iii) such Party is in the category of Parties that has appointed him under Clause 6.3.1 to 6.3.5.

Deleted: y

Deleted: 3

Inserted: 3

SPAA EC Member Responsibilities and Protections

- 6.46 In the exercise of his powers and the performance of his duties and responsibilities as a SPAA EC Member, each of the members shall represent the interests of the category of Parties, by whom they are for the time being appointed. Each SPAA EC Member shall exercise reasonable skill and care to the standard reasonably expected of a director of a limited company in the performance of his duties and responsibilities as a SPAA EC Member.
- 6.47 The SPAA EC Members shall use their reasonable endeavours to consult as many of the Parties that they represent as possible before voting on a matter, and shall demonstrate this on request.
- 6.48 SPAA EC Members will vote according to the consultations that they have made and may be called to account for those decisions. Where a consultation of the relevant Parties shows that Parties have varying opinions, the SPAA EC Member must act in the best interests of the majority, whilst representing the minority view. Where a majority is not significant, the SPAA EC Member should consider whether abstention from the vote best represents the interests of her/ his constituents.
- 6.49 All Parties shall jointly and severally indemnify and keep indemnified:
 - 6.49.1 each SPAA EC Member and his alternate (including in his capacity as a director or alternate director of SPAA Ltd);
 - 6.49.2 the Secretary;
 - 6.49.3 the Company Secretary of SPAA Ltd;
 - 6.49.4 each person who serves on a sub-committee established by SPAA EC or the board of directors of SPAA Ltd;
 - 6.49.5 each member of the Secretariat who is employed by a Party; and
 - 6.49.6 each Party, or Affiliate of that Party, which is the employer of any person referred to in paragraphs 6.49.1 to 6.49.5 above,

as between each such Party rateably in accordance with the proportions set out in Clauses 8.8 and 8.9 from and against all and any costs (including legal costs), charges, expenses, damages or other liabilities properly incurred or suffered by the person or Party in relation to such function or the due exercise of the person's powers, duties or responsibilities under this Agreement (in the case of the Secretariat, as assigned or vested in it by SPAA EC pursuant to Clause 6.54) including in relation to negligence and all claims, demands or proceedings

Deleted: y

Deleted: SPAA_180204.DOC
Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

arising out of or in connection with the same except for any costs and expenses which are recovered in accordance with the procedures set out in Clause 8 and any such costs, charges, expenses, damages or other liabilities which are recovered under any policy of insurance in favour of any or all of the persons and Parties referred to in paragraphs 6.49.1 to 6.49.5 or suffered or incurred or occasioned by the wilful default or bad faith of, or breach of contract by, the relevant person.

6.50 The Parties hereby ratify and confirm the decisions of the Shadow SPAA EC taken prior to the Effective Date and those Parties providing the indemnity under Clause 6.49 confirm such indemnity shall extend fully to all costs, charges, expenses, damages and other liabilities suffered or incurred by Shadow SPAA EC Members and their alternates before or after the date of this Agreement in relation to activities of the Shadow SPAA EC.

Sub-Committees

- 6.51 SPAA EC may establish such sub-committees from time to time and consisting of such persons as it considers desirable. Each sub-committee shall be subject to such written terms of reference and such procedures as SPAA EC may determine.
- 6.52 The Authority shall be entitled to send a representative to any meeting of any sub-committee, who shall be entitled to speak but not to vote on any issue.
- 6.53 Resolutions of sub-committees shall not have binding effect unless SPAA EC has formally delegated the decision-making powers to the sub-committee or has ratified the resolution in question.

Secretariat

- 6.54 SPAA EC may from time to time appoint and remove, or make arrangements for the appointment and removal of, any such person as SPAA EC requires to assist it or any sub-committee of it, the SPAA EC Chairman or Secretary in the proper performance of its or their duties and responsibilities in each such case upon such terms and conditions as SPAA EC sees fit (any such person or persons to be known as the "Secretariat"). In no event shall the Secretariat be a Party or any of its Affiliates, or an employee of a Party or any of its Affiliates.
- 6.55 Any person referred to in Clause 6.54 shall undertake such administrative duties and responsibilities and exercise such powers as SPAA EC may from time to time assign to or vest in any such person.
- 6.56 SPAA EC may make arrangements for the remuneration of any such person as is referred to in Clause 6.54 and the payment of any such person's costs and expenses and the same shall be recovered in accordance with Clause 8.

Confidentiality

6.57 Prior to the appointment of any person as a SPAA EC Member, Secretary, Secretariat or to a member of a SPAA EC sub-committee, such person shall execute a confidentiality undertaking on the same terms as those contained in Clause, 1 last if in each such Clause there was substituted for the name of the Party the name of the person providing the undertaking under this Clause 6.57.

7. CONSTITUTION OF THE SPAA FORUM

Objects

Deleted: .1 to 11.

Deleted: 7

Deleted: 10

Inserted: 10

Deleted: SPAA_180204.DOC

Deleted: s

March.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers
amended draft) 18

| | 7.1 | and fo | urpose of the SPAA Forum is to or informing the Parties genera- nine any matters from time to time | ally regarding the | operation of this A | | | |
|----------|-----|--|--|--------------------------------------|------------------------------|--------------------|---------------------------|--|
| | 7.2 | | ng in this section shall precological means, subject to all the | | | | | |
| | | Memb | oership | | | | | |
| | 7.3 | | Party shall be entitled to send of SPAA Forum on its behalf to re | | representative to a | ttend any meeting | Deleted: Supplier | |
| | 7.4 | | et to Clause 7.22, each represent o speak and to vote at such mee | | ursuant to Clause | 7.3 shall have the | Deleted: s | |
| | | Proxie | es | | | | | |
| | 7.5 | | epresentative of a Party entitled d to appoint another person as it | | | | | |
| | 7.6 | | nstrument appointing a proxy sised representative of the Party. | | | | | |
| | 7.7 | | strument appointing a proxy an it is signed or a certified copy of | ority, if any, under | | | | |
| | | 7.7.1 shall be deposited at the office of the Secretary or at such other place within the United Kingdom as is specified for that purpose in the notice convening the meeting of the SPAA Forum, not less than 24 hours before the time appointed for the taking of the vote and in default the instrument of proxy shall not be treated as valid; or | | | | | | |
| | | 7.7.2 if sent by e-mail, shall be produced in a scanned image file (.PDF format) sent to [insert e-mail address] or such other e-mail address as is specified for that purpose in the notice convening the meeting of the SPAA Forum, not less than 24 hours before the time appointed for the taking of the vote and in default the instrument of proxy shall not be treated as valid. In order for the instrument appointing proxy to be deemed valid, the e-mail that sends such instrument shall be required to quote the code or identifier assigned to that <u>Party</u> by the Secretary. | | | | | | |
| 1 7.8 | 7.8 | | strument appointing a proxy shanstances admit: | all be in the following | ing form or a form | as near thereto as | | |
| | | the SP. | ment, hereby appoint the Chairn or, of AA Forum meeting to be held onment thereof. | nan of the Meeting fa , as our | iling proxy to vote for u | of him, | | |
| | | | | FOR | AGAINST | | | |
| | | | Resolution 1 | | | | | |
| | | | | | | | Deleted: SPAA_180204.DOC | |
| | | | Resolution 2 | | | i | Inserted: SPAA_180204.DOC | |

(see Note 2)

Deleted: SPAA (suppliers amended draft) 18
March.DOC

Signed (name

this day of 20

Notes

- 1. A Party may appoint a proxy of its own choice. If such an appointment is made, delete the words 'Chairman of the Meeting' and insert the name of the person appointed as proxy in the space provided
- If this form is returned without any indication as to how the person appointed as proxy should vote, he may exercise his discretion as to how he votes or whether he abstains from voting."
- 7.9 A vote given in accordance with the terms of an instrument of proxy shall be valid notwithstanding the previous revocation of the proxy or of the authority under which the proxy was executed, provided that no notice in writing of such revocation shall have been received by the Secretary (including by e-mail provided such e-mail quotes the code or identifier assigned by the Secretary to the Supplier seeking to revoke the proxy) at his office before the commencement of the meeting or adjourned meeting at which the proxy is used.

Meetings

- 7.10 Subject to Clauses 7.12 to 7.14 meetings of the SPAA Forum shall be convened:
 - 7.10.1 by the Secretary on the instructions of SPAA EC pursuant to Clauses 8.3 or upon receipt of notice of an appeal pursuant to Clauses 6.43 or 6.44 or Clauses 5.13.4 or 5.15 or Clauses 9.15, 9.16, 9.17, or 9.18;

Deleted: 5.11.4 **Deleted:** 5.13

7.10.2 by a Party if the Secretary fails to convene a meeting of the SPAA Forum to hear an appeal of that Party pursuant to Clauses 6.43 or 6.44 or if it makes an appeal pursuant to Clauses 5.13.4 or 5.15 or Clauses 9.15, 9.16, 9.17 or 9.18;

Deleted: 5.11.4

7.10.3 by SPAA EC, forthwith upon receipt of a requisition by 4 Parties; or

Deleted: 5.13

Deleted: Suppliers

7.10.4 by the Secretary in any event, at least once every 12 months if no meeting has been convened pursuant to Clauses 7.10.1 to 7.10.3.

Deleted: Suppliers

7.11 Any requisition by <u>Parties</u> as referred to in Clause 7.10.3 shall state the objects of the meeting and must be signed by or on behalf of each of the requisitioners and deposited with the Secretary, and may consist of several documents in like form each signed by one or more requisitioners. If SPAA EC does not within 5 Working Days from the date of the deposit of the requisition proceed to convene a meeting of the SPAA Forum for a date not later than 15 Working Days after the date of deposit, the requisitioners may themselves convene a meeting provided that such meeting is held within a further 20 Working Days. Any meeting convened in accordance with this Clause shall be convened in the same manner, as nearly as possible, as that in which meetings of the SPAA Forum are convened by the Secretary.

Notice of Meetings

7.12 All meetings of the SPAA Forum shall be convened by the Secretary on at least 10 Working Days' notice (or such other period of notice as SPAA EC may determine) to those entitled to attend pursuant to Clauses 7.3 and 7.4. the Authority and energywatch. Notice may be given by e-mail.

Deleted: 3

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

Deleted: 2

7.13 The notice shall specify the date, time and venue of the meeting, an agenda setting out the business to be transacted together with a categorisation of each item on the agenda as being relevant to, the Domestic Suppliers, the Industrial and Commercial Suppliers, to all Suppliers to Large Transporters, to Small Transporters or to all Transporters. Any item on the agenda concerning approval of a draft budget revision to a budget and/or any limits to be granted to the SPAA EC in respect of budget revisions, and/or any item contained in an appeal in respect of a Change Proposal in respect of any provision in Parts I to VI of this Agreement and/or Schedules 2, 3, or 4 or 5 shall be deemed to concern all <u>Parties</u>. Notice shall be given to all Parties, all SPAA EC Members, the Authority and energywatch. If any Party wishes to dispute the categorisation of an item on the agenda of business to be transacted at a meeting of the SPAA Forum, it may do so by sending its grounds for dispute to the SPAA EC at least 8 Working Days before the meeting of the SPAA Forum. Any such notice may be sent by email. The SPAA EC shall consider the disputed categorisation and shall inform all the Suppliers if the disputed categorisation is upheld at least 5 Working Days prior to the meeting of the SPAA Forum. This notice may be sent by e-mail and/or posted on the Gas Forum website.

Deleted: or

Deleted: X

Deleted: Suppliers **Deleted:** Suppliers

Deleted: Supplier

7.14 The accidental omission to give notice of a meeting to, or the non-receipt of notice by, any person entitled to receive notice shall not invalidate the proceedings at the meeting.

Proceedings at Meetings

- 7.15 All business of the SPAA Forum shall be transacted at meetings of the SPAA Forum.
- 7.16 At the first meeting of the SPAA Forum held after 1 April in each year, the SPAA Forum shall elect from its number, by simple majority of those representatives present (including by proxy), a person to act as Forum Chairman ("Forum Chairman") for the following twelve months. The Forum Chairman shall preside at each meeting of the SPAA Forum at which he is present. If the Forum Chairman is unable to be present at a meeting of the SPAA Forum, the <u>Parties</u> who are represented at the meeting shall elect one of their number to act as Forum Chairman of that meeting. The Forum Chairman of any meeting of the SPAA Forum shall not be entitled to any casting vote in his capacity as such.

Deleted: Suppliers

Deleted:

- 7.17 A quorum at meetings of the SPAA Forum shall be:
 - 7.17.1 persons representing (including by proxy) Domestic Suppliers together holding Weighted Votes totalling 50% or more of the Total Weighted Votes of Domestic Suppliers;
 - 7.17.2 persons representing (including by proxy) Industrial and Commercial Suppliers together holding Weighted Votes totalling 50% or more of the Total Weighted Votes of Industrial and Commercial Suppliers.

ghted

7.17.3 persons representing (including by proxy) Transporters together holding Weighted Votes totalling 50% or more of the Total Weighted Votes of Gas Transporters.

and in each case, the Weighted Vote of each <u>Party in Clauses 7.17.1_7.17.2</u> and 7.17.3 shall be determined by the Secretary from the Report submitted by each <u>Party in respect of the previous month pursuant to Clause 20.2</u>.

7.18 If within half an hour from the time appointed for holding any meeting of the SPAA Forum a quorum is not present or during the course of a meeting the meeting becomes inquorate, the meeting shall be adjourned to a time and place reasonably determined by the Forum Chairman and, where the meeting is adjourned until later the same day, communicated to those present at the meeting. Where the meeting is adjourned to another date, notice of the adjourned meeting shall be given to all <u>Parties</u>, as if it were a new meeting. If, where the meeting is adjourned to another date, at the adjourned meeting a quorum is not present within half an

Deleted: Supplier

Deleted: Supplier

Deleted: and

Deleted: Suppliers **Deleted:** SPAA_180204.DOC

Deleted: SFAA_100204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

hour from the time appointed, those <u>Parties</u> represented at the adjourned meeting shall constitute a quorum.

Deleted: Suppliers

7.19 The Forum Chairman at a meeting of the SPAA Forum at which a quorum is present may, with the consent of the meeting (and shall if so directed by the meeting), adjourn the meeting from time to time and from place to place, provided that no business shall be transacted at any adjourned meeting other than the business left unfinished at the previous meeting. When a meeting is adjourned to another date, notice of the adjourned meeting shall be given as if it were a new meeting, but it shall otherwise not be necessary to give notice of an adjourned meeting.

Formatted: Heading 2, Justified

Resolutions and Voting

- 7.20 At any meeting of the SPAA Forum, a resolution put to the vote of the meeting shall be taken in such manner as the Forum Chairman of the meeting directs and the result of the vote shall be deemed to be the resolution of the meeting and recorded in the minutes.
- 7.21 A declaration by the Forum Chairman of the meeting that a resolution has on a vote been carried or lost and an entry to that effect in the book containing minutes of the proceedings of meetings of the SPAA Forum shall be conclusive evidence of the fact.
- 7.22 A resolution of the SPAA Forum will be carried where:
 - 7.22.1 subject to Clause 7.22.6, for a matter relating to the Domestic Suppliers only, Domestic Suppliers' representatives together casting Weighted Votes totalling 65% or more of the Total Weighted Vote of Domestic Suppliers in each case present or represented (including by proxy) at the meeting; or

Deleted: 4

7.22.2 subject to Clause 7.22 6, for a matter relating to the Industrial and Commercial Suppliers only, Industrial and Commercial Suppliers' representatives together casting Weighted Votes totalling 65% or more of the Total Weighted Vote of Industrial and Commercial Suppliers in each case present or represented (including by proxy) at the meeting; or

Deleted: 4

7.22.3 subject to clause 7.22.6, for a matter relating to both Domestic Suppliers and Industrial and Commercial Suppliers, Domestic Suppliers' representatives together casting Weighted Votes totalling 65% or more of the Total Weighted Vote of Domestic Suppliers, and Industrial and Commercial Suppliers' representatives together casting Weighted Votes totalling 65% of the Total Weighted Vote of Industrial and Commercial Suppliers, in each case present or represented (including by proxy) at the meeting; or

Deleted: 4

- 7.22.4 subject to clause 7.22.6, for a matter relating to the Transporters only, Transporters' representatives together casting Weighted Votes totalling 65% or more of the Total Weighted Vote of Transporters in each case present or represented (including by proxy) at the meeting; or
- 7.22.5 subject to clause 7.22.6, for a matter relating to all Parties, Domestic Suppliers' representatives together casting Weighted votes totalling 65% or more of the Total Weighted Vote of Domestic Suppliers, Industrial and Commercial Suppliers' representatives together casting Weighted Votes totalling 65% or more of the Total Weighted Vote of Industrial and Commercial Suppliers, and Transporters' representatives together casting Weighted Votes totalling 65% of the Total Weighted Vote of Transporters, in each case present or represented at the meeting.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

7.22.6 for a matter relating to a Mandatory Schedule, or any change of a Voluntary Schedule and/or Elective Schedule to a Mandatory Schedule, or the introduction of a new Mandatory Schedule:

(A) Domestic Suppliers' representatives together casting Weighted Votes totalling 65% of the Total Weighted Vote, Industrial and Commercial Suppliers' representatives together casting Weighted Votes totalling 65% of the Total Weighted Vote, and Transporters' representatives together casting Weighted Votes totalling 65% of the Total Weighted Vote (provided in each case the category of Parties has an interest in the resolution); and

Deleted: and

Deleted: Supplier

(B) 65% or more by number of representatives of Domestic Suppliers (not being in a Group of Domestic Suppliers) and Groups of Domestic Suppliers' representatives, 65% or more by number of representatives of Industrial and Commercial Suppliers (not being in a Group of Industrial and Commercial Suppliers) and Groups of Industrial and Commercial Suppliers' representatives, and 65% or more by number of representatives of Transporters (not being in a Group of Transporters) and Groups of Transporters' representatives (provided in each case the category of Parties has an interest in the resolution),

Deleted: and

Deleted: Supplier

in each case present or represented (including by proxy) at the meeting.

On a vote, each <u>Parties'</u> representative shall be entitled to Weighted Votes of the <u>Party</u> whom it is representing, calculated in accordance with Clauses 6.8 to 6.11 (inclusive). A <u>Party</u> shall be entitled to vote on matters in which they have an interest as determined pursuant to Clause 7.13 or on matters affecting all Parties.

Deleted: Suppliers'

Deleted: Supplier

Deleted: Supplier

Deleted: Suppliers

7.24 The Authority shall be entitled to send a representative to any <u>SPAA Forum</u> meeting who shall be entitled to speak but shall not be entitled to vote. energywatch shall be entitled to send a representative to any SPAA Forum meeting as an observer who shall <u>be entitled to speak but shall</u> not be entitled to vote.

Deleted: but shall be entitled to speak

Minutes

7.25 The Secretary shall prepare minutes of all meetings of the SPAA Forum and shall circulate copies of such minutes to all <u>Parties</u>, each <u>SPAA EC Member</u>, <u>energywatch</u> and the Authority as soon as practicable (and in any event within 10 Working Days) after the meeting has been held. If any representative of <u>a Party or any SPAA EC Member disapproves of the minutes</u>, he shall, within 10 Working Days of receipt of those minutes, notify the Secretary of those aspects with which he disagrees and the Secretary shall incorporate those aspects of the minutes upon which there is disagreement into the agenda for the next following meeting of the SPAA EC and shall be copied to all <u>Parties</u>, each <u>SPAA EC Member</u>, energywatch and the Authority.

Deleted: Suppliers

Deleted: the

Deleted: Suppliers

Deleted: Suppliers

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: Supplier

Deleted: y

Deleted: Supplier

Deleted: 10

Deleted: Supplier's

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers

amended draft) 18

March.DOC

Appeals

Where a <u>Party</u> reasonably believes that a resolution passed by the SPAA Forum pursuant to Clause 7.22 or the SPAA Forum's failure to pass any resolution put to the vote at any meeting of the SPAA Forum, and where such resolution related to provisions of this Agreement which have been designated to be Mandatory, or the introduction of a Mandatory Schedule, or the designation of any part of this Agreement as Voluntary, Elective or Mandatory and (i) will or is likely to prejudice unfairly the interests of that <u>Party</u>, or (ii) will cause that <u>Party</u> to be in breach of this Agreement or (iii) will cause that <u>Party</u> to be in breach of its Gas Suppliers, Licence or Gas Transporters Licence (as appropriate) or (iv) the Act, such <u>Party</u> may within <u>15</u> Working Days of receiving notice of the decision pursuant to Clause 7.25 appeal the matter to the Authority. The Authority will then consider the <u>Party's grounds</u> for appeal and

SPAA 040304.DOC

29

determine whether to overturn or uphold the SPAA Forum's decision. The Authority's decision shall be final and binding. Pending the outcome of any such appeal, the resolution shall have no effect.

SPAA Ltd

7.27 The Parties agree that Schedule 4 shall apply with regard to the regulation of their rights and obligations as shareholders in SPAA Ltd.

8. COSTS

- 8.1 SPAA EC shall be entitled to recover, in accordance with the procedures set out in this Clause 8, all its reasonable costs and expenses properly incurred, which may include:
 - 8.1.1 any general administration costs associated with SPAA EC, the SPAA Forum and the Secretariat including any costs incurred in holding any meetings; and
 - 8.1.2 any costs and expenses of any consultant or adviser retained by SPAA EC in the proper performance of his duties and responsibilities.

Deleted: its or

Preparation and Approval of Budgets

8.2 Not earlier than 60 nor later than 40 Working Days before the commencement of each Financial Year except for the Financial Year starting on the Effective Date, SPAA EC shall circulate to all representatives on the SPAA Forum a draft budget for that Financial Year, which shall set out SPAA EC's good faith estimate of the costs that are anticipated to be incurred pursuant to Clause 8.1 over that Financial Year. The draft budget for the Financial Year commencing on the Effective Date and ending on the 31st March of the year next following the Effective Date shall be prepared by the Shadow SPAA EC. Such budget shall be presented for approval pursuant to Clause 8.3 to a meeting of the SPAA Forum convened on the date of execution of this Agreement. In no event shall such draft initial budget exceed £200,000. The Parties hereby waive the requirements of Clauses 7.12 and 7.13 in respect of such SPAA Forum meeting.

Approval of budgets

- 8.3 The draft budget shall be presented to the SPAA Forum for approval by resolution. The SPAA Forum may, by resolution, approve the draft budget, or amend the draft budget and approve it as amended. In the event of such SPAA Forum resolution not being carried the provisions of Clause 8.4 shall apply. Where the resolution to approve the budget is carried by the SPAA Forum such budget shall be the approved budget for that Financial Year and the funding for the Secretariat in accordance with that budget shall be approved.
- 8.4 In the event of a failure of the SPAA Forum to approve the form or content of any draft budget, the following provisions shall have effect:
 - 8.4.1 pending the approval of a draft budget, SPAA EC shall not be entitled to carry out any activities which are the subject of dispute, except insofar as necessary in order to comply with legally binding obligations which it has previously incurred in accordance with this Agreement or insofar as the carrying out of such activities falls within the terms of any previous approved budget; and
 - 8.4.2 the matter shall be referred forthwith to the Authority whose decision as to the contents of the budget shall be final and binding.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

Amendments to budgets

8.5 During the course of any Financial Year SPAA EC may request any changes to be made to the approved budget. SPAA EC may approve revisions to the approved budget within limits defined from time to time by the SPAA Forum each Financial Year. The procedure for the approval of changes greater than those limits shall be the same as that set out in Clauses 8.3 and 8.4 for the approval of a draft budget.

Payment of Costs Incurred

- 8.6 SPAA EC or a named person approved by them, shall approve all costs incurred under Clause 8.1 which have been included in the approved budget for the relevant Financial Year in advance of submitting the same to the Secretariat for payment.
- 8.7 Upon receipt of an invoice or other statement relating to costs which have been approved by or on behalf of SPAA EC in accordance with Clause 8.6, the Secretariat shall pay the amount stated in such invoice or other statement (together with Value Added Tax thereon, if applicable) to such person or persons as SPAA EC shall direct. SPAA EC shall specify in advance the authority levels of the Secretariat as appropriate.

Recovery of Costs

8.8 For the purposes of Clause 8.10, the amount which each Supplier shall be obliged to pay towards the costs to be incurred in accordance with the most recent approved budget in respect of any Quarter shall be calculated as follows:

$$SP = \frac{A \times C}{T}$$

Where:

SP = the amount due from each Supplier;

- A= the average number of MPRNs of the Supplier whose SP is being calculated contained on all Gas Transportation Databases across the last three (3) months for which Reports have been submitted pursuant to Clause 20.2, which shall be determined by summing the number contained in those three Reports over all Gas Transportation Databases and dividing that figure by three (3);
- C = estimated costs for the Quarter included in the most recent budget approved pursuant to Clause 8.3 or 8.4; and
- T = the average number of MPRNs for all Suppliers held on all Gas Transportation Databases across the last three (3) months for which Reports have been submitted pursuant to Clause 20.2, which shall be determined by summing the total number of MPRNs contained in those Reports and dividing that figure by three (3).
- 8.9 The Secretariat, on behalf of the SPAA EC, shall arrange for collection from Suppliers of their respective proportionate share of the costs to be incurred in accordance with the most recent approved budget in any Quarter, calculated in accordance with Clause 8.8, as appropriate, (together with Value Added Tax thereon, if applicable) in accordance with such procedures as may be agreed by SPAA EC from time to time (which may include collection in advance) and the Suppliers shall comply with such collection procedures and, in particular, shall pay the amounts which that Supplier is obliged to pay within the time period prescribed by such procedures, following the receipt of an invoice or other statement issued by the Secretariat.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

8.10 Within 20 Working Days of the 1 April in each Financial Year the Secretariat shall calculate each Suppliers' proportionate share, in accordance with the proportions set out in Clause 8.8, of the actual costs incurred during the previous Financial Year and shall reconcile them against amounts paid by each Supplier in respect of estimated costs pursuant to Clause 8.9. Where the aggregate amount paid by the Suppliers in accordance with Clause 8.9 in respect of the previous Financial Year is greater than the aggregate amount as calculated in accordance with this Clause 8.10 in respect of that Supplier, the Secretariat shall reimburse that Supplier (as appropriate) with the difference by means of a credit against the invoice to be raised pursuant to Clause 8.9 in respect of the second quarter of the current Financial Year.

Audit

8.11 SPAA EC shall arrange for the costs incurred pursuant to Clause 8.1 to be audited by a firm of chartered accountants on an annual basis. SPAA EC shall copy the auditor's report to all Suppliers within 15 Working Days of receipt. The auditors shall be appointed in accordance with a guideline that SPAA EC will circulate amongst the Suppliers within one month of the beginning of each Financial Year.

Review of Cost Recovery mechanism

8.12 Each Supplier acknowledges that the cost recovery SPAA EC mechanism included in this Clause 8 has been agreed to on the basis that the scope of this Agreement is limited to those activities that are described in this Agreement. Each Supplier agrees to review the cost recovery SPAA EC mechanism included in this Clause 8 if the scope of this Agreement (as so described and provided for) is materially amended.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

PART III CHANGE CONTROL

9. CHANGE CONTROL

- 9.1 The Parties acknowledge and agree that, notwithstanding any other provision of this Agreement, a Change of Proposal which has been passed which amends or varies any of the matters dealt with in any of the following provisions of this Agreement shall not take effect until the written consent of the Authority has been obtained:
 - 9.1.1 Clause 4.2, Clause 4.4, Clauses 5.1 to 5.5 (inclusive), Clause 5.15, Clauses 6.1 to 6.17 (inclusive), Clause 6.30, Clause 6.37, Clause 6.38, Clause 6.40, Clause 6.43, Clause 6.44, Clause 6.48 to Clause 6.50 (inclusive), Clause 6.52, Clause 7.1 to 7.4 (inclusive), Clause 7.13, Clause 7.22 to Clause 7.26 (inclusive), Clause 8, Clause 9, Clause 10 and Clause 14.
 - 9.1.2 any change to any definition in Clause 1.1 which materially affect the provisions in the Clauses set out in Clause 9.1.1;
 - 9.1.3 any provision of this Agreement which requires or permits any matter to be referred to the Authority for approval, consent, direction or decision or confers any rights or benefits upon the Authority.
 - 9.1.4 any change to the SPAA Products;
 - 9.1.5 any change:
 - (i) to a Voluntary Schedule or Elective Schedule, that makes it Mandatory;
 - (ii) a change to an existing Mandatory Schedule; or
 - (iii) the introduction of a new Mandatory Schedule.
- 9.2 SPAA EC shall:
 - 9.2.1 give due and prompt consideration to any matter referred to it in writing by the Authority;
 - 9.2.2 if reasonably requested by the Authority, give the Authority in writing reasons for such decision or action; and
 - 9.2.3 if reasonably requested by the Authority (having regard, in particular, to the resources available to SPAA EC), in relation to any proposal for a change to any provision of this Agreement provide or procure the provision of advice and assistance to the Authority as soon as reasonably practicable as to the implications of the change and the actions necessary to implement it (including any relevant impact assessment),

provided that none of the foregoing shall in any way oblige the SPAA EC to propose a Change Proposal and shall confer no right on the Authority to raise a Change Proposal.

Change Proposals

9.3 SPAA EC may from time to time appoint and remove on such conditions it thinks fit any such person as SPAA EC requires to assist it or any sub-committee of it in connection with communications relating to, co-ordination and voting on, and implementation of, a Change Proposal ("Change Control Administrator") which person shall be part of the Secretariat (if appointed) and the SPAA EC shall inform the Parties of the identity of the Change Control Administrator.

Inserted: 4 (includsive),

Deleted: 5.1255

Deleted: Clause 5.13.

Deleted: Clause

Deleted: 9.1

Deleted: 4

Deleted: d

Deleted:, Clause 9.2 and

Deleted: SPAA_180204.DOC
Inserted: SPAA_180204.DOC
Deleted: SPAA (suppliers

amended draft) 18 March.DOC

- Each Party shall appoint an appropriate person to co-ordinate all communications from and to such Party and/or undertake all procedures in respect of a Party relating to the change control process pursuant to this Clause 9 (each "Party Change Administrator" and together the "Party Change Administrators") and shall notify the Change Control Administrator of the identity of such Party Change Administrator from time to time. A Party Change Administrator shall be fully authorised to carry out all acts or obligations of the Party that appointed it in relation to the change control process pursuant to this Clause 9. The Change Control Administrator and/or SPAA EC shall be entitled to assume that a Party Change Administrator shall be authorised to perform all acts or obligations in relation to the change control process in respect of the Party that appointed it. If a Party fails to notify a Party Change Administrator to the Change Control Administrator such party's Contract Manager shall be deemed to be its Party Change Administrator until such time as a Party Change Administrator is notified to the Change Control Administrator.
- Any Party. New Party or energywatch may propose a Change Proposal to the Change Control Administrator on the form issued by SPAA EC from time to time and shall deliver such form to the Change Control Administrator together with any other documents referred to in the form. Within a reasonable time of receipt of the Change Proposal, the Change Control Administrator shall notify the relevant Party that the Change Proposal has been accepted for logging as a Change Proposal or rejected as a Change Proposal.
 - 9.6 The Change Control Administrator may reject a Change Proposal in its absolute discretion if it considers that the Change Proposal:
 - 9.6.1 is not sufficiently specific;
 - 9.6.2 is not authorised or not properly authorised;
 - 9.6.3 is ambiguous or incomplete; and/or
 - 9.6.4 is not materially different from a Change Proposal that has already been submitted to it and not yet been voted upon pursuant to Clause 9.9.

When rejecting a Change Proposal, the Change Control Administrator may recommend any amendment and/or additional information that ought to have been provided.

- 9.7 The SPAA EC shall set out dates during each Financial Year on which Change Proposals raised during that Financial Year will be voted upon ("Change Voting Dates") and shall notify the Parties of the Change Voting Dates. The SPAA EC may change such Change Voting Dates from time to time and if it does so shall inform the Parties of such changed dates.
- 9.8 The Change Control Administrator shall, no later than 10 Working Days before the Change Voting Date, send each Change Proposal(s) that has not been rejected by it before such time to each Party, SPAA EC, the Authority and energywatch indicating, where appropriate, the proposed amendment date, proposed product release date, proposed implementation date and any product technique. The Change Control Administrator shall invite each Party to consider if it is legitimately interested or impacted by each such Change Proposal and if so, to identify any technical, business and/or implementation impact of the Change Proposal(s) on their organisation and to respond and vote by the Change Voting Date as to whether each such interested or impacted Party approves or rejects the Change Proposal(s).
- 9.9 Each Party that considers it is legitimately interested in or impacted by a Change Proposal shall before the Change Voting Date submit to the Change Control Administrator, by e-mail, its vote as to whether a Change Proposal(s) should be approved or not, together with any comments it may have on that Change Proposal(s), on the form issued by SPAA EC from time to time. Each such Party which in accordance with this Clause 9.9 considers itself

Deleted: and

March.DOC

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

entitled to vote shall be entitled to Weighted Votes calculated in accordance with Clauses 6.8 to 6.11 (inclusive). Following the Change Voting Date, the Change Control Administrator shall determine, in accordance with Clause 7.22, *mutatis mutandis*, from the Weighted Votes cast by the relevant Parties whether a Change Proposal shall be approved or rejected and collate comments made by Parties. The Change Control Administrator shall notify SPAA EC as to the outcome of the vote on each Change Proposal.

- 9.10 Where a Change Proposal has been approved pursuant to Clause 9.9, the Change Control Administrator shall notify the Authority that it has been so approved and where Clause 9.1 applies shall request the Authority to consent to the Change Proposal. Pending the grant of the Authority's consent, SPAA EC shall take no steps to update the SPAA Products or amend this Agreement pursuant to Clause 9.12.
- 9.11 The Change Control Administrator shall prepare a report after the vote on a Change Proposal is determined pursuant to Clause 9.9. The report, will set out details of the Weighted Votes cast, any comments submitted by Parties pursuant to Clause 9.9, and, where Clause 9.1 applies, state that implementation of the Change Proposal is subject to Authority approval and shall be sent to the Parties, SPAA EC, the Authority and energywatch ("Change Report").
- 9.12 Where a Change Proposal has been approved pursuant to Clause 9.9 and (if Clause 9.1 applies) Clause 9.10 and/or where a Change Proposal has been rejected pursuant to Clause 9.9 but the SPAA Forum or the Authority has allowed an appeal pursuant to Clauses 9.15 to 9.19 (inclusive) the SPAA EC shall decide the appropriate process for implementing the Change Proposal, which shall be in accordance with any procedures for implementation set out in the Change Proposal, or if such Change Proposal is absent recommendations on implementation procedures, as agreed by SPAA EC from time to time, which in any event shall include parameters for:
 - 9.12.1 timing of the process and the timing of the implementation; and
 - 9.12.2 SPAA EC resolutions that are to be passed to update the SPAA Products and/or amend this Agreement in accordance with the Change Proposal.

Once the final form of the Change Proposal has been agreed or determined, this Agreement shall be amended and the SPAA Products shall be updated (as the case may be) in accordance with a resolution of SPAA EC.

- 9.13 Nothing in this Clause 9 shall entitle any Party, or its Party Change Administrator, to make changes to a Change Proposal other than the Party, or Party Change Administrator, that proposed it.
- 9.14 SPAA EC shall agree and issue appropriate procedures in relation to Change Proposals submitted pursuant to this Clause 9 (which shall be subordinate to and shall not be inconsistent with the procedures set out in this Clause 9) and the Parties agree to comply with those procedures as issued from time to time.

Appeals

- 9.15 If a Change Proposal is rejected in accordance with Clause 9.9, any Party that is legitimately interested or impacted by the Change Proposal in question may appeal such decision, within 10 Working Days of receipt of the Change Report, to the SPAA Forum for its determination. None of the foregoing shall entitle any other Party to appeal a vote to reject a Change Proposal to the SPAA Forum.
- 9.16 If a Change Proposal is approved in accordance with Clause 9.9, any Party that is legitimately interested or impacted by the Change Proposal in question may appeal such decision, within 10 Working Days of receipt of the Change Report, to the SPAA Forum for its determination.

Deleted: and

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

- 9.17 Within 10 Working Days of receipt of the Change Report, any Party may appeal to the SPAA Forum against a Change Proposal which has been approved or rejected, if it considers that a Party has voted on a Change Proposal in which it had no legitimate interest or by which it would not be impacted.
- 9.18 If a Party appeals to the SPAA Forum pursuant to Clauses 9.15 to 9.17 (inclusive), it shall provide the SPAA EC with notice of such appeal and provide the SPAA EC with copies of its appeal and any other documentation submitted to the SPAA Forum in support of such appeal. Pending determination of any appeal against the approval of a Change Proposal to the SPAA Forum and the Authority if applicable, SPAA EC shall take no steps to update SPAA Products and/or amend this Agreement pursuant to Clause 9.14.

Emergencies

9.19 Where any change is proposed to this Agreement pursuant to the terms of this Clause 9 which SPAA EC decides is of an urgent nature and is a change which should be decided upon promptly, SPAA EC may decide to reduce the timescales set out in this Clause 9 accordingly.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

PART IV: EVENTS OF DEFAULT, CONSEQUENCES OF DEFAULT AND LIMITATION OF LIABILITY

10. EVENTS OF DEFAULT CONSEQUENCES OF DEFAULT AND LIMITATION OF LIABILITY

- 10.1 It shall be an **"Event of Default"** if:
 - 10.1.1 a Party (the "Defaulting Party") is in breach of any of the material terms or conditions of this Agreement and/or in persistent breach of any of the terms or conditions of this Agreement and, if the breach is or was capable of remedy, it fails to remedy the breach within 20 Working Days of receipt of a notice from SPAA EC giving full details of the breach, requiring the Party to remedy the breach and stating that a failure to remedy the breach may give rise to consequences set out in this Clause 10;
 - 10.1.2 a Party passes a resolution for its winding-up, or a court of competent jurisdiction makes an order for the winding-up or dissolution of the Party;
 - 10.1.3 a Party has an administration order is made in respect of it or a receiver is appointed over, or an encumbrancer takes possession of or sells, any substantial part or parts of the Party's assets, rights, or revenues;
 - 10.1.4 a Party makes an arrangement or composition with its creditors generally or makes an application to a court for protection from its creditors generally;
 - 10.1.5 a Party is unable to pay its debts within the meaning of Section 123 of the Insolvency Act 1986, but as if in that Section the sum of £10,000 was substituted for the sum of £750;
 - 10.1.6 without prejudice to Clause 12, a circumstance of Force Majeure which affects the performance by a Party of substantially all of its obligations under this Agreement continues for more than 180 days;
 - 10.1.7 a Supplier has its Gas Suppliers, Licence revoked or, the Authority provides for Section C or Section D of a Supplier's Gas Suppliers, Licence to cease to have effect, or a Supplier, having acceded to this Agreement pending the grant of a Gas Suppliers Licence, notifies SPAA EC that its application for a Gas Suppliers Licence has been refused;

10.1.8 a Transporter has its Gas Transporter Licence revoked, or a Transporter, having acceded to this Agreement pending the grant of a Gas Transporters Licence, notified SPAA EC that its application for a Gas Transporters Licence has been refused.

Any Party may report any suspected Event of Default to SPAA EC and upon receiving such report, SPAA EC shall notify the Party to whom the suspected Event of Default report relates.

In the event of a Party being in default pursuant to Clause 10.1.1 in respect of obligations owed by a category of Parties that Party shall not be entitled to receive the benefit of any Mandatory Schedule relating to such category of Parties and SPAA EC may decide that those provisions will be suspended as regards such Party with immediate effect or from a specified future date.

10.4 Where SPAA EC is notified by another Party pursuant to Clause 10.2 or otherwise discovers any of the circumstances referred to in this Clause 10 and such breach if capable of remedy is not remedied within any timeframe indicated in the aforementioned, SPAA EC may decide that the relevant Defaulting Party should not exercise any voting rights pursuant to Clauses 6

Deleted: y

Deleted: y

Deleted: y

Deleted: y

Deleted: y

Deleted: y

Deleted:

Deleted: SPAA_180204.DOC
Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

Mandatory Schedule relating to the category of Parties, in default from a date to be determined Deleted: y by SPAA EC until SPAA EC determines otherwise. 10.5 A SPAA EC Member shall be disqualified from acting, and shall not act in his capacity as a SPAA EC Member in relation to a resolution pursuant to this Clause 10 and his alternate shall act in his place (unless the provisions of Clauses 10.5.1 and 10.5.2 apply to him as well) in relation to a resolution where: 10.5.1 his employer is the Defaulting Party or an Affiliate of the Defaulting Party; or 10.5.2 his employer is the Party reporting the suspected Event of Default, or an Affiliate of the Party reporting the suspected Event of Default. 10.6 If both a SPAA EC Member and his alternate are disqualified from acting in relation to a particular resolution as a result of Clause 10.5, that SPAA EC Member shall appoint a further alternate who is employed by a Party or an Affiliate from the category of Parties, that Deleted: y appointed him, and who is not disqualified pursuant to Clause 10, to act as his alternate in relation to that particular resolution. 10.7 Where SPAA EC makes a decision pursuant to Clause 10.4 it shall: 10.7.1 notify the relevant Party of its decision; 10.7.2 notify the Authority of its decision; 10.7.3 notify the other Parties that the relevant Party is in a category of Parties, and is a Deleted: y Defaulting Party and that such Parties may decide not to give such Party the benefit of any Mandatory Schedule in respect of such category of Parties, from a date Deleted: v determined by SPAA EC until SPAA EC determines otherwise. 10.8 On receipt of notification from SPAA EC pursuant to Clause 10.7.3, a Party shall be entitled to decline to give a Defaulting Party the benefit of any part of a Mandatory Schedule relating to a category of Parties, in accordance with such notification, until it is notified by SPAA EC Deleted: y that SPAA EC has determined otherwise. 10.9 Any Party that is a Defaulting Party pursuant to the terms of Clause 10.4 may apply to SPAA EC to have the restrictions referred to in Clause 10 removed. SPAA EC shall consider such application and may levy a fee on the relevant Defaulting Party to remove the relevant restrictions. 10.10 A Supplier shall cease to be a Party to this Agreement upon giving SPAA EC 30 Working Deleted: Supplier Days' notice of its intention to cease to be a Party to this Agreement, where: 10.10.1 it is no longer Registered for any MPRNs on any Gas Transportation Database; or Deleted: v 10.10.2 it does not hold a Gas Suppliers, Licence; and 10.10.3 it has paid all charges and/or met all funding requirements for which it is or will become liable under the terms of this Agreement. A Transporter shall cease to be a Party to this Agreement upon giving SPAA EC 30 Working Days' Notice of its intention to cease to be a Party to this Agreement, where: Deleted: SPAA_180204.DOC 10.11.1 it no longer has Registered MPRNs on its Gas Transportation Database; or Inserted: SPAA 180204.DOC 10.11.2 it does not hold a Gas Transporters Licence; and Deleted: SPAA (suppliers amended draft) 18 March.DOC

SPAA 040304.DOC

to 9 (inclusive) and/or exercise any rights under or benefit from any provisions in a

10.11.3 it has paid all charges and/or met all funding requirements for which it is or will become liable under the terms of this Agreement.

A Party shall cease to be a party to this Agreement with immediate effect if its Gas Suppliers. Licence or Gas Transporters Licence (as appropriate) is revoked or is not granted. A Party shall cease to be a party to this Agreement as a Domestic Supplier on and from the date that Section C of such Supplier's Gas Suppliers, Licence ceases to have effect if the Authority provides that Section C of such Supplier's Gas Suppliers Licence shall cease to have effect. A Party shall cease to be a Party to this Agreement as an Industrial and Commercial Supplier on and from the date that Section D of such Supplier's Gas Suppliers, Licence ceases to have effect if the Authority provides that Section D of such Supplier's Gas Suppliers, Licence shall cease to have effect.

Deleted: such
Deleted: Supplier's
Deleted: y
Deleted: y
Deleted: y

10.13 A Party ceasing to be a party to this Agreement shall be without prejudice to the accrued rights and liabilities of that Party prior to the date of it ceasing to be a Party and shall not affect any continuing obligations of that Party under this Agreement.

_ - **Deleted:** 26

Deleted: v

Deleted: y

Where a Party ceases to be a party to this Agreement, Clauses 10, 11, 12, 13, 15 to 16 (inclusive), 20.1, 21 and 22 shall remain in full force and effect as regards it.

Subject to Clause 10.18 and 10.20 and save as provided in this Clause 10.16 and Clause 10.17 no Party (the "Party Liable") nor any of its officers, employees or agents shall be liable to any other Party for loss arising from any breach of this Agreement other than for loss directly resulting from such breach and which at the date hereof was reasonably foreseeable as not unlikely to occur in the ordinary course of events from such breach in respect of:

Deleted: 1110.16

Deleted: 10.18

Deleted: 10.18

10.15.1 physical damage to the property of that other Party, its officers, employees or agents; and/or

10.15.2 the liability of such other Party to any other person for loss in respect of physical damage to the property of any person,

provided that the liability of any Party in respect of claims for such loss arising from any incident or series of related incidents shall in no circumstances exceed £1 million.

10.16 Nothing in this Agreement shall exclude or limit the liability of the Party Liable for death or personal injury resulting from the negligence of the Party Liable or any of its officers, employees or agents and the Party Liable shall indemnify and keep indemnified any other Party, its officers, employees or agents, from and against all such liability which such other Party may suffer or incur by reason of any claim on account of death or personal injury resulting from the negligence of the Party Liable or any of its officers, employees or agents.

Deleted: 10.18

.17 Subject to Clause 10.20, no Party, nor any of its officers, employees or agents shall in any circumstances whatsoever be liable to any other Party for:

10.17.1 any loss of profit, loss of revenue, loss of use, loss of contract or loss of goodwill; or

10.17.2 any indirect or consequential loss; or

10.17.3 loss resulting from the liability of such other party to any other person howsoever and wheresoever arising save as provided in Clauses 10.15 and 10.16

10.18 The rights and remedies provided by this Agreement to the parties are exclusive and not cumulative and exclude and are in place of all substantive (but not procedural) rights or remedies express or implied and provided by common law or statute in respect of the subject matter of this Agreement, including any rights any Party may possess in tort for delict which shall include actions brought in negligence and/or nuisance. Accordingly, each of the Parties

Deleted: 5

Deleted: 10.14

Deleted: SPAA_180204.DOC
Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

hereby waives to the fullest extent possible all such rights and remedies provided by common law or statute, and releases the Party Liable, its officers, employees and agents to the same extent from all duties, liabilities, responsibilities or obligations provided by common law or statute in respect of the matters dealt with in this Agreement and undertakes not to enforce any of the same except as expressly provided herein.

- 10.19 Save as otherwise expressly provided in this Agreement, this Clause 10 insofar as it excludes or limits liability shall override any other provision in this Agreement provided that nothing in this Clause 10 shall exclude or restrict or otherwise prejudice or affect any of:
 - 10.19.1 the rights, powers, duties and obligations of any Party which are conferred or created by the Act, any licence granted pursuant to the Act or any subordinate legislation made under the Act; or
 - 10.19.2 the rights, powers, duties and obligations of the Authority or the Secretary of State under the Act, any such licence or otherwise howsoever.
- 10.20 Each of the sub-clauses of this Clause 10 shall:
 - 10.20.1 be construed as a separate and severable contract term, and if one or more of such sub-clauses is held to be invalid, unlawful or otherwise unenforceable the other or others of such Clauses shall remain in full force and effect and shall continue to bind the parties; and
 - 10.20.2 survive termination of this Agreement.
- 10.21 Each Party hereby acknowledges and agrees that each of the other Parties holds the benefit of Clauses 10.15, 10.16 and 10.17 for itself and as trustee and agent for its officers, employees and agents.
- 10.22 Each Party hereby acknowledges and agrees that the provisions of this Clause 10 have been the subject of discussion and negotiation and are fair and reasonable having regard to the circumstances as at the date hereof.
- 10.23 For the avoidance of doubt, nothing in this Clause 10 shall prevent or restrict any Party enforcing any obligation (including suing for a debt) owed to it under or pursuant to this Agreement.

Deleted: 10.15

Deleted: 10.15

Inserted: 10.15 **Deleted:** 10.14

Deleted: 10.16

Inserted: 10.16

Deleted: 16

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

PART V: CONFIDENTIALITY

11. CONFIDENTIALITY

General

11.1 Each Party hereby undertakes with each other Party that it shall preserve the confidentiality of and shall not directly or indirectly Disclose or use for its own purposes Confidential Information. The exceptions to this obligation are set out in Clause 11.2.

11.2 Exceptions to Confidentiality Obligation

A Party shall be entitled to Disclose or use Confidential Information if and to the extent that one or more of the following apply:

- 11.2.1 the Party is required or permitted to Disclose Confidential Information pursuant to the terms of a Designated Agreement, to the extent of such requirement or permission; or
- 11.2.2 the Party believes, on reasonable grounds, that market arrangements set out or contemplated by this Agreement require or permit it to Disclose Confidential Information to another person or to use Confidential Information to the extent of such requirement or permission and it has given the other Parties prior written notice of such intended Disclosure; or
- 11.2.3 the person to whose affairs the Confidential Information relates gives its prior written consent to the Disclosure or use, to the extent of such consent; or
- 11.2.4 the Confidential Information, before it is furnished to the relevant Party is in the public domain; or
- 11.2.5 the Confidential Information, after it is furnished to the party:
 - (A) is acquired by the Party in circumstances in which this Clause does not apply;
 - (B) is acquired by a Party in circumstances in which this Clause does apply and thereafter ceases to be subject to the restrictions imposed by this Clause; or
 - (C) enters the public domain,

and in any such case otherwise than as a result of (i) a breach by the Party of its obligations in this Clause or (ii) a breach by the person who disclosed that Confidential Information of that person's confidentiality obligation and the Party is aware of such breach; or

- 11.2.6 the Party is required or permitted to Disclose Confidential Information to any person:
 - (A) in compliance with any provisions of any Relevant Instrument; or
 - (B) in compliance with any other requirement of law or of a Competent Authority; or
 - (C) in response to a requirement of any stock exchange or regulatory authority or the Panel on Take-Overs and Mergers; or
 - (D) pursuant to any judicial or arbitral process or tribunal having jurisdiction relation to the Party; or

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

- 11.2.7 the Party, Discloses Confidential Information to its Affiliates or Related Undertakings, its or its Affiliates' or Related Undertakings' employees, directors, agents, consultants and professional advisers; or
- 11.2.8 the Party Discloses Confidential Information to the Authority.
- 11.3 Confidential Information which a Party is permitted or obliged to Disclose or use pursuant to Clause 11.2 shall not cease to be regarded as Confidential Information in all other circumstances by virtue of such Disclosure or use.

Internal Procedures

- 11.4 With effect from the date of this Agreement each Party shall adopt procedures within its organisation for ensuring the confidentiality of all Confidential Information which it is obliged to preserve as confidential under Clause 11.1. These procedures are:
 - 11.4.1 the Confidential Information will be disseminated within the Party only on a "need to know" basis;
 - 11.4.2 employees, directors, agents, consultants and professional advisers of the Party in receipt of Confidential Information will be made fully aware of the party's obligations of confidence in relation thereto; and
 - 11.4.3 any copies of the Confidential Information, whether in hard copy or computerised form, will clearly identify the Confidential Information as confidential.
- 11.5 Each Party shall take all reasonable steps to ensure that any person referred to in Clause 11.2.6 to whom the Party Discloses Confidential Information does not use that Confidential Information for any purpose other than that for which it is provided and does not Disclose that Confidential Information otherwise than in accordance with this Clause 11.

Affiliate or Related Undertaking

11.6 Each Party shall procure that each of its Affiliates and Related Undertakings observes the restrictions in Clauses 11.1 to 11.5 as if in each Clause there was substituted for the name of the Party the name of the Affiliate or Related Undertaking.

Data Protection Act

- 11.7 Each Party warrants that it has effected, and undertakes that it will during the term of this Agreement effect and maintain all such notifications as it is required to effect and maintain under the Data Protection Act to enable it lawfully to perform the obligations imposed on it by this Agreement. Each Party undertakes to comply with the Data Protection Act in the performance of this Agreement.
- 11.8 Each Party undertakes that, in any case where information to be disclosed by it under this Agreement may lawfully be disclosed only with the prior consent of the person to whom the information relates, it will use its reasonable endeavours to obtain such prior consents so as to enable it, to promptly to perform its obligations under this Agreement.
- 11.9 The Parties acknowledge that, for each of SPAA EC, and each of its sub-committees, to properly to carry out its duties and responsibilities under this Agreement, it may decide or be obliged to keep confidential to it (and may instruct its sub-committees to keep confidential) matters, reports, data and other information produced by or for, or made available to or held by it and in any such case, it shall neither disclose the same to the Parties nor be required by such Parties so to disclose. Each of the Parties agrees to respect the position of SPAA EC, its sub-committees, and the SPAA EC accordingly.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

SPAR 040304, DDC 42

11.10 Each of the Parties agrees, subject to any relevant confidentiality provision binding on it, to provide SPAA EC, the Secretary and the Secretariat with all data and other information reasonably requested by and necessary for SPAA EC, the Secretary and/or the Secretariat properly to carry out their duties and responsibilities under this Agreement.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

PART VI: FORCE MAJEURE

12. FORCE MAJEURE

- 12.1 If any Party (the "Affected Party") shall be unable to carry out any of its obligations under this Agreement (for the avoidance of doubt including any obligation under a Mandatory Schedule or, to the extent an election has been made pursuant to Clause 5, an Elective Schedule) due to a circumstance of Force Majeure this Agreement shall remain in effect but:
 - 12.1.1 the Affected Party's obligations;
 - 12.1.2 the obligations of each of the other Parties owed to the Affected Party under this Agreement; and
 - 12.1.3 any other obligations of such other Parties under this Agreement owed between themselves which the relevant Party is unable to carry out directly as a result of the suspension of the Affected Party's obligations,

shall be suspended without liability for the period during which the circumstance of Force Majeure prevails provided that:

- (i) the Affected Party gives the other Parties prompt notice describing the circumstance of Force Majeure including the nature of the occurrence and its expected duration and where reasonably practicable continues to furnish regular reports with respect thereto during the period of Force Majeure; and
- (ii) the suspension of performance is of no greater scope and of no longer duration than is required by the circumstance of Force Majeure; and
- (iii) no obligations of any Party that arose before the circumstance of Force Majeure causing the suspension of performance are excused as a result of the Force Majeure; and
- (iv) the Affected Party uses all reasonable efforts to mitigate the impact of the circumstance of Force Majeure and to remedy its inability to perform as quickly as possible; and
- (v) immediately after the end of the circumstance of Force Majeure the Affected Party notifies the other Parties in writing of the same and resumes performance of its obligations under this Agreement.

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

PART VII: DISPUTES

13. DISPUTES

13.1 Save where expressly stated in this Agreement to the contrary, and subject to any contrary provision of the Act, any Gas Suppliers Licence or Gas Transporters Licence issued pursuant to the Act or the rights, powers, duties and obligations of the Authority or Secretary of State under the Act, any such Gas Suppliers Licence or Gas Transporters Licence or otherwise howsoever arising, any dispute or difference of whatever nature and howsoever arising under, out of or in connection with this Agreement (a "Dispute") shall be resolved according to the provisions of this Clause 13.

Deleted: y

Contract Management

Any Party shall refer a Dispute to the Contract Managers, by notice in writing to all other Parties to the Agreement who are Party to the Dispute (the Party referring the Dispute and the other Parties to the Dispute each being a "Disputing Party"). The Contract Managers of the Disputing Parties shall endeavour to resolve the Dispute between them. The Contract Managers of the Disputing Parties shall have authority to negotiate in relation to and to resolve the Dispute including authority to bind the Party nominating them. The joint and unanimous decision of the Contract Managers of the Disputing Parties shall be binding upon the Parties to the Dispute.

Arbitration

- 13.3 If the Contract Managers of the Disputing Parties are not able to resolve the Dispute within 10 Working Days of the reference of a Dispute to them, then any Disputing Party may, within 20 Working Days of such reference, refer the Dispute to arbitration before an arbitral tribunal composed of a single arbitrator pursuant to the rules of the London Court of International Arbitration.
- 13.4 Whatever the nationality residence or domicile of any Disputing Party and wherever the Dispute or any part thereof arose the laws of England and Wales shall be the proper law of any reference to arbitration hereunder and in particular (but not so as to derogate from the generality of the foregoing) the seat of any such arbitration shall be London and the provisions of the Arbitration Act 1996 shall apply to any such arbitration wherever the same or any part of it shall be conducted.
- 13.5 Notwithstanding the provision of the rest of this Clause 13, any Party may apply at any time to any court of competent jurisdiction for any emergency interim interlocutory relief as may be necessary.

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

SPAR 040304, DDC 45

PART VIII: MISCELLANEOUS

14. DEROGATIONS

- 14.1 SPAA EC may resolve to grant a derogation to any Party or Parties in relation to any obligation contained in any Mandatory Schedule to or of this Agreement, which may be subject to conditions and shall specify the term, scope and application of such derogation, and may amend or retract any such derogation, or any such conditions relating thereto, from time to time as it sees fit.
- 14.2 A Party may, by notice in writing to the Secretary, apply to SPAA EC for a derogation pursuant to Clause 14.1 ("Application for Derogation").
- 14.3 Where the Secretary receives an Application for Derogation from a Party pursuant to Clause 14.2, it shall ensure that the Application for Derogation is added to the agenda for the next SPAA EC meeting, and shall give notice to all Parties and the Authority, at least 10 Working Days prior to the SPAA EC meeting at which the application is to be considered, stating:
 - 14.3.1 that the Application for Derogation has been made, setting out the terms of the derogation sought, and the identity of the Party making the Application for Derogation; and
 - 14.3.2 the time (not being less than 10 Working Days from the date on which notice is provided) within which Parties, energywatch and the Authority may make representations or objections with respect to the derogation which has been applied for.
- 14.4 If pursuant to Clause 14.3.2 any Parties make representations or objections with respect to the derogation that has been applied for to the Secretary, the Secretary shall as soon as reasonably practicable and in any event before the SPAA EC meeting at which the Application for Derogation is to be considered, publish such representations and objections on the SPAA Forum Website www.spaa.co.uk unless such representations or objections have been marked as confidential.
- 14.5 Where a Party is granted a derogation by SPAA EC in accordance with this Clause 14, that Party shall be excused from complying with the obligations specified in the terms of that derogation, and shall be deemed not to be in breach of this Agreement for failing to comply with the relevant obligations for the term of the derogation, but shall be required to comply with any modified obligations which are specified as a condition of the derogation.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

15. CONTRACT MANAGEMENT

- 15.1 Each Party shall appoint an appropriate person (each a "Contract Manager" and together the "Contract Managers") to manage all matters arising under or in connection with this Agreement and to monitor the general operation of this Agreement.
- 15.2 Each Contract Manager appointed by a Party shall ensure that procedures are in place in respect of that Party to ensure that there is adequate support for operations provided under this Agreement and timely resolution of problems that may occur including a point of contact to process and resolve such problems.
- 15.3 At times determined by SPAA EC a meeting of Contract Managers shall be convened to discuss operational issues, with particular regard to inter-operability issues and performance of third parties.
- 15.4 Each Party shall notify the other Parties, SPAA EC, the Secretariat and the Change Control Administrator of the name and contact details of the Contract Manager appointed by it for the purposes of this Agreement from time to time.

16. ENTIRE AGREEMENT

- 16.1 This Agreement and any document referred to herein represents the entire understanding, and constitutes the whole agreement, in relation to its subject matter and supersedes any previous agreement between the Parties with respect thereto and without prejudice to the generality of the foregoing excludes any warranty, condition or other undertaking implied at law or by custom.
- 16.2 Each Party confirms that, except as provided in this Agreement and without prejudice to any liability for fraudulent misrepresentation, no Party has relied on any representation, warranty or undertaking which is not contained in this Agreement or any document referred to herein.

17. SEVERABILITY

17.1 If any provision of this Agreement shall be held to be invalid or unenforceable by a judgement or decision of any court of competent jurisdiction or any authority (including the Authority) whose decisions shall be binding on the Parties, the same shall be deemed to be severable and the remainder of this Agreement shall remain valid and enforceable to the fullest extent permitted by law. In any such case, the Parties will negotiate in good faith with a view to agreeing one or more provisions which may be substituted for such invalid or unenforceable provision in order to give effect, so far as practicable, to the spirit of this Agreement.

18. WAIVERS

18.1 The failure by any Party to exercise, or the delay by any Party in exercising, any right, power, privilege or remedy provided by this Agreement or by law shall not constitute a waiver thereof nor of any other right, power, privilege or remedy. No single or partial exercise of any such right, power, privilege or remedy shall preclude any further exercise thereof or the exercise of any other right, power, privilege or remedy.

19. ASSIGNMENT AND SUB-CONTRACTING

19.1 Subject to Clauses 19.2 and 19.3, and except as provided elsewhere in this Agreement, no Party shall assign any of its rights under this Agreement without the prior written consent of all other Parties such consent not to be unreasonably withheld or delayed.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

- 19.2 Subject to Clauses 10.13 and 10.14 any Party may transfer its rights and obligations under this Agreement to any successor holder of its Gas Suppliers Licence (or part thereof) or Gas Transporters Licence (or part thereof) (as appropriate) provided that:
 - 19.2.1 the Authority has approved the transfer of such Gas Suppliers, Licence or Gas Transporters Licence (as appropriate) pursuant to the Act; and
 - 19.2.2 the successor holder executes an Accession Agreement with SPAA EC.
- 19.3 If Clause 19.2.2 applies, each Party hereby authorises and instructs any delegate appointed by SPAA EC to sign any Accession Agreement on its behalf and undertakes not to withdraw, qualify or revoke any such authority or instruction at any time. Upon execution of the Accession Agreement, the successor holder shall become a Party for all purposes of this Agreement from the date specified in the Accession Agreement.
- 19.4 Any Party may sub-contract or delegate the performance of all or any of its obligations under this Agreement to any appropriately qualified and experienced third party, but shall at all times remain liable to any other Parties in relation to all sub-contracted or delegated obligations.
- 19.5 Each Party shall notify other Parties, on request, of any subcontractors appointed by it for the purposes of this Agreement.

20. NOTICES

- Any notice, request, claim form or other communication to be made by a Party to another Party under or in connection with this Agreement shall be in writing and shall be delivered by hand or by fax or by first class post, or (provided this Agreement expressly so provides) sent by e-mail, to that Party's address as included in Parts 2, 3 and 4 of Schedule 1 as may be varied from time to time by notice from a Party to all other Parties (marked for the attention of the Contract Managers) and notified to the SPAA EC in accordance with the terms of this Clause.
- 20.2 Unless otherwise stated in this Agreement, a notice, request or other communication sent in accordance with Clause 19.1 shall be deemed received:
 - 20.2.1 if delivered by hand, when left at the address referred to above;
 - 20.2.2 if sent by post, 2 Working Days after the date of posting;
 - 20.2.3 if sent by fax, upon production by the sender's equipment of a transmission report indicating that the fax was sent to the fax number of the recipient in full without error;
 - 20.2.4 if sent by e-mail upon receipt by the sender of a delivery receipt e-mail message which states that the e-mail has been successfully delivered to the recipient.
- Without any prejudice to Clause 20.2, notice, request or communication sent by e-mail shall also be sent by hand by post or by fax.
- 20.4 This Clause 20 shall apply mutatis mutandis to SPAA EC, the Secretary, the Secretariat and the Change Control Administrator save that their address and details shall be as set out below or varied from time to time by notice from such persons to the other Parties in accordance with the terms of this Clause 20.

SPAA EC

Address:

Deleted: 10.13

Inserted: 10.13

Deleted: 10.12

Deleted: 10.14

Inserted: 10.14

Deleted: 10.13

Deleted: y

Deleted: y

Deleted: and

Deleted: SPAA_180204.DOC
Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

| | | Fax: | |
|---|------|--|--|
| | | e-mail: | |
| | | | |
| | | SECRETARY | |
| | | Address: | |
| | | Fax: | |
| | | e-mail: | |
| | | | |
| | | SECRETARIAT | |
| | | Address: | |
| | | Fax: | |
| | | e-mail: | |
| | | | |
| | | CHANGE CONTROL ADMINISTRATOR | |
| | | Address | |
| | | Fax | |
| | | e-mail | |
| | | | |
| | 21. | AUDIT AND RECORDS | |
| | 21.1 | On request by SPAA EC each <u>Party shall ensure that any auditor appointed by SPAA EC has</u> access at reasonable times and on reasonable notice to: | Deleted: Supplier |
| ı | | | (|
| | | 21.1.1 any records maintained by that <u>Party in relation to any MPRN for which it is or has</u> been Registered, or has Registered on its Gas Transportation Database (as | Deleted: Supplier |
| | | appropriate) in the [28] months prior to that date; | |
| | | 21.1.2 any records maintained by it in relation to any MPRN comprised in its Gas Suppliers, Licence or Gas Transporters Licence in the [28] months prior to that date; | Deleted: y |
| | | 21.1.3 any software, hardware, data or information held by the <u>Party</u> or its agents where | Deleted: Supplier |
| , | | reasonably required by the auditor; and | |
| | | 21.1.4 the Party's premises. | Deleted: Supplier's |
| | 21.2 | Each [Party] shall provide the [Secretary] [Secretariat] with a report detailing the number of | Deleted: Supplier Deleted: in respect of which a |
| | | MPRNs in respect of which, in the case of Suppliers, that Supplier was Registered, or in the case of Transporters, that Transporter has Registered on its Gas Transportation Database, on | Supplier was Registered Deleted: SPAA 180204.DOC |
| ı | | the [15 th]day of each calendar month, taking into account all (if any) resolutions of objections which may change the number of such MPRNs on that [15 th]day (the " Report "). The report | Inserted: SPAA_180204.DOC |
| l | | shall be provided by the last day of the following calendar month. | Deleted: SPAA (suppliers amended draft) 18 March.DOC |
| | | $i_{i}^{\prime\prime}$ | |

[Note: Still need to ascertain day of the month to which the Report relates.]

21.3 The [Secretary] confirms that in accordance with Clause 6.57, it shall not disclose details of any Report delivered to it pursuant to Clause 21.2 to the SPAA EC, SPAA Forum, the Secretariat or any other Party.

Deleted: 0

22. COUNTERPARTS

22.1 This Agreement may be executed in any number of counterparts each of which when executed and delivered shall be an original, but all the counterparts together shall constitute the same document.

23. CONTRACTS (RIGHTS OF THIRD PARTIES) ACT 1999

No term of this Agreement is enforceable under the Contracts (Rights of Third Parties) Act 1999 by a person who is not a party to this Agreement.

24. GOVERNING LAW

- 24.1 This Agreement is governed by, and shall be construed in accordance with English law and the Parties submit to the exclusive jurisdiction of the courts of England.
- 24.2 Each Party agrees that without preventing any other mode of service, any document in an action including any writ of summons or other originating process or any third party or other Party notice) may be served on any Party by being delivered to or left for that Party at its address for service of notices and each Party undertakes to maintain such an address at all times in the United Kingdom and to notify the other Party in advance of any change from time to time of the details of such address in the manner prescribed in Clause 20.

Deleted: 19

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

SCHEDULE 1

| rart I |
|-------------------------------------|
| PARTIES |
| <u>Part 2</u> |
| DOMESTIC SUPPLIERS |
| Address: |
| Fax: |
| e-mail: |
| Part 3 |
| INDUSTRIAL AND COMMERCIAL SUPPLIERS |
| Address: |
| Fax: |
| e-mail: |
| <u>Part 4</u> |
| TRANSPORTERS |
| Address: |
| <u>Fax:</u> |
| e-mail: |

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

SCHEDULE 2

Accession Agreement

] between:

| | | | | - | | - | | | |
|-----|---|-----|----------|-------------|---------|---------------------|-------------|----------|----|
| (1) | [| |], a co | mpany incor | porated | under the laws of [|] [(r | number [| |
| | |])] | and | having | its | [Registered] | [principal] | office | at |
| | [| |] (the ' | 'New Party' | '); and | | | | |

(2) [SPAA EC] (the "Nominee") on behalf of all the parties to the Supply Point Administration Agreement referred to below.

WHEREAS:

- (A) The Parties named therein (1), and SPAA Limited (2) have entered into an agreement, (the "Supply Point Administration Agreement") on [].
- (B) The New Party has requested that it be admitted as a Party in the capacity of [Industrial and Commercial Supplier] and/or[Domestic Supplier] or [Transporter] pursuant to Clause 4 of the **Supply Point Administration Agreement** and each of the parties hereby agrees to such admission.

NOW IT IS HEREBY AGREED as follows:

THIS AGREEMENT is made on [

- Unless the context otherwise requires, words and expressions defined in the Supply Point Administration Agreement shall bear the same meanings respectively when used herein.
- 2. The Nominee (acting on behalf of each of the Parties) hereby admits the New Party as an additional Party under the **Supply Point Administration Agreement** on the terms and conditions hereof and with effect from [insert effective date of admission].
- 3. The New Party hereby accepts its admission as a Party and undertakes with the Nominee (acting on behalf of each of the Parties) to perform and to be bound by the terms and conditions of the **Supply Point Administration Agreement** as a Party as from the [insert effective date of admission].
- 4. For all purposes in connection with the **Supply Point Administration Agreement** the New Party shall as from the [insert effective date of admission] be treated as if it had been a signatory of the **Supply Point Administration Agreement** as a [Industrial and Commercial Supplier] and/or [Domestic Supplier] or [Transporter] and as if this Agreement were part of the **Supply Point Administration Agreement**, and the rights and obligations of the Parties shall be construed accordingly.
- 5. The New Party's name and contact details shall be inserted into [Part 2] and/or [Part 3] or [Part 4] of Schedule 1 as a [Domestic Supplier] and/or [Industrial and Commercial Supplier] or [Gas Transporter].
- 6. This Agreement and the Supply Point Administration Agreement shall be read and construed as one document and references in the Supply Point Administration Agreement to the Supply Point Administration Agreement (howsoever expressed) shall be read and construed as references to Supply Point Administration Agreement and this Agreement.

Deleted: SPAA 180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

| 7. | This Agreement shall be governed by and construed in all respects in accordance with English |
|----|--|
| | law and the provisions of Clause 24 of the Supply Point Administration Agreement shall |
| | apply hereto mutatis mutandis. |

| 8. | No term of this Agreement is enforceable under the Contracts (Rights of Third Parties) Ac | ٢t |
|----|---|----|
| | 999 by a person who is not a party to this Agreement. | |

| , , |
|---|
| AS WITNESS the hands of the duly authorised representatives of the parties hereto the day and year first above written. |
| [New Party] |
| By: |
| Notice details (Clause 19 of the Supply Point Administration Agreement) for insertion into Schedule 1 |
| Address: |
| Facsimile Number: |
| Attention: |
| [Nominee] |
| (for and on behalf of each of the parties to the Supply Point Administration Agreement) |
| By: |
| |
| |
| |
| * Delete/complete as appropriate. |
| |

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

SCHEDULE 3

Self-Certification Form

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

SCHEDULE 4

SPAA Limited

1. BACKGROUND

- 1.1 Establishment of joint venture: The Parties have agreed to establish a joint venture company to carry on the Business.
- 1.2 SPAA Ltd: SPAA Ltd was incorporated in England on 1st February 2002 and at the date of this Agreement has an authorised share capital of £1,000,000 divided into 1,000,000 ordinary shares of £1 each of which two shares have been issued.
- 1.3 Shareholders of SPAA Ltd: It is intended that the shareholders of SPAA Ltd shall be limited to the Parties to the SPAA for the time being and from time to time, each of whom shall hold a single share, and any nominee as referred to in paragraph 3.2.1 of this Schedule.
- 1.4 Regulation or rights: The Shareholders have agreed that their respective rights as shareholders in SPAA Ltd shall be regulated by the provisions of this Schedule (which, for the avoidance of doubt, shall include the provisions of the Annexes hereto) and SPAA Ltd has agreed with the Shareholders to comply with such of the matters contained in this Schedule as relate to SPAA Ltd.

Deleted: ab

2. ADDITIONAL DEFINITIONS AND INTERPRETATION

1.1 Definitions: In this Schedule, except where the context otherwise requires:

"Articles" means the Articles of Association of SPAA Ltd set out in

Annex 3 to this Schedule, as the same may be amended

from time to time;

"Board" means the board of directors of SPAA Ltd;

"Business" means acting as a corporate vehicle in relation to the

SPAA pursuant to (i) a resolution of SPAA EC (or any sub-committee of it) passed pursuant to Clause 5 of this Agreement and effective by virtue of the provisions of that Clause or (ii) a decision of the Secretariat acting within the scope of its authority which (in each case) it is necessary or desirable to implement by means of a binding contract

on an arms-length basis;

"Chairman" means the chairman of the Board for the time being and

from time to time;

"Company Secretary" means the company secretary of SPAA Ltd for the time

being and from time to time;

"Directors" means the directors of SPAA Ltd for the time being and

from time to time;

"Intellectual Property" means patents, registered design rights, unregistered design

rights, copyrights, rights in trade marks whether registered

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

or not, goodwill and rights in confidential information and know-how and any associated or similar rights (including in all cases, applications and rights to apply therefor);

"Shareholders" means the persons for the time being and from time to time

registered as holders of Shares; and

"Shares" means ordinary shares of £1 each in the ordinary share

capital of SPAA Ltd and any shares issued in exchange therefore by way of conversion or reclassification and any shares representing or deriving from such shares as a result of any increase in or reorganisation or variation of the

ordinary share capital of SPAA Ltd;

"SPAA Ltd" means SPAA Limited (registered number 04365599; and

"SPAA Completion means the date falling [] Working Days after the date on

Date" which this Schedule takes effect or such later date as shall

be agreed by SPAA EC.

- 1.2 *Interpretation*: The Parties and SPAA Ltd acknowledge and agree that, notwithstanding any other provision of this Agreement:
 - 1.2.1 SPAA Ltd is a Party to this Agreement solely for the purposes of this Schedule and is bound only to the extent of those obligations on its part which are expressly set out or referred to in this Schedule and not by any other provision of this Agreement;
 - 1.2.2 SPAA Ltd shall have only such rights under or in respect of this Agreement as are expressly set out or referred to in this Schedule;
 - 1.2.3 the consent or agreement of SPAA Ltd shall not be required to any modification, abrogation, amendment or suspension of any provision of this Agreement which is not expressly set out in this Schedule and SPAA Ltd hereby irrevocably waives any rights which it might be considered or held to have to consent or agree to any such modification, abrogation, amendment or suspension;
 - 1.2.4 within this Agreement the rights of the parties as Shareholders are set out exclusively in this Schedule and no other provision of this Agreement shall apply in the regulation of the rights and obligations of Shareholders inter se in their capacity as Shareholders or as between the Shareholders (or any of them) and SPAA Ltd; and
 - 1.2.5 SPAA Ltd shall take no action (and the Shareholders shall not take any step which could cause SPAA Ltd to take any such action) which could prejudice in any way the rights or interests of any Party under this Agreement.

3. ESTABLISHMENT OF SPAA LTD AND NEW PARTIES

- 3.1 *Completion:* On the SPAA Completion Date, each of the Parties, the Shareholders and SPAA Ltd shall perform its obligations set out in, and comply with the provisions of, Annex 1 to this Schedule and procure that the subscribers and the first directors of SPAA Ltd shall comply with the provisions of the same Annex 1.
- 3.2 *New Parties*: Upon the accession of a New Party to the SPAA pursuant to an Accession Agreement the Directors shall either:

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

32.1 transfer to such New Party one Share held by a nominee in accordance with the provisions of paragraphs 10.3 or 10.4 of this Schedule; or

Deleted: 1

3 2.2 allot to such New Party one unissued Share (and the Shareholders agree that where no Shares are otherwise available for issue that they will exercise the voting rights attaching to their Shares to procure that all necessary steps are taken to create and/or authorise the issue of further Shares).

Deleted: 1

4. SPAA LTD'S BUSINESS

- 4.1 Compliance: Each Shareholder agrees with the other Shareholders to exercise its rights under this Schedule and as a Shareholder in SPAA Ltd so as to ensure that:
 - 4.1.1 SPAA Ltd performs and complies with all its obligations under this Schedule and complies with the restrictions (if any) imposed upon it by the Articles; and
 - 4.1.2 the Business is conducted in accordance with sound and good business practice with the intention of breaking even.
- 4.2 Sole business of SPAA Ltd: The Shareholders and SPAA Ltd acknowledge and agree that, unless and until the Shareholders give their explicit written consent, the business of SPAA Ltd shall be confined to the Business.
- 4.3 Independence of operations: Each Shareholder acknowledges and agrees with the other Shareholders and SPAA Ltd that SPAA Ltd will have complete independence in its operations and undertakes not to take any action which obstructs or interferes with, or seeks to obstruct or interfere with, the Business provided that this paragraph 4.3 shall not affect the manner in which any Shareholder may exercise its rights in respect of Shares held by it.

5. THE MANAGEMENT OF SPAA LTD

- 5.1 Directors:
 - 5.1.1 The Shareholders shall procure that the Directors shall be all the SPAA EC Members for the time being and from time to time and each Director shall have as his alternate for the purposes of this Schedule the alternate appointed by him pursuant to Clause 6.18 of this Agreement.
 - 5.1.2 The Parties shall indemnify SPAA Ltd as set out in paragraphs 5.1.2.1 to 5.1.2.5 of this Schedule against all claims, demands, liabilities, losses, costs and expenses which SPAA Ltd may suffer or incur by reason of any claim by any Director in connection with his removal from office as a Director and the liability to indemnify shall be met:
 - 5.1.2.1 in the case of the removal of either of the Directors who are the I&C Members, severally and rateably in accordance with Clause 8.8 by the Suppliers entitled to appoint the I&C Member;
 - 5.1.2.2 in the case of the removal either of the Directors who are the Large Domestic Members, severally and rateably in accordance with Clause 8.8 by the Suppliers entitled to appoint the Large Domestic Members:
 - 5.1.2.3 in the case of the removal of the Director who is the Small Domestic Member, severally and rateably in accordance with Clause 8.8 by the Suppliers entitled to appoint the Small Domestic Member:

Deleted: 3

Deleted: SPAA_180204.DOC
Inserted: SPAA_180204.DOC
Deleted: SPAA (suppliers

amended draft) 18 March.DOC

- 5.1.2.4 in the case of the removal of the Director who is the Large
 Transporter Member, severally and rateably in accordance with
 Clause 8.8 by the Gas Transporters entitled to appoint the Large
 Transporter Member;
- 5.1.2.5 in the case of the removal of the Director who is the Small

 Transporter Member, severally and rateably in accordance with

 Clause 8.8 by the Transporters entitled to appoint the Small

 Transporter Member.

Deleted:

- 5.2 Chairman: The Chairman shall be the SPAA EC Chairman for the time being and from time to time. If the Chairman is unable to be present at a meeting, he may nominate another Director (or any Director's alternate) to act as Chairman. If neither the Chairman nor his nominee is present within half an hour after the time appointed for holding the meeting, the Directors present may appoint any of their number to be Chairman of that meeting.
- 5.3 Committees: The Directors may delegate any of their powers to committees of the Board consisting of such persons as the Directors may resolve. Any such committee shall exercise only powers expressly delegated to it and shall comply with any regulations imposed on it by the Board.
- 5.4 Company Secretary: The Company Secretary shall be the Secretary for the time being and from time to time.
- 5.5 Proceedings at Board Meetings:
 - 5.5.1 <u>Voting Rights</u>: Each Director shall have one vote. The Chairman shall have no vote in his capacity as Chairman.
 - 5.5.2 <u>Frequency</u>: The Board shall meet at intervals of not less than once in any period of three months unless otherwise agreed by the Directors and insofar as reasonably practicable meetings of the Board shall follow on immediately from meetings of SPAA EC. A meeting of the Board may be convened at any reasonable time at the request of any Director by written notice to the Company Secretary.
 - 5.5.3 <u>Meetings</u>: Meetings of the Board may be held by conference telephone call provided that participants acknowledge that they can speak to and hear each other.
 - 5.5.4 Notice: Each of the Directors shall be given notice by the Company Secretary of each meeting of the Board setting out details of the time, date and place of meeting at least 5 Working Days prior to the date of such meeting, provided that such period of notice may be shortened for particular meetings by unanimous written consent of all Directors entitled to attend and vote thereat.
 - 5.5.5 Quorum: The quorum for meetings of the Board shall be constituted by the attendance of one (1) I&C Member, one (1) Large Domestic Member, one (1) Small Domestic Member, one (1) Large Transporter and one (1) Small Transporter (or their alternates): in person or participating by conference telephone call throughout such meeting.
 - 5.5.6 Resolutions: All resolutions of the Board shall be made by unanimous vote of the Directors present or participating by conference telephone call.

Deleted: and

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

- 5.5.7 Written resolutions: A written resolution signed by all Directors shall be as valid and effective as a resolution passed unanimously by a meeting of the Board properly convened and constituted in accordance with the terms of this Schedule and the Articles.
- 5.5.8 Minutes: No later than 5 Working Days after each Board meeting, the Company Secretary shall circulate minutes of that meeting to each of the Directors.
- 5.6 Exercise of Shareholders' Rights: the Shareholders shall exercise the rights attaching to their Shares in the manner best calculated to secure the implementation of decisions taken by the SPAA EC or SPAA Forum (or, on appeal, by the Authority) pursuant to this Agreement, and shall not exercise their rights in a manner which is inconsistent with any such decision.

6. RESERVED MATTERS

- 6.1 The Shareholders shall exercise the rights attaching to their shares to procure, so far as they are able, that SPAA Ltd shall not take any action and no resolution relating to such action shall be passed by SPAA Ltd in respect of the matters set out in Annex 2 to this Schedule, except pursuant to a decision of SPAA EC or the SPAA Forum (as appropriate) or, on appeal, of the Authority, taken in accordance with this Agreement.
- 6.2 For the avoidance of doubt, to the extent that SPAA Ltd takes any action in respect of any matter set out in Annex 2 to this Schedule which is pursuant to a decision of SPAA EC or the SPAA Forum (as appropriate) or, on appeal, of the Authority, taken in accordance with this Agreement, no resolution of the Shareholders shall be required to ratify such action.

7. SPAA LTD EXPENDITURE

- 7.1 *Inclusion of expenditure in budgets*: Anticipated expenditure of SPAA Ltd shall be included in any budget prepared pursuant to Clause 8 of this Agreement, and shall be subject to approval in accordance with that clause.
- 7.2 SPAA Ltd obligations: The Shareholders shall exercise the rights attaching to their shares to procure that SPAA Ltd shall not incur costs unless authorised by a budget approved pursuant to Clause 8 of this Agreement, except insofar as necessary in order to comply with legally binding obligations to which it is subject or insofar as such costs fall within the terms of any previous approved budget.
- 7.3 Authorisation and reimbursement: Expenditure by SPAA Ltd shall be authorised by SPAA EC, submitted to the Secretariat for payment, and reimbursed by the Parties in accordance with the provisions of Clause 8 of this Agreement

8. ACCOUNTS

- 8.1 Annual Accounts: At the end of each of SPAA Ltd's financial years, or as soon as reasonably practicable thereafter, SPAA Ltd shall procure that an account shall be taken of all the assets and liabilities of SPAA Ltd and of all dealings and transactions of SPAA Ltd during such financial year and that the Board shall prepare a report and accounts in accordance with the Companies Act 1985 (as amended) to be audited within three months after the end of each financial year.
- 8.2 Audit: Any Party shall have the right at any time to require SPAA Ltd to instruct the auditors of SPAA Ltd to conduct a review in respect of the financial affairs of SPAA Ltd. The cost of such review shall be borne by the Party requesting such review, unless such review is approved by the Directors in which case it shall be borne by

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

SPAA Ltd. If any such review is requested, SPAA Ltd shall procure that SPAA Ltd's auditors are given all reasonable assistance to complete the review within a reasonable period of time.

9. DISTRIBUTION POLICY

The Shareholders shall take such action as may be necessary to procure that:

- 9.1 Annual General Meeting: SPAA Ltd's annual general meeting at which audited accounts in respect of the preceding financial year are laid before the Shareholders is held not later than the date falling six months after the end of that financial year;
- 9.2 Auditors' Report: SPAA Ltd's auditors shall at the expense of SPAA Ltd be instructed to report as to the amount of the profits available for distribution by SPAA Ltd for each accounting reference period at the same time as they sign their report on SPAA Ltd's audited accounts for the accounting reference period in question;
- 9.3 Distribution of Profits: SPAA Ltd distributes to and among the Shareholders within 30 days of approval of the audited accounts 100 per cent. of its profits available in each year, subject to the appropriation of such reasonable and proper reserves for working capital or otherwise as the Board may consider appropriate; and
- 9.4 Shareholder guarantees: If any indemnity, guarantee or other assurance against loss is given by a Shareholder for any obligation or liability of SPAA Ltd at the request of SPAA Ltd, all the other Shareholders shall indemnify such Shareholder in respect of any liability arising out of such indemnity, guarantee or other assurance against loss severally and rateably in accordance with Clause 8.8 of this Agreement.

10. TRANSFER OF SHARES

- 10.1 *Restriction on transfer*: Otherwise than in accordance with the following provisions of this paragraph 10 no Shareholder shall:
 - 10.1.1 pledge, mortgage (whether by way of fixed or floating charge) or otherwise encumber its legal or beneficial interest in its Shares; or
 - 10.1.2 sell, transfer or otherwise dispose of any of such Shares (or any legal or beneficial interest therein); or
 - 10.1.3 enter into any agreement in respect of the votes attached to Shares; or
 - 10.1.4 agree, whether or not subject to any condition precedent or subsequent, to do any of the foregoing.
- 10.2 Permitted transfers: A Shareholder may transfer its Share to a successor holder of such Shareholder's Gas Suppliers, Licence or Gas Transportation Licence (as appropriate) in circumstances where such person becomes a Party at the same time as such Shareholder ceases to be a Party.
- 10.3 Retiring Shareholders: If any Shareholder ceases to be a Party for any reason (the "Retiring Shareholder"), then upon written notice to the Retiring Shareholder by any other Shareholder, the Retiring Shareholder shall transfer at par to a nominee for all the Shareholders (other than the Retiring Shareholder) selected by the Directors the Share held by the Retiring Shareholder. All costs and expenses of such transfer shall be for the account of the Retiring Shareholder.
- 10.4 *Enforced transfer*: If a Retiring Shareholder shall fail or refuse to transfer any Shares in accordance with its obligations under paragraph 10.3, the Directors may authorise

Deleted: y

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

SPAA Ltd to execute and deliver a transfer from the Retiring Shareholder to a nominee on behalf of the Retiring Shareholder. SPAA Ltd may accept the consideration for the transfer and hold it on trust for the Retiring Shareholder, which acceptance shall be a good discharge to the nominee and may set off such amount against the costs and expenses of the transfer. The Directors shall cause the transferee to be registered as the holder of such Share and following the registration of the transfer the validity of the proceedings shall not be questioned by any person.

10.5 Nominee's holding: The nominee referred to in paragraphs 10.3 and 10.4 shall hold Shares transferred to it until such time as it is directed by the Directors to transfer them (or some of them) to one or more Parties. For the avoidance of doubt, wherever in this Schedule a percentage figure of the number of Shares in issue is referred to, this figure shall be calculated as if all Shares held by the nominee were not in issue.

11. DURATION AND TERMINATION

This Schedule shall continue in full force and effect until the first to occur of the following events:

- the termination of this Agreement pursuant to Clause 3.3;
- 11.2 all the Shareholders agree in writing to terminate the arrangements set out in this Schedule;
- 11.3 an effective resolution is passed or a binding order is made for the winding up of SPAA Ltd,

provided, however, that this Schedule shall cease to have effect as regards any Party who, having been entitled under the terms of this Schedule to hold Shares, ceases to hold any Shares.

12. SHAREHOLDERS GENERALLY

The Shareholders shall procure that:

- save for any nominee referred to in paragraph 10.3, only the Parties shall acquire Shares (whether by transfer or allotment) and that no Party shall be a Shareholder unless and until it has agreed to be bound by this Schedule in the capacity of a Shareholder (which a Party, shall be taken to have done by being a signatory to this Agreement or executing an Accession Agreement); and
- 12.2 the Directors shall neither transfer nor allot any Share or Shares other than as set out in paragraphs 3.2, 10.4 or 10.5 of this Schedule and that, save in the case of a nominee as referred to in paragraphs 10.3 and 10.4, no Party shall at any point hold more than one Share.

13. INTELLECTUAL PROPERTY

If and to the extent that any Shareholder discloses any of its Intellectual Property to SPAA Ltd for use in connection with the Business, unless it is unable to do so it shall grant, and shall be deemed to have granted from the date of such disclosure, licences of such Intellectual Property to SPAA Ltd for use in connection with the Business and for no other purpose whatsoever. Any such licence shall be irrevocable, non-exclusive, perpetual and royalty-free. Such licences may only be assigned, sub-let or otherwise dealt with on such terms as may be agreed by resolution of SPAA EC passed pursuant to Clause 6 of this Agreement.

14. CONFLICT WITH THE ARTICLES

Deleted:

Deleted: SPAA_180204.DOC
Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

In the event of any ambiguity created by or discrepancy between the provisions of this Schedule and the Articles, then it is the intention that the provisions of this Schedule shall prevail and accordingly the Shareholders shall exercise all voting and other rights and powers available to them so as to give effect to the provisions of this Schedule and shall further, if necessary, procure any required amendment to the Articles.

FURTHER ASSURANCE 15.

Each Shareholder shall co-operate with the other Shareholders and execute and deliver to the other Shareholders such other instruments and documents and take such other actions as may be reasonably requested from time to time in order to carry out, evidence and confirm their rights under, and the intended purpose of, this Schedule.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

ANNEX 1 TO SCHEDULE 4

Completion

- 1. *Meeting of the Directors:* On the SPAA Completion Date the Parties shall procure the holding of a meeting of the Board and the passing thereat of resolutions:
 - (A) appointing the SPAA EC Members as Directors:
 - (B) accepting the resignation as Directors of those persons (if any) who are not SPAA EC Members;
 - (C) appointing a bank nominated by SPAA EC as SPAA Ltd bankers and passing such resolutions relating to such appointment as the bank may require;
 - (D) approving the transfer of the two subscriber shares each to a Party other than or SPAA Ltd;
 - (E) convening an Extraordinary General Meeting of the Company immediately following the conclusion of the meeting of the Directors for the purposes referred to in paragraph 2;
 - (F) accepting the resignation as Company Secretary of that person who is not the Secretary;
 - (G) appointing the Secretary as Company Secretary;
 - (H) approving the change of registered address of SPAA Limited to [];
 - (I) approving the appointment of [] as auditors of SPAA Limited.
- 2. Extraordinary General Meeting: Upon the calling of the Extraordinary General Meeting referred to in paragraph 1(E), the Shareholders shall procure the giving of consents to short notice in respect of such Extraordinary General Meeting and shall vote thereat in favour of resolutions:
 - (A) authorising the directors to issue and allot at par one Share to each Party other than the SPAA Ltd and the two parties to whom the two subscriber shares in SPAA Ltd have been transferred;
 - (B) amending Clause 3 of the Memorandum of Association of SPAA Ltd by insertion of the new sub-clause set out in Annex 3 to this Schedule as sub-clause 3(1) and renumbering the remaining sub-clauses accordingly; and
 - (C) adopting the regulations in the form set out in Annex 4 to this Schedule as the Articles of Association of SPAA Ltd.
- 3. *Initial Subscription for Shares:* Forthwith upon the passing of the resolution of SPAA Ltd pursuant to paragraph 2 each party, save for and those two parties to whom transfers are approved under paragraph 1(D) above, shall complete, sign and deliver to SPAA Ltd an application for the allotment to that party of one share in consideration of the payment by cash of £1 on allotment in respect of each such share.
- 4. *Board Meeting:* SPAA Ltd shall procure the holding of a further meeting of the Board and the passing thereat of resolutions:
 - (A) approving the application of each party other than and the two parties to whom the two subscriber shares in SPAA Ltd have been transferred for a Share; and

Deleted: SPAA 180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

- (B) authorising the name of each party other than SPAA Ltd to be entered in the Register of Members of SPAA Ltd as holder of one Share and directing the sealing of a certificate in respect thereof.
- 5. *Allotment:* Upon receipt of the relevant consideration moneys referred to in paragraph 3, SPAA Ltd shall allot and issue one Share to each party whose application has been approved pursuant to paragraph 4 and shall register each such party as the holder of that share and shall prepare, seal and deliver to each party a share certificate in respect thereof in its name.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

 $i_{i}^{\prime\prime}$

ANNEX 2 TO SCHEDULE 4

Limitations on Dealings

- (i) The acquisition or disposal by SPAA Ltd of any share capital or other securities of any person.
- (ii) The reduction of SPAA Ltd's share capital, any variation of the rights attaching to any class of shares in its capital or any redemption, purchase or other acquisition by SPAA Ltd of any shares or other securities of SPAA Ltd.
- (iii) The making of decisions relating to material contracts or arrangements to which SPAA Ltd is a party.
- (iv) The making by SPAA Ltd of a material claim, disclaimer, surrender, election or consent for tax purposes.
- (v) The incurring of costs in respect of any period which are not envisaged by a budget drawn up and approved pursuant to Clause 8 of this Agreement.
- (vi) Entering into any contract or guarantee with a Shareholder or an Affiliate of a Shareholder.
- (vii) Entering into any contract of a material nature.
- (viii) The obtaining by SPAA Ltd of finance from a third party lender.
- (ix) The making of any change to SPAA Ltd's Memorandum of Association or the Articles.
- (x) The presentation of any petition for the winding-up of SPAA Ltd or the making of any application for an administration order in relation to SPAA Ltd or for the appointment of an administrator or receiver of SPAA Ltd.
- (xi) The commencement, settlement or defence of any litigation, arbitration or other proceedings brought by or against SPAA Ltd in an amount in excess of £25,000.
- (xii) The increase of the amount of debt with a maturity greater than 3 months owed by SPAA Ltd.
- (xiii) The entering into of an agreement of a type or length which is unusual in the context of the Business.

Deleted: SPAA_180204.DOC

Inserted: SPAA 180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

ANNEX 3 TO SCHEDULE 4

Amendment to Objects clause

To carry on the business of acting as a corporate vehicle in relation to the Supply Point Administration Agreement dated [x] in accordance with the terms and conditions of that Agreement.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

ANNEX 4 TO SCHEDULE 4

Form of New Articles

ARTICLES OF ASSOCIATION

THE COMPANIES ACT 1985-1989

ARTICLES OF ASSOCIATION

of

SPAA LIMITED (the "Company")

(Registered No. 04365599)

(adopted by Special Resolution passed on x)

1. Adoption of Table A

In these articles "Table A" means Table A scheduled to the Companies (Tables A to F) Regulations 1985 as amended prior to the date of incorporation of the Company. The regulations contained in Table A shall, except where they are excluded or modified by these articles, apply to the Company and, together with these articles, shall constitute the articles of the Company. No other regulations set out in any statute concerning companies, or in any statutory instrument or other subordinate legislation made under any statute, shall apply as the regulations or articles of the Company.

2. Interpretation

- 2.1. Words and expressions which bear particular meanings in Table A shall bear the same meanings in these articles.
- 2.2. In these articles:
- "Affiliate" means, in respect of any body corporate, a body corporate which is its subsidiary or holding company, or a company which is a subsidiary of that holding company, and each such company;
- "Authority " means the Gas and Electricity Markets Authority established by Section 1(1) of the Utilities Act 2000;
- "SPAA EC" means the SPAA Executive Committee appointed pursuant to the SPAA;
- "SPAA EC Members" means the members of the SPAA EC appointed pursuant to the SPAA;
- "SPAA" means the Supply Point Administration Agreement dated x;
- "SPAA Forum" means the body of that name appointed pursuant to the SPAA;
- "parties" means parties to the SPAA and "party" means a party to the SPAA;

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

<u>SPAA 040304.DOC</u> 67

"Retiring Shareholder" has the meaning given to that expression in Article 10.4; and

"shareholder" means the holder of a share or shares in the Company.

- 2.3. References in these articles to writing include references to any method of representing or reproducing words in a legible and non-transitory form.
- 2.4. Headings are for convenience only and shall not affect construction.
- 2.5. If, and for so long as, the Company has only one member, these articles shall (in the absence of any express provision to the contrary) apply with such modification as may be necessary in relation to such a company.

3. Share Capital

The authorised share capital of the Company at the date of adoption of this article is £1,000,000 divided in 1,000,000 shares of £1 each.

4. Restriction on Share Ownership

Save with the prior written consent of the directors, no person other than a nominee as referred to in Articles 9.4 and 9.5 shall be the holder of more than one share of the Company at any time.

5. Rights Attaching to Shares

- 5.1. The Shareholders shall exercise all rights attached to their shares to procure, so far as they are able, that no action shall be taken or resolution passed by the Company in respect of those matters set out in Article 5.2 except pursuant to a decision of SPAA EC or the SPAA Forum (as appropriate) or, on appeal, by the Authority, taken in accordance with the provisions of the SPAA. The right to vote on the matters set out in Article 5.2 shall constitute rights attaching to the Shares.
- 5.2. The matters referred to in Article 5.1 are:
 - 5.2.1. the acquisition or disposal by the Company of any share capital or other securities of any person;
 - 5.2.2. the reduction of the Company's share capital, any variation of the rights attaching to any class of shares in its capital or any redemption, purchase or other acquisition by the Company of any shares or other securities of the Company.
 - 5.2.3. the making of decisions relating to material contracts to which the Company is a Party;
 - 5.2.4. the making by the Company of a material claim, disclaimer, surrender, election or consent for tax purposes;
 - 5.2.5. the incurring of costs in respect of any period which are not envisaged by a budget drawn up and approved pursuant to Clause 8 of the SPAA;
 - 5.2.6. the making of any contract or guarantee with a member or an Affiliate of a shareholder;
 - 5.2.7. the making of any contract of a material nature;

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

- 5.2.8. the obtaining by the Company of finance from a third Party lender;
- 5.2.9. the making of any change to the Company's Memorandum of Association or these articles;
- 5.2.10. the presentation of any petition for the winding-up of the Company or the making of any application for an administration order in relation to the Company or for the appointment of an administrator or receiver of the Company;
- 5.2.11. the commencement, settlement or defence of any litigation, arbitration or other proceedings brought by or against the Company in an amount in excess of £25,000;
- 5.2.12. the increase of the amount of debt with a maturity greater than 3 months owed by the Company; and
- 5.2.13. the entering into of an agreement of a type or length which is unusual in the context of the business of the Company.
- 5.3. Each shareholder shall be entitled to dividends in respect of its share calculated in the same proportions as are set out in Clauses 8.10 of the SPAA rather than in proportion to the amounts paid up on the shares. Regulation 104 of Table A shall be modified accordingly.
- 5.4. Subject to the provisions of the Act and to any rights conferred on the holders of any other shares, any share may be issued with or have attached to it such rights and restrictions as the Company may by ordinary resolution decide or, if no such resolution has been passed or so far as the resolution does not make specific provision, as the directors may decide. Regulation 2 of Table A shall not apply.

6. Unissued Shares

Subject to the provisions of the Act and to these articles, any unissued shares of the company (whether forming part of the original or any increased capital) shall be at the disposal of the directors who may (subject to Article 4) offer, allot, grant options over or otherwise dispose of them to such persons at such times and for such consideration and upon such terms and conditions as they may determine.

7. Initial Authority to Issue Relevant Securities

Subject to any direction to the contrary which may be given by the Company in general meeting and to Article 4, the directors are unconditionally authorised to exercise all powers of the Company to allot relevant securities. The maximum nominal amount of relevant securities that may be allotted under this authority shall be the nominal amount of the unissued share capital at the date of incorporation of the Company or such other amount as may from time to time be authorised by the Company in general meeting, The authority conferred on the directors by this article shall remain in force for a period of five years from the date of incorporation of the Company but may be revoked varied or renewed from time to time by the Company in general meeting in accordance with the Act.

8. Exclusion of Rights to Offers on a Pre-emptive Basis

Section 89(1) of the Act shall not apply to the allotment by the Company of any equity security.

Deleted: SPAA 180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

9. Transfer of Shares

- 9.1. The instrument of transfer of a subscriber's share which is not fully paid need not be executed by or on behalf of the transferee. Regulation 23 of Table A shall be modified accordingly.
- 9.2. Otherwise than in accordance with Articles 9.3 and 9.4 no shareholder shall:
 - 9.2.1. pledge, mortgage (whether by way of fixed or floating charge) or otherwise encumber its legal or beneficial interest in its shares; or
 - 9.2.2. sell, transfer or otherwise dispose of any of such shares (or any legal or beneficial interest therein); or
 - 9.3.3. enter into any agreement in respect of the votes attached to shares; or
 - 9.4.4. agree, whether or not subject to any condition precedent or subsequent, to do any of the foregoing.
- 9.3. A member may transfer its shares to a successor holder of its Gas Suppliers, Licence or Gas Transporters Licence (as appropriate) in circumstances where such Affiliate becomes a Party at the same time as such member ceases to be a party.
- 9.4. If any shareholder ceases to be a Party for any reason (the "Retiring Shareholder"), then upon written notice to the Retiring Shareholder by any other shareholder, the Retiring Shareholder shall transfer at par to a nominee for all the shareholders (other than the Retiring Shareholder) selected by the directors the Share held by the Retiring Shareholder. All costs and expenses of such transfer shall be for the account of the Retiring Shareholder.
- 9.5. If a Retiring Shareholder shall fail or refuse to transfer any Shares in accordance with its obligations under Article 9.4 the directors may authorise the Company to execute and deliver a transfer from the Retiring Shareholder to a nominee on behalf of the Retiring Shareholder. The Company may accept the consideration for the transfer and hold it on trust for the Retiring Shareholder, which acceptance shall be a good discharge to the nominee and may set off such amount against the costs and expenses of the transfer. The directors shall cause the transferee to be Registered as the holder of such shares and following the registration of the transfer the validity of the proceedings shall not be questioned by any person.
- 9.6. The nominee referred to in Articles 9.4 and 9.5 shall hold shares transferred to it until such time as it is directed by the directors to transfer them (or some of them) to one or more parties. For the avoidance of doubt, wherever in these Articles a percentage figure of the number of shares in issue is referred to, this figure shall be calculated as if all shares held by the nominee were not in issue.

10. Proceedings at General Meetings

10.1. The quorum at any general meeting shall consist of a majority in number of the Shareholders in person or by proxy. PROVIDED THAT one shareholder must be a representative of an Industrial and Commercial Supplier, two shareholders must be representatives of Domestic Suppliers and two shareholders must be representatives of Transporters. Regulation 40 of Table A shall be modified accordingly. (The terms Industrial and Commercial

Deleted: y

Deleted: and

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

Suppliers Domestic Suppliers and Transporters shall have the same meaning as that provided in the most recent version of SPAA).

Deleted: and

- 10.2. If, and for so long as, the company has only one member, that member or the proxy for that member or, where that member is a corporation, its duly authorised representative shall be a quorum at any general meeting of the company or of the holders of any class of shares. Regulation 40 of Table A shall be modified accordingly.
- 10.3. The chairman at any general meeting shall not be entitled to a second or casting vote. Regulation 50 of Table A shall not apply.
- 10.4. In the case of a corporation a resolution in writing may be signed on its behalf by a director or the secretary of the corporation or by its duly appointed attorney or duly authorised representative. Regulation 53 of Table A shall be extended accordingly.

Deleted: e

11. Votes of Members

At a general meeting, but subject to any rights or restrictions attached to any shares, on a show of hands every member who (being an individual) is present in person or (being a corporation) is present by a duly authorised representative and every proxy for any member (regardless of the number or the holdings of the members for whom he is a proxy) shall have one vote, and on a poll every member who is present in person or by proxy shall have one vote for every share of which he is the holder. Regulation 54 of Table A shall not apply.

12. Delivery of Proxies

The instrument appointing a proxy shall be in writing (hard copy or e-mail) from a duly authorised representative of the member. The instrument appointing a proxy and the power of attorney or other authority, if any, under which it is signed or a certified copy of that power or authority:

- shall be deposited at the registered office of the company or at such other place within the United Kingdom as is specified for that purpose in the notice convening the meeting of the members, not less than 24 hours before the time appointed for the meeting or adjourned meeting at which the person named in the instrument proposes to act or in the case of a poll taken subsequently to the date of the meeting or the adjourned meeting, before the time appointed for taking of the poll and in default the instrument of proxy shall not be treated as valid; or
- 12.2 if sent by e-mail, shall be produced in a scanned image file (.PDF format) sent to [insert e-mail address] or such other e-mail address as is specified for that purpose in the notice convening the meeting of the members, not less than 24 hours before the time appointed for the meeting or adjourned meeting at which the person named in the instrument proposes to act or in the case of a poll taken subsequently to the date of the meeting or the adjourned meeting, before the time appointed for taking of the poll and in default the instrument of proxy shall not be treated as valid. In order for the instrument appointing proxy to be deemed valid, the e-mail that sends such instrument shall be required to quote the code or identifier assigned to that member by the Company Secretary.

Regulation 62 of Table A shall not apply.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18
March.DOC

13. Alternate Directors

Each director shall have as his alternate for the purposes of these Articles the alternate appointed by him pursuant to Clause 6.17 of the SPAA. Regulation 65 of Table A shall not apply.

14. Delegation of Directors' Powers

The Directors may delegate any of their powers (with power to sub-delegate) to committees consisting of such person or persons (whether directors or not) as they may resolve. Any such committee shall exercise only powers expressly delegated to it and shall comply with any regulations imposed on it by the directors. Regulation 72 of Table A shall be modified accordingly and references in Table A to a committee of Directors or to a Director as a member of such a committee shall include a committee established under this article or such person or persons.

15. No Age Limit or Share Qualification

No Director shall be required to retire or vacate his office, and no person shall be ineligible for appointment as a director, by reason of his having attained any particular age. No shareholding qualification for directors shall be required.

16. Exclusion of Rotation Requirements and Other Provisions

The Directors shall be the SPAA EC Members from time to time. Regulations 73 to 80 (inclusive) and the last sentence of Regulation 84 of Table A shall not apply.

Deleted: r

17. Disqualification and Removal of Directors

The office of a Director shall be vacated if he ceases to be a SPAA EC Member. Regulation 81 of Table A shall not apply.

18. Directors' Gratuities and Pensions

Regulation 87 of Table A shall not apply.

19. Notice and Conduct of Board Meetings

Notice of a meeting of the Directors shall be deemed to be properly given to a Director if it is given to him personally or sent in writing to him at his last known address or any other address given by him to the company for this purpose, or by any other means authorised in writing by the Director concerned. Notice shall be given in this manner to all Directors including any Director who is for the time being absent from the United Kingdom. A director may waive notice of any meeting either prospectively or retrospectively. All resolutions of the Board shall be made by unanimous vote of the Directors present or participating by conference telephone. In the case of an equality of votes, the chairman shall not have a second or casting vote. Regulation 88 of Table A shall be modified accordingly.

20. Quorum for Board Meetings

The quorum for meetings of the board shall be constituted by the attendance of the one (1) I&C Supplier Member, one (1) Large Domestic Supplier Member, one (1) Small Domestic Supplier Member, one (1) Large Transporter Member and one (1) Small Transporter Member (each as defined in the SPAA) (or their alternates) in person or participating by conference telephone call throughout such meeting. Regulation 89 of Table A shall not apply.

Deleted: and

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

March.DOC

SPAA 040304.DOC

72

21. Participation in Board Meetings by Telephone

All or any of the members of the board or any committee of the board may participate in a meeting of the board or that committee by means of a conference telephone call provided that participants acknowledge that they can speak to and hear each other. A person so participating shall be deemed to be present in person at the meeting and shall be entitled to vote or be counted in a quorum accordingly. Such a meeting shall be deemed to take place where the largest group of those participating is assembled, or, if there is no such group, where the chairman of the meeting is then situated.

22. Resolution in Writing

A resolution in writing executed by all the Directors or by all the members of a committee for the time being shall be as valid and effective as a resolution passed unanimously at a meeting of the board or, as the ease may be, of the committee properly convened and constituted. The resolution may be contained in one document or in several documents in like form each executed by one or more of the Directors or members of the committee concerned. A resolution signed by an alternate Director need not also be signed by his appointor and, if it is signed by a Director who has appointed an alternate Director, it need not be signed by the alternate Director in that capacity. Regulation 93 of Table A shall not apply.

23. Directors May Vote When Interested

A Director who to his knowledge is in any way, whether directly or indirectly, interested in a contract or proposed contract with the company shall declare the nature of his interest at a meeting of the directors in accordance with the Act. Subject where applicable to such disclosure, a director shall be entitled to vote in respect of any contract or proposed contract in which he is interested and if he shall do so his vote shall be counted and he shall be taken into account in ascertaining whether a quorum is present. Regulations 94 and 95 of Table A shall not apply.

24. Official Seal

The company may exercise all the powers conferred by the Act with regard to having any official seal and such powers shall be vested in the Directors. Subject to the provisions of the Act, any instrument to which an official seal is affixed shall be signed by such persons, if any, as the directors may from time to time determine.

25. Notices

Any notice or other document may be served on or delivered to any member by the company either personally, or by sending it by post addressed to the member at his Registered address or by fax or telex to a number provided by the member for this purpose, or by leaving it at his Registered address addressed to the member, or by any other means authorised in writing by the member concerned. In the case of joint holders of a share, service or delivery of any notice or other document on or to one of the joint holders shall for all purposes be deemed a sufficient service on or delivery to all the joint holders. Regulation 112 of Table A shall not apply.

26. Time of Service

Any notice or other document, if sent by the Company by post, shall be deemed to have been served or delivered forty eight hours after posting and, in proving such service or delivery, it shall be sufficient to prove that the notice or document was properly addressed, stamped and put in the post. Any notice or other document left by the company at a Registered address otherwise than by post, or sent by fax or telex or

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18

<u>SPAA 040304.DOC</u> 73

other instantaneous means of transmission, shall be deemed to have been served or delivered when it was so left or sent. Regulation 115 of Table A shall not apply.

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC

SCHEDULE 5

Possible Future Developments

1. Purpose of this Schedule

The circumstances listed in this Schedule 5 have been identified by certain Parties as ones which may be capable of resolution by modification of this Agreement. The Parties, therefore, agree to investigate the following, and charge and request the SPAA EC to investigate these matters as a priority.

2. Issues

- 2.1 The incorporation of any part of the draft domestic code of practice into this Agreement.
- 2.2 The incorporation of any part of the Review of the Gas Metering Arrangements project into this Agreement.
- 2.3 The incorporation of any part of the Supply Point Administration provisions from the Gas Transporter's Network Codes.

Deleted: ¶

The incorporation of any part of the draft industrial and commercial code of practice into this Agreement.

Deleted: <#>The issue of Gas transporters and other relevant parties who may hold a licence obligation to join the SPAA in the future, and to agree a working relationship with such parties.¶

Formatted: Bullets and Numbering

Deleted: SPAA_180204.DOC

Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers
amended draft) 18
March.DOC

SCHEDULE 6

SPAA Products

None.

Deleted: SPAA_180204.DOC Inserted: SPAA_180204.DOC

Deleted: SPAA (suppliers amended draft) 18 March.DOC